

# THE UNIVERSITY of EDINBURGH

This thesis has been submitted in fulfilment of the requirements for a postgraduate degree (e.g. PhD, MPhil, DClinPsychol) at the University of Edinburgh. Please note the following terms and conditions of use:

- This work is protected by copyright and other intellectual property rights, which are retained by the thesis author, unless otherwise stated.
- A copy can be downloaded for personal non-commercial research or study, without prior permission or charge.
- This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author.
- The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author.
- When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

## **An Investigation of Governance in Scottish Charities**

Vicky Lambert

PhD (Management)

The University of Edinburgh

2010

#### **ACKNOWLEDGEMENTS**

I would like to thank first and foremost my supervisor, Professor Irvine Lapsley, for all his kind words of encouragement and support since my undergraduate days. I would also like to show my gratitude towards Dr. Ingrid Jeacle who acted as my second supervisor who has guided me with her level-headedness and sound advice.

I owe special thanks to my husband, Adam and my parents who have shown unexpected levels of patience and support. Thanks are also due to my friends and work colleagues who have helped along the way with words of encouragement, in particular Mary Ho.

In the completion of this thesis, I would also like to thank all those who participated in this study. Their insights and responses were invaluable to this study and greatly appreciated.

#### **DECLARATION**

I declare that this is my own work.

Signed: V. lambot.

Vicky Lambert

#### **ABSTRACT**

This thesis investigates governance in large Scottish charities. In particular, the focus is on internal aspects of governance, predominantly the functioning of the board. The influence of business on charities is also investigated through considering the impact of individuals with business expertise on the boards of charities. Charity governance in the Scottish context has been under-research, this thesis attempts to fill this research gap. A multi-method approach was employed, encompassing both quantitative and qualitative research. The study utilised a survey in the form of a self-completion questionnaire, and three case studies to examine governance in Scottish charities. Social construction provides the framework of the analysis. This thesis provides an insight into the functioning of the boards of Scottish charities, in-line with previous research in other countries, governance issues are apparent. However, in relation to the impact of business expertise in charities, there is evidence that business members of charitable boards make contributions to the governance of these organisations, and charities can gain considerable value from having outside specialists on the board.

#### An Investigation of Governance in Scottish Charities

PART ONE: RESEARCH CONTEXT	1
Chapter 1. Introduction	1
1.1 The Charity Sector in Scotland	1
1.2 Overview of Research and Contribution	4
1.3 Structure of Thesis	5
Chapter 2. Governance in Charities	7
2.1 Introduction	7
2.2 The Role of the Board	7
2.3 The Relationship Between the Board and Management	13
2.4 The Composition of the Board	16
2.5 Board and Organisational Effectiveness	18
2.6 Conclusion: Governance in Charities	23
Chapter 3. The Impact of Business Expertise on the Governance and	
Management of Charities	26
3.1 Introduction	26
3.2 The Impact of Business Expertise on Charity Governance	27
3.3 The Adoption of Business Practices by Charities	31
3.4 Conclusion: The Impact of Business Expertise on the Governance and Management of Charities	38
Chapter 4. Conceptual Framework: Social Construction	40
4.1 Introduction	40
4.2 Social Construction	40
4.2.1 Legitimation	42
4.2.2 Isomorphism	44
<ul><li>4.3 The Use of Social Construction in Organisational Research</li><li>4.4 The Use of Social Construction in Research on Nonprofit</li></ul>	48
Organisations	53
4.5 Conclusion and The Relevance of Social Construction to this	
Study	61
Chapter 5. Methodology	65
5.1 Introduction	65
5.2 Research Questions	66
5.3 Research Method	66
5.3.1 Defining the Population	68
5.3.2 The Survey	71
5 3 2 1 Questionnaire Design	75

5.3.2.2 The Pilot Study	78
5.3.2.3 The Main Survey	81
5.3.3 Case Study Research	82
5.4 The Research Design & Social Construction	87
5.5 Limitations of the Study	89
5.6 Conclusion: Methodology	90
3.0 Conclusion. Methodology	70
PART TWO: FINDINGS	
SURVEY RESULTS	91
Chapter 6. Governance in Scottish Charities: Survey Results	92
6.1 The Roles and Responsibilities of the Board	92
6.1.1 Defining the Roles and Responsibilities of the Board	93
6.1.2 Communicating the Roles and Responsibilities of the	75
Board	95
6.1.3 Ensuring Board Members are Aware of Roles and	98
Responsibilities	, ,
6.2 The Distinction between Strategy and Operations	101
6.2.1 Assessing the Proportion of Board Time Spent on	
Strategy and Operations	101
6.2.2 Types of Strategic and Operational Activities	
Undertaken by the Board	103
6.3 The Effectiveness of Governance in Charities	109
6.3.1 How Effective Governance is in Charities	109
6.3.2 Improving Effectiveness of Charity Governance	111
6.4 Control and Regulation in Charities	115
6.4.1 Charity Views Concerning the Nature of Regulation	115
6.4.2 Charities Justification for Views on Nature of	
Regulation	116
6.4.3 The Impact of OSCR on Charity Governance	120
6.5 Conclusion: Governance Survey Results	123
old Conclusion. Governance Bulvey Results	123
Chapter 7: The Impact of Business Expertise on Charity Boards: Survey	
Results	125
7.1 The Presence of Business People on Charity Boards	125
7.1.1 Proportion of Business People on Charity Boards	125
7.1.2 Charities' Satisfaction with the Proportion of Business	123
People on their Boards	127
7.2 Impact of Business People	131
<u>.</u>	131
7.2.1 Practices Introduced by Business People	131
7.2.2 Response to Practices Introduced by Business People	13/
7.3 The Motivations of Business People and their Understanding of	140
Charity Context  7.2.1 The Metivation of Pusiness People Sitting on Charity	140
7.3.1 The Motivation of Business People Sitting on Charity	1.40
Boards	140

7.3.2 The Extent to which Business People Understand	
Charity Context	140
7.3.3 Ensuring Business People Understand the Context of the	1.40
Charity	142
7.4 The Impact of Business People on Public Confidence	142
7.5 Conclusion: Business Expertise Survey Results	144
CASE STUDY RESEARCH	147
Chapter 8: Governance in a Social Care Charity: Results from a Case Study	148
8.1 Introduction	149
8.2 Background	149
8.3 Governance	150
8.3.1 Governance Structure	151
8.3.2 The Board Cycle	152
8.3.3 Board Composition	156
8.4 The Impact of Business People on the Board	159
8.4.1 Motivation of Business People	159
8.4.2 Business Peoples Understanding of the Organisation	160
8.4.3 The Use of Business Expertise on the Board	161
8.4.4 Pressures to Become more Business-Like	163
8.5 Conclusion: Robertson Social Care	164
Chapter 9: Governance in a Housing Association: Results from a Case Study	166
9.1 Introduction	166
9.2 Background	167
9.3 Governance	168
9.3.1 Governance Structure	168
9.3.2 The Role of the Board	170
9.3.3 The Distinction Between Strategy and Operations	172
9.3.4 Board and Management Relations	173
9.3.5 Board Composition	175
9.4 The Impact of Business People on the Board	179
9.4.1 Pressure to Become More Business-Like	179
9.4.2 The Use of Business Expertise on the Board	180
9.5 Conclusion: McPherson Housing Association	181
Chapter 10: Governance in a Further Education College: Results from a Case Study	183
10.1 Introduction	183
10.2 Governance Issues	184
10.2.1 The Role of the Board	184
10.2.1.1 The Role of the Board in Setting the Strategic	
Direction	186
10.2.1.2 The Role of the Board in Target Setting &	

$\mathcal{E}$	86
10.2.1.3 The Role of the Board in Monitoring &	
$oldsymbol{c}$	88
	90
1	94
1	95
	95
10.3.2 The Use of Individual Board Members Experience &	
1	96
10.4 Conclusion: Mackay College	99
PART THREE: CONCLUSIONS	
Chapter 11: Conclusions 2	202
11.1 Discussion & Conclusion 2	202
11.2 Contribution of Research	211
References 2	213
Appendices: 2	223
Appendix 1: Pilot Study Questionnaire 2	224
Appendix 2: Analysis of Pilot Study Questions 2	225
	227
	229
•	237
11	238

#### List of Tables

4.1 The Use of Social Construction in Nonprofit Research	53
<ul><li>5.1 Total Number of Charities Registered with OSCR as at 1/8/2006</li><li>5.2 The Use of Specific Research Methods to Explore Social Construction in</li></ul>	68
Charity Governance	87
6.1 Definition of Roles and Responsibilities of the Board	92
6.2 Location of Roles and Responsibilities	94
6.3 What is Done to Make Sure Board Members are Aware of their Roles	
and Responsibilities in Charities	97
6.4 Percentage of Time Spent on Strategic Rather than Operational Activities	101
6.5 Effectiveness of Governance in Charities	109
6.6 Control/ Regulation of Charities	114
7.1 Percentage of Business People on Charity Boards	124
7.2 Charities Satisfaction with the Proportion of Business People Serving on	
their Board	126
7.3 Examples of Business Practices Introduced by Business Members	129
7.4 Response to Practices Introduced by Business People	136
7.5 Level of Understanding of Charity Context	139
7.6 Impact of Business People on Public Confidence	141
8.1 Interviews Conducted at Robertson	147
8.2 Board Composition of Robertson	155
9.1 Interviews Conducted at McPherson	165
9.2 Board Composition of McPherson	174
10.1 Interviews Conducted at Mackay	182
10.2 Board Composition of Mackay	193

#### PART ONE: RESEARCH CONTEXT

#### **CHAPTER 1**

#### INTRODUCTION

#### 1.1 The Charity Sector in Scotland

The charity sector of many economies has assumed increasing significance as a provider of services (Bradley et al, 2003; Weisbrod, 1988). The United Kingdom is no exception to this trend, particularly in light of the current Governments idea of 'the Big Society' which aims to present a greater role in public services to the voluntary sector (Cabinet Office, 2010a, 2010b). In Scotland, the charity sector represents a significant proportion of the economy, with approximately 23, 114 charities generating an annual income of approximately £10.9 billion (OSCR, 2008). As a result of the considerable economic contribution and the increasing expectation placed on these organisations in the delivery of public services, the importance of good governance in charities has been recognised as key (OSCR, 2008; Crawford et al, 2009). However, it also presents itself as a considerable challenge.

Over the past decade, the charitable sector in Scotland has undergone significant change. The context of this change has been at a time when in Scotland, similar to other countries, the sector has been rocked by a number of high profile scandals; most notably that of Moonbeams, and Breast Cancer Research. These very public failures have led to serious questions of trust within the sector. Moonbeams was an Edinburgh based Cancer charity which raised money by selling confectionary in order to send seriously ill children from underprivileged backgrounds on holiday. This charity was closed down after it emerged that only £70,000 of the £3 million raised actually went to help these children (Davidson, 2003). Similarly, with the case of Breast Cancer research, only £1.5 million of the £13 million raised went to charitable causes (Robertson, 2003). These incidents had a serious impact on the

Scottish charity sector, with charity donations from the Scottish public falling by 30 per cent (Horton, 2006).

Failures such as these have raised concerns over the governance of these organisations, and have resulted in a tighter regulatory regime for charities which has sought to increase transparency and rigour in the sector, and to fix defects which were apparent under the previous framework. A new charity regulator – the Office of the Scottish Charity Regulator (OSCR) - was established in December 2003 as an Executive Agency and took up its full powers as a non-ministerial department of the Scottish Administration when the Charities and Trustee Investment (Scotland) Act 2005 came into force in April 2006. OSCR was set up as the independent regulator and registrar of charities, and is charged with ensuring that charities comply with legislation. OSCRs responsibilities include the following:

- Maintaining the Scottish Charity Register
- Granting charitable status
- Granting consent to proposed changes to a charity
- Monitoring charities' compliance with the new legislation, and
- Investigating apparent mismanagement or misconduct

(OSCR, 2006b)

In Scotland, only if an organisation has been entered into the Scottish Charity Register, can it be deemed to be a charity. In order to become a charity, organisations must pass the charity test, which means they must have one or more specified charitable purposes, and provide public benefit in Scotland or elsewhere. <sup>1</sup> Charities are governed by a board of volunteers, generally referred to as either trustees, board members or directors.

<sup>1</sup> For the purpose of this study the term charity, nonprofit and voluntary will be used interchangeably

OSCR sets out the following role of these charity trustees:

Charity trustees are responsible for the governance and strategy of their charity. They are responsible for making sure that their charity is administered effectively, and is able to account for its activities and outcomes both to us (OSCR) and to the public (OSCR, 2010).

The duties of charity trustees are set out in Section 66 of the Charities and Trustee Investment (Scotland) Act 2005. It sets out four general duties which are deemed to be fundamental to the trustees of the charity, and a number of specific duties which the trustees must comply with. The general duties are outlined in the Act as follows, trustees must:

- Act in the interest of the charity.
- Operate in a manner consistent with the charity's purposes.
- Act with due care and diligence.
- Ensure that the charity complies with the provisions of the 2005 Act and other relevant legislation.

A number of other duties outlined include: updating the charities details with OSCR; reporting to OSCR; financial record keeping and reporting; controlling fundraising; and providing information to the public. Charity trustees must fulfil these duties without remuneration.

Despite these regulatory changes, governance issues in Scottish charities still appear to be at large. A recent failure in the sector was that of One Plus, a charity providing support to one-parent families, this was a large charity with a turnover of £11m in 2005. In 2007 it went into liquidation, with OSCR launching an inquiry into the underlying causes of the financial problems which led to this. The governance of the charity was identified as a major cause of it's failure, such issues were particularly attributed to a lack of business expertise on the board, and included: the lack of skill of the board to run a multi-million pound 'business', particularly in the area of

finance; heavy reliance on the Chief Executive who had control of both strategic objectives and day-to-day management; and out-of-date and incomplete financial information being presented to the board.

The Scottish context therefore presents itself as an interesting and relevant setting for the study of governance in charities. As demonstrated above, despite regulatory changes, governance failures are still apparent. In addition, a need has been identified for charities to become more business-like in governing their organisations, however, the future governance and management of charitable organisations and their ability to adopt and deploy specialist business-related functions, while preserving their distinctive ethos, represents a fundamental challenge (NCVO, 1999; Sergeant et al, 2000; Storey, 2001; Bradley et al, 2003). As outlined above, the notorious scandals which have occurred in the last decade have placed increased scrutiny on charities, with the ultimate question in this situation being: where was the board?

#### 1.2 Overview of Research and Contribution

This research examined governance in Scottish charities. In particular, the focus was on the specific governance issues in charities, particularly in relation to the functioning of the board. The impact of external expertise, particularly business expertise was also investigated by considering the presence of business people on charity boards. These areas were explored using aspects of social construction – isomorphism and legitimation. A multi-method approach was employed, encompassing both quantitative and qualitative research. The study utilised a survey in the form of a self-completion questionnaire, and three case studies to examine governance in Scottish charities.

Despite the interesting study setting which the Scottish charity sector provides, there has been a surprising lack of academic research into the governance of these organisations in Scotland, this thesis attempts to address this gap. This research will contribute to the further understanding of governance in charities, and also the role of

external expertise – particularly business expertise – on charity boards. This thesis also aims to make a theoretical contribution to the literature by exploring these issues using social construction, particularly the ideas of Isomorphism (Di Maggio and Powell, 1983; Meyer and Rowan, 1977) and Legitimation (Berger and Luckmann, 1967; Meyer and Rowan, 1977). It is envisaged that this research will be of interest to academics; charities, particularly managers and trustees; policy makers; and the Scottish charity regulator.

#### 1.3 Structure of Thesis

This thesis is structured into a further ten chapters. The following four chapters present the context of the research. Chapters two and three provide an overview of the extant literature in this area. Chapter two focuses specifically on issues in charity governance, by first providing a definition of governance in charities and an overview of four key 'internal' aspects of governance in charities: the role of the board, the relationship between the board and management, board and organisational effectiveness, and the composition of the board. Chapter three then examines the impact of business practices and expertise on charities, initially it examines the specific research which focuses on the impact of business practices and expertise on charity boards, before considering literature which looks at this impact more broadly in the management of these organisations. Chapter four provides an overview of the conceptual framework of the study – social construction. This chapter begins by considering some seminal contributions in this area and draws specifically on two key aspects – legitimation and isomorphism. Initially, applications of these ideas are considered in the broader organisational literature, before looking specifically at nonprofit organisations. This chapter concludes with a discussion of how the framework will be applied in this study. Chapter five provides a rationale and justification for the design of this multi-method study of charity governance. The research questions are outlined, along with the research methods and data collection instruments. The design of the research is also considered with regard to the mobilisation of the conceptual framework.

The next section of the study presents the findings. Chapters six and seven focus on a survey of the largest section of Scottish charities. Chapter six examines governance aspects including: the role of the board, the division of strategic and operational tasks, board effectiveness, and regulation. Chapter seven focuses more specifically on the impact of business expertise on charity boards. Chapters eight, nine, and ten present findings of three case studies; a Housing Association, a Social Care charity, and a Further Education College. Finally, chapter eleven draws together the key findings of this research, and offers some conclusions and suggestions for future research.

#### **CHAPTER 2**

#### **GOVERNANCE IN CHARITIES**

#### 2.1 Introduction

Governance in charitable organisations has traditionally focused on the functioning of the Board. Such areas are classified by Ostrower and Stone (2006) and Stone and Ostrower (2007) as focusing on the role of the Board, the relationship between the Board and staff, the composition of the Board, and the link between Board and organisational effectiveness. This is also noted by Hyndman and McDonnell (2009), who classify these as 'internal aspects' of governance. They also consider the relationships charities have with donors, regulators and beneficiaries which are termed as 'external aspects' of governance. Hyndman and McDonnell (op cit.) report on the lack of a clear definition of charity governance, and propose a broad definition of charity governance as encompassing both internal and external aspects. They define charity governance as:

Relating to the distribution of rights and responsibilities among and within the various stakeholder groups involved, including the way they are accountable to one another; and also relating to the performance of the organisation, in terms of setting objectives or goals and the means of attaining them (Hyndman & McDonnell, p.9; 2009).

This chapter will consider the key aspects of charity governance as identified above. Due to the nature of this study, it will focus on 'internal aspects' of governance, namely: 2.2 the role of the Board; 2.3 the relationship between the Board and staff, particularly management; 2.4 the structure and composition of the Board; and 2.5 Board and organisational effectiveness.

#### 2.2 The Role of the Board

A key aspect of good Governance is widely regarded as being the clear documentation and articulation of the roles and responsibilities of the Board. These are laid down in a variety of publications which outline guidance for the sector, such examples for Scottish charities include OSCR *Guidance for Charity Trustees* (OSCR, 2006) which provides guidance for trustees of Scottish charities on the expectation placed on charity trustees under the Charities and Trustee Investment (Scotland) Act 2005. Although this is aimed at trustees of small to medium sized charities (with an income of below £500 000) because they make up the greatest majority of charities in Scotland, it is also perceived as a useful tool for trustees of larger charities. Charities are also covered in the *Good Governance Standard for Public Services* (CIPFA, 2004). Such guidance is also issued on a sectoral basis, with the Regulatory Code of Governance for Housing Associations issued by the Scottish Housing Regulator (Scottish Housing Regulator, 2008), and the *Guide for College Board Members* issues by Scotland's Colleges (2006).

The above guidance places clear emphasis on the importance of clear roles and responsibilities to ensure effective charity governance. The emphasis of such guidance is that the nature of the role is strategic, this is mirrored in the academic literature where a plethora of studies exist in defining what the key roles and responsibilities of the board are (Houle, 1990; Axelrod, 2005). Prescriptive lists such as this include Houle's (1990) widely cited list, which includes: determining mission and purpose; appointing, supporting and appraising the Chief Executive; approving and monitoring the services and programmes of the organisation; raising money; ensuring effective fiscal management; strategic planning; enhancing the public image of the organisation; developing itself as a board, from recruitment to assessment of its own performance; understanding the relationship between staff, board and volunteers; organising itself so that it operates effectively; and ensuring sound risk management policies. A similar group of roles is also outlined by Ingram (2002). Axelrod (2005) lists four primary responsibilities that a Board should carry out, namely; definition and advancement of the mission of the organisation, development

and conservation of the organisation's resources, ensuring oversight of management and assessment of the organisation and developing relationships between the organisation and its stakeholders.

The importance attached to such prescriptive Board roles has also been investigated. One such study is that of Brown & Guo (2010), who investigate how chief executives of U.S. community foundations perceive role importance. The key roles identified included: fund development, such as securing donations for the organisations; strategy and planning, where executives talked about setting the direction of the organisation through long-range strategic planning and attending retreats; financial oversight; public relations; Board member vitality, which was defined in terms of the commitment and engagement of board members; policy development and monitoring; and the board relationship with the executive, particularly in terms of monitoring the executives performance. Key influences that shaped the Boards role were also identified, for example, organisations that reported the Board role as one of resource development were found to operate in environments that were resource constrained. Focus on a strategic role was related to organisational complexity, with the oversight role typical of large diversified organisations.

Further investigation of the importance attached to these various Board roles has been investigated by Leatherwood and O'Neal (1996), who considered both corporate and nonprofit boards in the U.S. They found that members of nonprofits reported that their Boards were more responsible for what they termed service than for strategy or control. Service was described as looking after the organisations reputation, advising top management, and selecting new board members.

Inglis et al (1999), surveyed community nonprofit organisations, asking them to rate the importance of fourteen roles and responsibilities and the extent to which they were currently being fulfilled in the organisation. A gap between role importance and fulfilment was found to exist; the results showed that the most important roles were perceived as being roles associated with responding to community needs and ensuring a mission and vision for the organisation, however the fulfilment ratings were found to be lower than the ratings related to importance, pointing towards a need for improvement in the fulfilment of role outcomes – only a satisfactory level of performance was reported. A framework was developed for roles and responsibilities which consisted of three areas: strategic activities, operations and resource planning. A further study by Inglis and Weaver (2000) considered whether applying this framework to the design of Board agendas would result in increased fulfilment of these key roles; the findings here reported that both board members and the executive felt that the framework allowed them to keep their focus on important items relating to their key roles, therefore enhancing their ability to govern. Similarly a study by Green et al (2001) reports gaps between perceived and actual behaviour, with differences reported as to what functions and activities boards and chief officers believe the other should and does perform.

In a similar study to Inglis et al (1999), Iecovich (2004) considers fulfilment of Board roles in an Israeli nonprofit. In this study, nine board roles and responsibilities were examined and perceptions of these roles and their fulfilment by chairs and executive directors were sought, the roles investigated were: annual budget allocations, fundraising, change in top management, hiring decision on paid senior staff, change in general admin procedures, job descriptions for senior staff, changes in specific programs or services, connections with local and national agencies, and advocacy and lobbying. Here, fulfilment of Board roles were reported, and three major groupings of these roles were uncovered: internal focus, such as HR; external-internal focus, such as fiscal management and policy making; and external focus such as maintenance of relationships with the task environment. There was also a general consensus between the chairs and the executive directors in terms of the boards level of involvement in various roles and responsibilities, this was specifically in terms of fiscal issues and the relationship with the task environment, with chairs perceiving higher levels of involvement in these roles than the executive directors.

Specific focus is given to the role of the Board in Strategic Planning and Monitoring by Cornforth and Edwards (1999), Stone, 1991, Parker (2003) and Miller (2002).

The specific contribution of Boards to strategy is investigated by Cornforth and Edwards (1999), who consider this issue in public and voluntary organisations. The findings from this study show that the strategic contribution of boards varies widely between organisations; in the cases in this study, it was found that only slight strategic contributions were made by the Board of the local voluntary organisation and the Local Authority School, in contrast to the national voluntary organisation and College, where the contribution was much larger. This was the result of the interaction of a number of issues, these were found to include: regulation and legal responsibilities, which meant that the compliance role was dominant in all organisations and tended to move the focus away from strategic issues; the traditions of the sector and norms of governance; the way that board members are chosen; the skills and experience of board members, for example, the requirements placed on the College to recruit people with business experience meant that the Board saw strategy making as part of its role and were more able in dealing with strategic issues; the size and status of the organisation; and the way that Boards are organised and run. In addition, it is also reported that wider institutional factors which have an impact on these issues can leave boards experiencing tensions and trade-offs that can result in the strategic role being overshadowed by other roles. Issues surrounding the way certain governance processes were managed impacted on the strategic contribution of the board. Another important finding was the lack of clarity of the role of the board, highlighting the need for greater clarification of this role within organisations.

More specifically, Strategic planning is investigated by Stone (1991) who found that a focus on high level policy issues were present in nonprofits with clearly understood missions, and clearly structured boards. Formal planning was also dependent on clearly structured boards. Parker (2003) considers the Boards role in the financial management strategy of a community welfare organisation. The commitment to the organisations core philosophies shown by the directors was a key influence on the financial management strategy of the organisation. Issues of balance between the level of attention given to social, welfare, operational, and financial strategy issues were found, and as a result, financial management strategy dominated board discussions. Findings show that governance in the community welfare organisation is

predominantly financially focussed in terms of both strategic orientation and predominance of boardroom discussions.

The importance of monitoring as a Board role is recognised in the literature (Houle, 1990; Carver, 2006; Axelrod, 2005) and is the subject of specific investigation by Miller (2002). Findings from this study of twelve nonprofit boards show that board members carry out the monitoring function in relation to their personal or professional competencies rather than considering measures that would show progress in achieving mission related goals and objectives, examples given include accountants concerning themselves with organising financial documents and lawyers focusing on legal or contract issues, this is at the expense of aspects related to progression of goals and initiatives that are related to the organisations mission.

The literature above has highlighted the existence of issues surrounding clarity and perception of board roles, and also highlighted instances where prescribed roles are not always carried out (Leatherwood & O'Neal (1996); Cornforth & Edwards (1999). Parker (2003) also uncovered ongoing differences of opinion between the executive and the non-executive directors as to the appropriate role of the board on his study of community welfare organisations. Other studies consider these issues in more depth, Widmer (1993) considers role conflict, role ambiguity and role overload in US human service organisations and finds that there are differences in the way that board members perceived or carried out their roles. Three situations were found to exist: some board members carried out the role of trustee which is described in the prescriptive literature; some members also carry out additional roles alongside this trustee role, these were identified as the role of worker which was the most dominant role, an expert, representative and figurehead; and some members carry out other roles to the exclusion of their trustee role. Similar conflicts are found to exist by Harris (1989) as a result of gaps between formal roles outlined in organisational documents, and role perceptions of staff and governing body members.

The above issue of role clarity is considered by McAdam and Gies (1985). This paper describes a situation where nonprofit board members are neither given, nor

develop, clear expectations of their roles and responsibilities. To assist clarification, three key areas of questions are proposed for members to ask within their first year of joining a board: firstly, administration and management information such as ascertaining whether the objectives of the organisation are stated in writing; secondly, questions about fiscal information such as whether the financial statements of the organisation are both accurate and complete; and finally, ask questions about human resource management, such as whether there is an adequate human resource development program for the staff. It is proposed that this process of asking relevant questions will stimulate improved institutional behaviour.

This section has considered the role of charity Boards, and has highlighted the emphasis on the strategic role of the Board in both governance guidance, and the prescriptive literature. Although such roles are deemed important, a gap appears to exist between importance and fulfilment, in particular there is evidence that operational tasks are also carried out by the Board. The next section specifically considers this issue of the distinction between strategic and operational roles, and also how this relates to the relationship between the Board and management.

#### 2.3 Relationship Between the Board and Management

The distinction between strategic and operational activities presents itself as a fundamental challenge in all organisations (Baysinger & Hoskisson, 1990). The ability of the board to fulfil its strategic roles and responsibilities without straying into managements role of implementing the strategy is widely observed (Helmer, 1996; Ingley & Van der Walt, 2001; Lorsch, 1995; Parker, 2007). A recurrent theme of the literature on the role of charity Boards is that they should have a strategic focus, as discussed in the previous section. However, in charities, Boards are found to engage in activities of an operational nature (Leatherwood & O'Neal, 1996; Steane & Christie, 2001; Cornforth & Edwards, 1999), this can be alongside, or at the expense of such prescribed strategic roles (Widmer, 1993). An example of such a situation is highlighted in a study of Australian Nonprofits by Steane & Christie

(2001), they report they way that boards engage in day-to-day issues as well as strategic activities. Similarly, Parker (2007) reports this challenge in Australian nonprofits, where boards have a tendency towards 'micro-management'.

The involvement of Boards in operational/ day-to-day issues is reported to be a fundamental cause of tensions between the Board and management (Chait and Taylor, 1989; Taylor et al, 1996; Cornforth, 2003). Chait and Taylor (1989) describe a situation whereby instead of governing, boards concern themselves with administrative activities at the cost of issues crucial to the survival of the organization. A number of consequences arise from this, including; the loss of perspective of these trustees when they concern themselves with daily management tasks which detracts their attention away from the overall goals of the organisation. The heavy involvement of trustees in admin affairs also has a negative effect on administrators, leaving them feeling undermined and discouraged. In addition, the more time trustees spend on operational matters, the less they have to spend on important policy and strategy matters, they also become less knowledgeable about the organization as a whole which may result in missed opportunities and loss of credibility. Reasons given for this tendency towards management rather than governance for these trustees, included special interests that some trustees may have in specific parts of an organization. Additionally, CEOs are reported to purposely feed the Board management information in order to deflect the Board attention from specific questions. This is also reported to cause tensions between the Board and Management.

Similar tensions are highlighted by Taylor et al (1996), in order to combat such situations, it has been proposed that a more partnership approach to governance should be adopted (Taylor et al, 1996; McClusky, 2002). Although Taylor et al (1996) believe the work of the board should be strategic and not at a lower more operational level, they also advocate a more partnership orientated approach to governance between management and the board. They highlight a tendency for board members to deviate towards more low-level operational matters, with reasons for this occurrence including the presence of a strong chief executive who is reluctant to

fully engage with his board and the presence of board members who lack understanding of both the organisation and its context. To rectify this recurrent situation, Taylor et al propose a framework which emphasises strategic issues through focusing on the board and management working together and focusing on issues that are of key importance to the organisation, which they term the 'new work' of the board. They describe the 'new work' as having four basic elements; it focuses on issues which are key to the success of the organisation, it is results driven, has clear success measures and requires engagement from the internal and external constituencies of the organisation, these elements contribute towards increasing interest and participation among board members. They propose that the board and management should work together on identifying important issues, determining the agenda, and setting and implementing policy. The important thing here is that they should act on what matters rather than dividing issues into policy and administration as they argue that implementation of policy can have bigger consequences than formulation. In addition, committees should be structured around the strategic priorities of the organisation rather than functions and board meetings should be goal driven.

Similarly McClusky (2002) also challenges the prescriptive literature which tries to draw a line between the work of the Board, management and other staff (such as Carver, 2006). Although McClusky agrees that Boards should always govern, he also feels that they should share in managerial and sometimes operational activities. The author disagrees with the one size fits all governance structures and the proposal in this paper is that the division of roles and responsibilities is dependent on a number of factors, including: the size of the organisations budget, staff and Board; the number of volunteers and the roles they perform; the stage of the organisations life cycle; the level of trust and confidence between the Chief Executive and the Board; Executive transition; the presence of organisational crisis; and environmental factors such as the nonprofits relationship with primary funders.

The different perceptions of senior management and the board regarding the boards involvement in strategy are investigated by Siciliano (2008) and Green et al (2001).

Siciliano (op. cit) differences in the extent to which the board and top management rate the boards involvement in strategy. In particular, the board viewed its participation at higher levels at all stages of the strategic management process. Similar findings were apparent from a study by Green et al (2001), where board members also rated their involvement in strategy as higher than top management, here tensions were uncovered between the board and management with the board believing they lacked credit for their strategic participation.

#### 2.4 The Composition of the Board

The composition of the board of trustees is extremely significant to charitable organisations, the organisation must make sure that it has the right mix of people on the board to enable the organisation to achieve its mission. A number of studies have been undertaken which consider the impact of board composition on a number of aspects of the organisation.

One such study is that by Austin and Woolever (1991), who consider the board composition of a US international voluntary association. They find that the composition of the board was influenced by characteristics of the local community and that the membership of the organisation was related to the composition of the board. Membership tended to have a higher proportion of lower-status occupations than the board members. Racial minorities tended to represent an approximately equal percentage of membership and board membership and board membership was predominantly male. A positive relationship was found between minority membership and minority board composition and also between community status, status of members and status of the board members. Community and membership racial compositions are related to board characteristics and boards tend to be more representative of the community and membership in relation to race than relation to social class.

Siciliano (1996) investigates the diversity of board membership in YMCA organisations and developed an index of board member diversity. The findings showed that when board members had greater occupational diversity, this resulted in higher levels of social performance and fundraising results. Gender diversity compared favourably to the organisations level of social performance but there was a negative association for the level of funds raised. Higher levels of donations were linked with diversity in board member age. The importance of board representation was dependent on the type of performance being examined with the strongest impact on board diversity being on the organisations social performance. The findings suggest that boards with a greater diversity in member backgrounds and gender were able to keep their social agency purpose in the forefront. Regarding the level of donations, all composition variables were related to this measure. Board member occupation, age diversity and, to a lesser extent, board size appeared to enhance donations, gender diversity did not.

O'Regan and Oster (2005) conducted a survey of executive directors of New York nonprofits. They explored the way in which the functioning of nonprofit board members in the areas of philanthropy, monitoring and work – varies by several features of that board; namely, the independence of the board against director strength and the size of the board. The findings include that large board size and executive director control are both associated with positive performance features, as are long tenure and multiple board service.

Abzug and Galaskiewicz (2001) consider the inclusion of a specific group of individuals of the board; the recruitment of trustees with college education and/ or professional managerial occupations. The findings show that trustees with college education, managers and professionals continue to have significant representation on nonprofit boards. Also, many boards are also increasingly less exclusive with respect to gender, race and religion. Some select nonprofit boards, however, continue to be dominated by different gender, racial and religious identities, suggesting that nonprofit boards also serve the purpose of different identity and/ or interest groups in the community.

Other papers specifically focus on the impact which business people have on nonprofit boards. Austin (1998) conducts a series of surveys of Harvard MBA graduates and finds that business people have a positive impact on nonprofit boards; nonprofits feel that they benefit from both their expertise and their distinct managerial and business perspective. Bowen (1994) also advocates the inclusion of business leaders on nonprofit boards; believing that they have much to learn from those individuals. In particular; routine use of benchmarking and monitoring of results against planned outcomes. A similar view is taken by McCambridge (2004). However, both Bowen and McCambridge highlight potential pitfalls in the inclusion of business people as charity trustees, this will be considered in greater detail in the next chapter.

#### 2.5 Board and Organisational Effectiveness

The challenges to the governance of nonprofit organisations have resulted in a number of studies which aim to understand what factors contribute to the effectiveness of the board and the organisation. Relatively few studies consider the relationship between both board and organisational effectiveness, however, two studies identified which do investigate this relationship are Green & Griesinger (1996) and Bradshaw et al (1992). Green & Griesinger (1996) explore this relationship between board and organisational effectiveness via a series of interviews and questionnaires with board members and CEOs of 16 social service organisations. The findings report the existence of a significant relationship between board performance and organisational effectiveness; the activities of the board which were strongly related to organisational effectiveness were found to be policy formulation, strategic planning, program monitoring, financial planning and control, resource development, board development and dispute resolution. Bradshaw et al (1992) consider the impact of board structure and processes in organisational and board effectiveness. Through a survey of Canadian nonprofits, they report a positive relationship between the perception of board effectiveness and the use of the most

widely advocated prescriptions on how a nonprofit board should operate, in particular, strategic planning was found to account for 30 percent of the variance in perceptions of board effectiveness.

A similar study to Bradshaw et al (1992) is that of Cornforth (2001) who also considers the impact of board structures and practices, but in relation to board effectiveness on nonprofits in England and Wales. The findings of this study suggest that board effectiveness is dependent on whether: board members have the time, skills and experience to do the job; clear roles and responsibilities; the board and management share a common vision of how to achieve their goals; and the board and management periodically review how they work together. Board effectiveness is also investigated by Brown (2005) who looks specifically at six dimensions of effective board performance, as suggested by Taylor et al (1991), namely; contextual, political, strategic, analytic, educational and interpersonal. The results suggest that strategic contributions from the board are more robust in organisations with higher financial performance. In addition, organisations that are judged to be higher performing also reported having high-performing boards across all dimensions. In particular, the interpersonal dimension provided a unique explanation of judgements of organisational performance.

Herman and Renz (1997, 1998, 2000) and Herman et al (1997) have reported on a number of findings from their study of stakeholder judgements on nonprofit organisational effectiveness. Herman and Renz (1997) report on findings from a series of focus groups with practitioner-experts and questionnaires which were sent to key individuals. The two main conclusions from this study were that: practitioner-experts do not rely on bottom line outcomes as meaningful indicators of objective organisational effectiveness, preferring instead evidence of following correct procedure or doing things right; and all stakeholders seem to use some socially constructed evidence of board effectiveness in forming judgements of organisational effectiveness. Board effectiveness is the most important determinant of organisational effectiveness for all stakeholders. Herman et al (1997) investigate the relationship between the extent to which nonprofit boards use prescribed practices

and stakeholder judgements of the effectiveness of those boards. The findings suggest that there is wide variation in the use of the prescribed board practices, that judgements of board effectiveness can differ substantially, and that chief executives' judgements of the effectiveness of their boards is moderately related to the extent of use of recommended board practices.

Herman and Renz (1998) focus on a subset of especially effective and less effective NPOs from a larger sample and find that the especially effective organisations have more effective boards (as judged by various stakeholder groups), have boards with higher social prestige, use more practitioner-identified correct management procedures and use more change management strategies. Herman and Renz (2000) review evidence in support of the hypothesis that NPO's effectiveness is related to the effectiveness of their boards of directors. It also asks whether various recommended board practices and processes affect board effectiveness. The study focuses on a subset of especially effective and less effective nonprofit organisations from a larger sample. The results show that the especially effective organisations have more effective boards and that the more effective boards use significantly more of a set of recommended board practices. The results also show that nonprofit organisations using more of the prescribed board practices are also more likely to use other correct procedures.

Nonprofit organisational effectiveness is considered by Herman and Renz (1999) and (2008). Herman and Renz (1999) review the literature on organisational effectiveness and nonprofit organisation effectiveness to advance six thesis about the effectiveness of public benefit charitable NPOs, namely: nonprofit organisational effectiveness is always a matter of comparison; nonprofit organisational effectiveness is multidimensional and will never be reducible to a single measure; boards of directors make a difference in the effectiveness of NPOs, but how they do this is not clear; more effective NPOs are more likely to use correct management practices; nonprofit organisational effectiveness is a social construction; and programme outcome indicators as measures of NPO effectiveness are limited and can be dangerous. This paper is advanced by Herman and Renz (2008), this includes the initial five thesis

plus an additional four: it is unlikely that there are any universally applicable 'best practices' that can be prescribed for all NPO boards and management; responsiveness is a useful overarching criterion for resolving the challenge of differing judgements of NPO effectiveness by different stakeholder groups; it is useful to differentiate among different types of nonprofit organisations in assessing the merits of different approaches to understanding nonprofit effectiveness; and level of analysis makes a difference in understanding effectiveness, and it is important to differentiate effectiveness at program, organization and network levels.

Herman and Tulipana (1985) conduct a study of board-staff relations and perceived effectiveness in nonprofits, with a specific emphasis on the internal role of the board. The main findings here were that: participants experience less actual influence than they think they should have; board members' influence is positively related to the frequency of board meetings and the extent to which board members feel informed of their duties, though neither is very strong.; ratings of organisational effectiveness are positively related to board member ratings of staff educational efficiency and the extent to which board members feel informed of their duties; and at the organisation level of analysis, organisational effectiveness is strongly related to the amount of total influence in the organisation. The relationship between board effectiveness and trustee motivation in Colleges is investigated by Taylor et al (1991), they find the motives of trustees to be both institution-specific and institution-centred, coming from a deep affection and a sense of connection to the College; this is found more among the members of effective rather than ineffective Boards.

The relationship between the structure and composition of the board and the effectiveness of the organisation is investigated by both Provan (1980) and Callen et al (2003). Provan (1980) looks at the importance of an externally powerful board of directors to the effectiveness of human service agencies. Findings – drawn from published data - suggest that by having a 'powerful' board of directors, an agency will be effective in terms of its ability to obtain large amounts of funding if these amounts are consistent with what has been received in recent years. However, as a means of enabling an agency to be effective in its efforts to acquire funding, the

results of the study strongly suggest that a powerful board of directors may be less important than previously believed. Callen et al (2003) investigate the relationship between nonprofit board composition and organisational efficiency. The results here show a significant association between the presence of major donors on the board and indicators of organisational efficiency.

Board effectiveness in the Scottish third sector is considered by Kakabadse et al (2009), who sought perceptions of the effectiveness of the performance of the board, the chair, and the chief executive, the main areas reported to inhibit board effectiveness were the relationship between the chair and the chief executive, the role of the chair in encouraging peer appraisal of board members and evaluating the performance of other board members; and the chair encouraging feedback of their own performance.

There are also studies which seek to develop tools to assess effectiveness; two such studies are Jackson & Holland (1998) and Brown (2007). Jackson and Holland (1998) develop the Board Self-Assessment Questionnaire (BASQ) to assess performance in six areas that previous research has shown to characterise highly effective boards, namely; contextual, educational, interpersonal, analytical, political and strategic. Data from board members was used to examine the instruments reliability, validity and sensitivity, including the relationship between board scores and selected indicators of the organisations financial performance. The results indicated that the BSAQ is a sound and useful tool for evaluating board performance. Similarly, Gill et al (2005) develop a Governance Self-Assessment Checklist; this is designed to identify strengths and weaknesses in the governance of organisations, to educate board members about the essentials of good governance and improve the organisations governance practices. The instrument consists of 144 items that assess the main factors in the performance of a nonprofit board of directors that are thought to influence the effectiveness of the organisation.

The impact of board development practices are investigated by Holland and Jackson (1998) and also Brown (2007). Holland and Jackson (1998) conduct a study of ten

nonprofits that did participate in development interventions and compared them with fourteen nonprofits which did not receive interventions. The intervention group showed significant improvements in board performance, the other group did not. Brown (2007) develops a model to determine if using recommended recruitment, board member orientation, and evaluation practices resulted in more competent board members and if the presence of these board members led to better board performance. Both CEOs and board chairs of nonprofits were surveyed and the results were found to support the contention that board development practices lead to more capable board members, and the presence of these board members tends to explain board performance. Brudney and Murray (1998) consider intentional efforts to improve boards and how many boards accept advice and intentionally change the way they operate; they surveyed Canadian nonprofits. The results show that threequarters of the sample did report that they had made intentional efforts to change the way they operate. In particular, this was due to dissatisfaction with roles and responsibilities, committee structures, relationships with management and meeting dynamics.

#### 2.6 Conclusion: Governance in Charities

This chapter has highlighted a number of key aspects surrounding the governance of charitable organisations. In particular, the focus was on what was classified as 'internal aspects' of governance (Hyndman & McDonnell, 2009), four key issues were considered, namely the role of the board, the relationship between the board and management, board composition, and board and organisational effectiveness.

In relation to the role of the board, importance of the strategic role of the board was emphasised in both professional guidance and academic literature (Axelrod, 2005; Cornforth & Edwards, 1999; Houle, 1990). There is a widely available literature outlining what the role of the board should be, however there are reports that such prescribed roles are not always carried out, with issues of role conflict and ambiguity apparent (Cornforth & Edwards, 1999; Leatherwood & O'Neal, 1996). Lack of

clarity and understanding of the role of the board were also reported (Harris, 1989; McAdam & Gies, 1985; Parker, 2003; Widmer, 1993). Role importance has received attention (Brown & Guo, 2010), with gaps identified between role importance and fulfilment (Green et al, 2001; Inglis et al, 1999). Factors which shaped board roles were found to include the environment and organisational complexity (Brown & Guo, 2010) and size – with the boards of larger organisations being more involved in strategy (Cornforth & Edwards, 1999).

The division of strategic and operational tasks was seen as a fundamental cause of tensions between the board and management (Chait & Taylor, 1989; Cornforth, 2003; Taylor et al, 1996). Such a distinction between strategy and operations is seen as a fundamental tension in all organisations (Baysinger & Hoskisson, 1990). Despite the emphasis on the boards strategic role, charities were reported to engage in activities of an operational nature (Cornforth & Edwards, 1999; Leatherwood & O'Neal, 1996; Steane & Christie, 2001). To combat such tensions arising from difficulties in role definition, a more partnership approach to governance between the board and management is suggested by Taylor et al (1996) and McClusky (2002).

With regard to the composition of the board, this has been investigated in relation to the performance of the organisation, with a particular interest on the representation of business people on nonprofit boards. Board composition was found to be related to the membership of the organisation (Austin & Woolever, 1992). The impact of board member diversity was also considered – boards with a greater occupational diversity in member backgrounds and gender were able to keep the main focus on the social purpose of the organisation (Siciliano, 1996). Attention has also been paid to the recruitment of business people onto nonprofit boards, Abzug & Galaskiewicz (2001) reported significant representation of trustees with college education and/ or professional management occupations. However, they also found a set of nonprofits still dominated by different gender, racial and religious identities, suggesting that boards also serve the purpose of different community interest groups. Other papers focus on the impact of business people on nonprofit boards, with a positive impact of Harvard MBA graduates on nonprofit boards reported by Austin (1998). Bowen

(1994) and McCambridge (2004) also consider the inclusion of business people on nonprofit boards.

Finally, the relationship between board and organisational effectiveness was considered, with a number of studies reporting a positive relationship between the performance of the board and the organisation (Green & Griesinger, 1996; Bradshaw et al, 1992). Other studies look specifically at what factors impact on the effectiveness of the Board (Brown, 2005; Cornforth, 2001). Herman & Renz (1997) consider the views of stakeholders, who consider board effectiveness to be the most important determinant of organisational effectiveness.

The above consideration of the literature on governance in nonprofit organisations highlights a number of key issues which are present in this area. However, with the exception of Cornforth (2001 & 2003) and Cornforth and Edwards (1999), this research is focused outwith the UK. Furthermore, the studies that do consider the UK are focused specifically on England and Wales, and only one study (Kakabadse et al, 2009) considers Scotland. This clearly highlights a significant gap in research on charity governance, and the need to consider whether such issues are apparent in the Scottish context.

The next chapter will consider in detail one specific aspect outlined above, that of the impact of business professionals on the governance and management of these organisations.

#### **CHAPTER 3**

### THE IMPACT OF BUSINESS EXPERTISE ON THE GOVERNANCE AND MANAGEMENT OF CHARITIES

#### 3.1 Introduction

The previous chapter considered issues which related to internal aspects of governance in charitable organisations. Board composition was identified as an important issue, with a specific aspect of this, which has an impact on both the governance and management of these organisations, being the impact of business practices and expertise on charities. This is a highly topical and highly contentious issue (Landsberg, 2004; McFarlan, 1999), with a divergence of views in terms of the impact which this has on individual organisations, and the sector as a whole. However, unlike other aspects of charity governance, the issue of the impact of business practices and expertise on governance has received significantly less attention, despite its topicality. In terms of this business influence on charities, significantly more attention has been paid to the adoption of business practices in the management of these organisations, and the increasing commercialisation of the sector, rather than the effect it has on governance practices and structures. There is a limited amount of empirical data on these issues, and as with the literature on charity governance, there is also a lack of studies which focus specifically on the UK, particularly Scotland.

Historically, the management and administration of charities was derived from the specific ethos of the organisation (Bush, 1992). However, the complex world in which modern day charities operate has resulted in increased competition from both government, the for-profit sector and also from other charities in a bid to raise the funds required to survive (Lapsley, 2004; Frumkin & Andre-Clark, 2000). In addition to the competition from private donations and government funding, there is

also increasing expectations from the general public and the government in terms of what the charities deliver and also that they account for their actions. This has increased the pressure on charities to be managed in a more business-like way (Dart, 2004; OSCR, 2008), adopting various business techniques as they strive to become more efficient in a bid to survive. However, this issue of the adoption of private management techniques by charities is one which is steeped in controversy. Some scholars believe that this is the only way that these organisations will survive the growing pressures and expectations of the sector (Bradley et al, 2003), others however, believe that it could have severe negative impacts on the sector and on the distinctive ethos which these organisations are built on (Landsberg, 2004).

This chapter will consider the previous work on the influence of business practices and expertise in charities in two ways; firstly, the impact on governance will be addressed, by looking at papers which consider the impact of business expertise on charity boards, and also the adoption of a more business-like approach to governance. Secondly, studies which examine the impact of the adoption of business practices in charities will be considered to further explore the impact they have at various levels within these organisations.

### 3.2 The Impact of Business Expertise on Charity Governance

Despite the universal acknowledgement that the adoption of business practices in charities is a significant issue, there is a considerable lack of research on the impact this has on governance, in particular the impact that business professionals have on charity boards, which, coupled with the increasing emphasis for charities to recruit these individuals to their boards (OSCR; 2008, Abzug & Galaskiewicz, 2001; Cornforth & Edwards, 1999), is surprising.

This issue of business people joining charity boards is considered by Bowen (1994), who ponders the divergence of performance of business executives on charity boards. Several factors are identified here which may imping on the success of these

individuals in their role as charity Trustees, such examples include, that professional staff in charities may be quick to dismiss business professionals as they may feel they will not understand the unique qualities and professional norms of the organisation, or may be insensitive towards them. Another problem identified is that there may be an amount of hostility directed towards business professionals from those within the organisation. Although in practice, this paper appears to highlight a clash between business professionals and those from within the charity, Bowen (op cit.) is still supportive of the idea of business people serving on the boards of charities, and notes that these organisations have a considerable amount to learn form their for-profit counterparts and the operations of corporate boards, calling for the adoption of certain business practices such as the routine use of benchmarking and monitoring discrepancies between results and planned outcomes. Another paper which looks at the impact of business professionals on charity boards is McFarlan (1999), who highlights the significant differences between nonprofit and for-profit boards; an issue raised here is that business professionals underestimate and are not fully aware of what being a charity trustee actually involves, as a result, this lack of understanding of the role, and it's time-consuming nature results in a high board turnover. The success of business people on charity boards, according to McFarlan, requires them understanding the differences between the nonprofit and private sectors and the realisation that not all their skills and expertise are appropriate for application in the nonprofit sector.

Dees (1998), in considering ways in which nonprofits can effectively become more business-like, highlights the importance of recognising that board members from business are a valuable resource. Dees notes that if nonprofits are to become more business-like, managers must be trained in business methods, although he still recognises the problematic nature of direct transfer:

One way to gain such training is to reach out for help. Nonprofit managers can begin in their own backyards by finding more effective ways to draw on board members with relevant business experience. The resulting exchange will be a learning experience for both parties. Business board members are often an underutilised source of management expertise, and they need coaching and coaxing to adapt

their business frameworks to the context of a social enterprise (Dees, 1998, p.66)

An empirical study of this issue is conducted by Austin (1998), who considers the involvement of business-people on charity boards in the USA. In considering the benefits to nonprofits, it was found that nonprofits benefited significantly from the expertise of business people, and their distinct managerial and business perspective, in addition, 80% of business people were found to be 'as or more effective' and 90% 'as or better prepared' than other board members. There was a tendency for them to be viewed differently to their counterparts, due to their skills, knowledge and experience, and the way they enriched board discussions.

The above studies – although highlighting potential issues with the involvement of business people on charity boards – are largely positive about their involvement and the impact that it has. In contrast, there are also studies where this is not deemed to be the case. An example of this is Landsberg (2004), who conducts a case study on a US mental health centre, and highlights that the result of more business people being recruited onto nonprofit boards is a greater emphasis on financial concerns in place of issues related to mission. This point can be seen from Siciliano (1996) who considers the relationship between board member diversity and organisational performance. Here, an example is given where the Board is considering a long-standing YMCA program, Siciliano describes the discussion:

The program had not been cost-effective for several years and was beginning to drain on other sources of revenue. Several board members with business backgrounds were strongly opposed to its continuance. However, viewpoints of other board members tended to stress the program's benefit of building family relationships. After hearing both sets of arguments, the board voted to continue the program for another year. (Siciliano, 1996 pp. 1318).

This example highlights Landsbergs concern that business people tend to focus on finance, however it also shows that a variety of knowledge and views can help organisation make difficult decisions. Similarly, Miller (2002) reports that board members carry out the monitoring function relating to personal/ professional

competencies rather than considering measures that would show progress in achieving mission related goals and objectives e.g. accountants concerning themselves with financial documents.

The impact of legislative requirements, and private sector philosophies has been found to have an impact on the strategic and compliance role in nonprofits. In a study of Australian nonprofits, Steane and Christie (2001, p.48) note that the boards of nonprofits 'are expected to mimic their corporate peers'. This is viewed as a result of legislative requirements pressuring nonprofits to place more emphasis on compliance than strategy. This is similar to findings of Cornforth and Edwards (1999) who find that the compliance role is emphasised by the legal responsibilities of the board. In the case of a further education college, they highlighted how new legislation and regulation had placed a requirement on colleges to recruit people with business experience onto their boards. As a result there was a growing view of 'the College as a business' (p. 358) with governance changing from one which emphasised local political accountability to a more managerial view of governance. There was also evidence of a desire to mimic businesses.

Parker (2003) considers the boards role in the financial management strategy of a community welfare organisation, here governance is predominantly financially focused in terms of both strategic orientation, and predominance of boardroom discussions. Parker (op cit) also considered the roles of those with accounting and non-accounting backgrounds, finding no distinction between their focus on financial versus operational strategies, attention to monthly budgetary control reports, level of participation in financial strategy deliberations, or general emphasis upon financial aspects of board meeting agenda. The aspects of accounting which were identified as key to directors' financial management strategising were central, whether they had an accounting or non-accounting background.

An additional piece of empirical research, which takes a negative view on such business influences is that by Alexander and Weiner (1998), they move away from the specific focus of business people on charity boards and consider the adoption of a for-profit model of governance for charities, the 'corporate governance model'; this solution is not seen as viable for the sector.

The above section has considered the existing literature which relates to the influence of business people and practices on charity governance. This is clearly a topical issue, but one in which there is a serious lack of empirical research. The next section will look more broadly at the adoption of business practices by charities and the increasing commercialisation of the sector.

# 3.3 The Adoption of Business Practices by Charities

A number of commentators consider the broader impact of the adoption of business practices on charitable organisations. There are a many who provide a balanced overview of such practices and their impact on the sector (Foster & Bradach, 2005), some who strongly advocate this approach in the sector (Bradley et al, 2003) and others who are strongly against such practices (Landsberg, 2004). This section will outline each of these views in turn.

The changing environment in which nonprofits operate has led to the increasing tendency for them to adopt more business-like practices. Historically, the management and administration of nonprofit organisations was steeped in the tradition of charity and philanthropy, where nonprofits were managed in a way that would encourage volunteerism and collective engagement of community problems (Bush, 1992). However in todays environment, the nonprofit sector faces a number of challenges which threaten these traditions.

Dees (1998) highlights a number of reasons why nonprofits are turning to more commercial sources of funding: firstly, there appears to be a growing acceptance of these practices via a new pro-business zeitgeist; many nonprofits want to be able to deliver goods and services in a way that does not create dependency in their communities; commercial activities are seen as more reliable sources of funding than

donations and grants; and there is increasing opportunities to make money on a more-commercial basis, where competition is increasing for traditional charitable funding.

In considering the debate surrounding the adoption of business management practices in nonprofit organisations, one of the most significant arguments of those who oppose such practices is the consideration of the nature of the sector. Jeavons (1992) highlights the importance of relating values to the management of these organisations and notes that the approaches to management that nonprofits employ send a message and they therefore need to be conscious about what that message is. Jeavons stresses the importance of understanding the differences between nonprofits, business and government in terms of managing the organisation; he notes it is essential to be aware of and to understand the context of the sector. He stresses that the fundamental difference is that these organisations are led by their values and that the effectiveness of promoting the values of the organisation will be dependent on the way in which the organisation operates; it must be seen to honour basic human, social and professional values. Jeavons highlights three areas which he sees as essential in terms of how the structure and functions of the organisation reflect the commitment of the organisation, namely: the honesty and accountability of the organisation; concern for the public good, not just immediate stakeholders and also concern for the personal welfare of employees and volunteers. Jeavons notes that the importance of these three areas means that it is not acceptable for nonprofit managers to think like other managers – especially about their organisations relationship to the public - they also must be able to analyze some of the issues facing their organisations in historical, cultural and moral terms that can go largely unconsidered in other managerial settings. If nonprofit managers fail to take into account these issues then the privileges and support from the public may be withdrawn which may threaten the existence of the organisation and in order to communicate their values they must be credible in the eyes of the public, this can only be achieved by honouring societies expectations.

The divergence in attitudes of the impact of business on the charity sector has been noted above. There are a number of commentators which view this influence as positive for the charities sector; in considering the possibilities that could evolve for the sector should they become more business-like, Bradley et al (2003) estimate that the US charities sector could increase income by \$100 billion if they made such changes. In terms of the increasing commercialism in charities, Young (1998) attempts to understand the growing trend of commercial practices and sales income in national nonprofit social service associations in the US. Through a series of case studies, Young highlights that these commercial initiatives do make direct contributions to the mission of the organisations, as well as producing revenue; although this does tend to be helped by missions being defined in very broad terms. Furthermore, necessary steps were found to be taken by the leaders of these associations to avoid engaging in activities that may damage their association's reputation and their abilities to pursue their missions. This is in contrast to Bush (1992) and Landsberg (2004) who highlight a detrimental effect on the nonprofits image as a result of the adoption of business practices.

In an exploratory study of the effects of commercialization on human service nonprofits, Guo (2006) looked at whether commercialization had a positive impact on the organisations. It was found that high levels of commercial income did contribute significantly to an organisations self-sufficiency, its ability to attract and retain staff, and it's reputation. However it did not contribute to its ability to attract donors and volunteers, mission and service delivery.

In contrast to the above views, there are a number of commentators who approach the issue of charities adopting business practices with a greater degree of scepticism. One such example, is that of Foster and Bradach (2005), they highlight the increasing trend for managers of US charities to launch commercial ventures, in a bid to shake off the tag of 'passive bureaucrats' and instead be viewed as 'active entrepreneurs'; such schemes also tend to be supported by board members, especially those with a business background. In addition to this, Foster and Bradach (2005) highlight additional pressures from foundations and other funders, who they note have been

'zealously urging nonprofits to become financially self-sufficient and have aggressively promoted earned income as a means to 'sustainability' (p. 94), contributing even more pressure on nonprofits to succeed in launching such commercial ventures. However, Foster and Bradach (2005) cast doubt over the viability of such schemes; they note that earned income is only a fraction of funding in most nonprofit domains, that few ventures do actually make money, and that nonprofits tend to be a lot more optimistic about these ventures than they should be. Greater cause for concern can be seen in their view that:

commercial ventures can distract nonprofits' managers from their core social missions and, in some cases, even subvert those missions. We're not saying that earned-income ventures have no role in the nonprofit sector, but we believe that unrealistic expectations are distorting managers' decisions, ultimately wasting precious resources and leaving important social needs unmet' (Foster & Bradach, op cit. p. 94).

However, Foster and Bradach do conclude that should a venture allow the enhancement of a nonprofits mission, while allowing it to recover some of the costs, the venture may be feasible even if it never breaks even.

A similar cautionary view is taken by Dees (1998), who highlights the same shift towards commercial funding as Foster and Bradach (2005). Dees also highlights issues with a reliance on commercial funding, such as the danger that it may shift focus away from the nonprofits original mission, potential for clashes of culture, lack of skills to succeed in commercial markets, and possible negative impact on the relationships which the nonprofit has with its local community. However, Dees does believe that there is a place in the nonprofit sector for more commercial approaches to funding and highlights that if programs do exist that can generate income on their own, this would allow the allocation of other philanthropic funds to activities that have more requirement for subsidies. The fundamental challenge for nonprofits here is the ability to adopt a 'financial structure that reinforces the organisations mission, uses scarce resources efficiently, is responsive to changes, and is practically achievable, Dees concludes that ultimately:

commercial operations will not – and should not – drive out philanthropic initiative. Many worthwhile objectives cannot effectively be pursued by relying on market mechanisms alone (Dees, 1998, p. 67).

Other studies, such as Myers and Sacks (2003) and Harris (2001) consider the influence that business practices can have on the nonprofit sector, however, both stress that these practices must be adapted in some way to account for the complex nature of the sector. For example, Myers and Sacks problematise this issue, and note that:

It is important to consider that although the voluntary sector can learn and interpret systems for improved organisation development from the private and public sectors, there are difficulties in importing ideas wholesale without interpreting the context and the culture of the sector and taking into account the internal relationships within individual organisations (Myers and Sacks (2003) p. 289).

The importance of adapting these techniques is echoed by Harris (2001) who notes that due to the complexities of nonprofit organisations:

...business management techniques and theories developed for the forprofit or governmental sectors are likely to require substantial adaptation before they can be applied to the voluntary sector (Harris, 2001, p. 102).

There are a number of commentators which after examining the impact of various business practices adopted by charities, conclude that they do not have a place in the voluntary sector. Bush (1992) notes the danger inherent with an over-reliance on business practices being a shift away from the mission as the primary focus of activity for the nonprofit. Dangers of increasing competition are noted as the basic philosophy of competition is secrecy and information which is ultimately bad for the sector. Another danger of this business-like approach is that the public may perceive the organisation in a different way. Bush calls for parameters to be set which would allow the sector to borrow, adapt, and use business techniques without undermining the unique character and traditions of these organisations.

Pressures for nonprofits to become increasingly more business-like is considered by Ryan (1999) who looks at the increasing level of competition which US social service nonprofits face from for-profit firms who now compete for government contracts; and the ability of the nonprofit to adapt to this changing environment without damaging their unique character. The resulting actions which nonprofits have taken in order to compete with for-profit firms has included subcontracting, partnership and at the extreme end, conversion to for-profit status; Ryan (op cit.) highlights his concern that:

the distinctions between these organisations will continue to be blurred. It is precisely this blurriness that could endanger nonprofits as they are forced to adapt — and perhaps abandon — the very qualities that enabled them to advance social causes in the first place' (Ryan, 1999, p. 134).

Similarly, Backman and Smith (2000) also consider the impact that more commercial forms of funding has on US nonprofits. They highlight the changing nature of these funding changes; traditional commercial sources of funding such as fees for products and services have made way for business income which is not mission-related, where government funding has shifted from grants to contracts, and where traditional philanthropy has shifted towards strategic philanthropy, where corporations are using philanthropic funds to contribute to corporate profits. Backman & Smith (2000) conclude that although this trend towards a more commercial approach to raising income may have a positive impact in terms of increased revenues and financial stability and may make the nonprofit a 'healthier organization'. The opposite effect may be seen in terms of the relationships the organisation has with key stakeholders and may actually undermine their capacity to create social capital, therefore resulting in 'less healthy communities'.

The views of Ryan (1999) and Backman & Smith (2000) are also echoed by Eikenberry and Kluver (2004). They highlight four ways in which nonprofits are becoming more business-like; increased generation of commercial revenue, shift from grants to contracts and vouchers which results in nonprofits behaviour becoming increasingly competitive; emergence of venture philanthropists who treat charities like potential business investments; and social entrepreneurship resulting in

the reluctance of nonprofits entering into unprofitable mission-related activities. The impact of these practices is seen as compromising the impact on the nonprofits contribution to civil society in terms of their key roles as value guardians, service providers and advocates and builder of social capital.

However, possibly the most extreme view of the impact of charities adopting business practices comes from Landsberg (2004) who believes that:

the business practices the nonprofit embraces to assure its survival threaten to undermine its culture, mission, and public image. In an effort to save its bottom line, the modern nonprofit risks losing its soul. (Landsberg, 2004, p. 1).

Landsberg believes that the for-profit business model, which places emphasis on quality, efficiency, flexibility, innovation and a focus on the financial bottom line cannot be directly transferred to the nonprofit sector; these organisations have more complexities than their for profit counterparts including the need to manage a diverse range of constituencies and stakeholders. These complexities result in multiple bottom lines rather than the single profit motive of businesses, therefore, Landsberg argues that these for-profit models may actually be too unsophisticated for nonprofits. He investigates the impact of a more business-like approach in a large US community mental health centre and finds that the introduction of business-like practices such as; restructuring, recruitment of business professionals, introduction of new financial management techniques and IT systems, the adoption business practices in all areas of administration and an emphasis on accountability and profitability, led to a number of problems within the organisation. In terms of the board, they were unsure of their role, there was low morale among clinical staff who were uncomfortable with the changes, and the public image also suffered with implications for donations, there were also questions over whether finance was overshadowing mission.

# 3.4 Conclusion: The Impact of Business Expertise on the Governance and Management of Charities

This chapter has considered the main issues surrounding the impact of business people and practices on the governance and management of charities. The impact of business expertise and practices on the management and governance of charities has presented itself as a highly contentious issue (Landsberg, 2004; McFarlan, 1999) with differing views on the impact this has on both individual charities, and the sector as a whole. However, unlike other aspects of charity governance presented in chapter 2, the impact of business practices and expertise at board level has received considerably less attention. Within this area, the focus has tended to be on the broader adoption of private sector practices in the management of charities, and the increasing commercialisation of the sector.

With regard to governance, benefits to be gained from the presence of business people on charity boards are recognised by Bowen (1994), Dees (1998) and McFarlan (1999) – although importance of the recognition of differences between the sectors is also highlighted. One of the few empirical studies in this area by Austin (1998) specifically considers the involvement of business people on charity boards in the USA, with a positive impact reported. In contrast, Landsberg (2004) cautions that individuals with business expertise focus on financial concerns over mission, with similar situations reported by Miller (2002) and Siciliano (1996). Of the few studies that focus on the adoption of the corporate governance model in nonprofits, Alexander and Weiner (1998) report that this is not transferable to the nonprofit sector.

The broader impact of business practices is more widely considered, with differing opinions as to their applicability. For example, the positive impact of such practices is reported by Bradley et al (2003), Dees (1998) and Young (1998). However, the problematic nature of such practices are reported by Bush (1992), Backman & Smith (2000), Eikenberry & Kluver (2004), Foster & Bradach (2005), Harris (2001), Landsberg (2004), Myers & Sacks (2003) and Ryan (1999).

The above consideration of the literature on the impact of business practices and expertise in the governance and management of nonprofit organisations presents this area as one of considerable complexity. However, such research is focused outwith the UK. This clearly highlights a further gap in research on charity governance, particularly relating to the impact of business practices and expertise in this area within the Scottish context.

After consideration of the extant literature of both charity governance, and the specific impact of business practices and expertise of the governance and management of charities, the following chapter will develop a theoretical framework for this study.

#### **CHAPTER 4**

#### CONCEPTUAL FRAMEWORK: SOCIAL CONSTRUCTION

#### 4.1 Introduction

The previous two chapters have highlighted the various issues that charities face in Governing their organisations, and also the tensions which exist when business people become involved in charities, particularly on their Boards. This chapter will look at Social Construction and how it can help to interpret and understand the behaviour of charitable organisations in relation to these issues. This chapter is structured as follows: (4.2) an overview of social construction, and the key concepts of legitimation and isomorphism will be given; (4.3) examples of previous studies which have drawn on these ideas will be discussed before turning to (4.4) examples specific to the nonprofit sector. Finally, (4.5) the importance of these ideas in relation to this study will be outlined.

### **4.2 Social Construction**

This section focuses on three influential contributions to the literature on social construction; Berger and Luckmann (1967), Meyer and Rowan (1977) and Di Maggio and Powell (1983). The seminal contribution to the literature on social construction was made by Berger and Luckmann (1967). The fundamental argument which they put across is that 'reality is socially constructed and that the sociology of knowledge must analyse the process in which this occurs' (Berger and Luckmann, 1967, p.13). Within this, 'reality' is defined as a quality relating to the phenomena that we recognise as having a being independent of our own volition (we cannot 'wish them away'). Knowledge is defined as the certainty that phenomena are real and that they possess specific characteristics. Although they highlight that there have been numerous definitions of the sociology of knowledge in terms of its nature and scope, there has been a general consensus that this sociology of knowledge is

'concerned with the relationship between human thought and the social context within which it arises' (Berger and Luckmann, 1967, p.16).

A more detailed description of this concept is outlined below:

The reality of everyday life maintains itself by being embodied in routines, which is the essence of institutionalization. Beyond this, however, the reality of everyday life is ongoingly reaffirmed in the individual's interaction with others. Just as reality is originally internalized by a social process, so it is maintained in consciousness by social processes. These latter processes are not drastically different from those of the earlier internalization. They also reflect the basic fact that subjective reality must stand in a relationship with an objective reality that is socially defined (p. 167, Berger and Luckmann, 1967).

As described above, a key aspect in understanding the way in which reality is socially constructed is through the process of institutionalization, which aides the understanding of the way in which social order emerges, is maintained, and is transmitted. Preceding this process of institutionalization is habitualization, this occurs when an action is so frequently replicated, that it evolves into a pattern, as a result, future performances of that action require a reduction in effort, however, the performer still holds this as the same pattern. A resulting implication of this is that future performances of the action will be identical in terms of manner and amount of effort required; they will become embedded. There is also a psychological aspect attached to habitualization, Berger and Luckmann (1967) term this 'psychological relief' (p. 71) whereby choices are reduced to one and the burden of decision making is removed, time and energy saved through habitualization facilitates deliberation and innovation.

Institutionalization is defined more specifically by Berger and Luckmann as follows:

Institutionalization occurs whenever there is a reciprocal typification of habitualized actions by types of actors. Put differently, any such typification is an institution. What must be stressed is the reciprocity of institutional typifications and the typicality of not only the actions but also the actors in institutions. The typifications of habitualized actions that constitute institutions are always shared ones. They are available to

all members of the particular social group in question, and the institution itself typifies individual actors as well as individual actions (p. 72, Berger and Luckmann, 1967).

In essence, the key to understanding what constitutes an institution are shared characterizations of these habitualized actions by types of actors.

Neo institutionalists draw on these key ideas of Social Construction, and specifically Berger and Luckmann (1967). Meyer and Rowan (1977) challenge previous assumptions of the rational organisation with clear, measurable objectives and managers who identify with the accomplishment of these goals. Institutional theorists propose that the formal structures of organisations reflect myths of the external environment of the organisation, as opposed to the actual work activities within the organisation. In addition, these activities which are core, are decoupled with the face of the organisation which is presented to the outside world (Meyer and Rowan, 1977).

Two key concepts which are central to Social Construction and Institutional Theory are Legitimation, and Isomorphism. These will be discussed in turn below.

#### 4.2.1 Legitimation

A key aspect of social construction is legitimation, which attaches credibility to institutionalised actions (Berger & Luckmann, 1967; Meyer & Rowan, 1977), this is defined as follows:

Legitimation produces new meanings that serve to integrate the meanings already attached to disparate institutional processes (p. 110, Berger and Luckmann, 1967).

Legitimation is not necessary in the first phase of institutionalization, it is required when the objectivations of the (now historic) institutional order are to be transmitted to a new generation. Here, legitimation is necessary in order to explain and justify important elements of the institutional tradition, as the character of the institution can't be maintained by the individual's own recollection and habitualization.

This process is described below:

Legitimation 'explains' the institutional order by ascribing cognitive validity to its objectivated meanings. Legitimation justifies the institutional order by giving a normative dignity to its practical imperatives. It is important to understand that legitimation has a cognitive as well as normative element. In other words, legitimation is not just a matter of 'values'. It always implies 'knowledge' as well (p. 111, Berger & Luckmann, 1967).

This further describes the process whereby 'credibility' is attached to these institutionalised actions by attributing validity.

Meyer and Rowan (1977) also advance the significance of legitimation. They propose that the formal structures of organisations reflect myths of the external environment of the organisation, as opposed to the actual work activities within the organisation. In addition, these activities which are core, are decoupled with the face of the organisation which is presented to the outside world, resulting in the creation of gaps between formal structure and actual work activities. Meyer and Rowan (1977) describe this as follows:

To maintain ceremonial conformity, or organisations that reflect institutional rules tend to buffer their formal structure from the uncertainty of technical activities by becoming loosely coupled, building gaps between their formal structure, and actual work activities (Meyer and Rowan, 1977, p. 341).

Meyer and Rowan (1977, p.352) suggest that decoupling enables organisations to maintain standardized legitimating, formal structures while their activities (the 'core') change in response to practical considerations. They describe the process of legitimation where organisations adopt practices that do not necessarily improve the efficiency of their organisation, but will allow the organisation to present itself as modern and up-to-date.

Meyer and Rowan (1977) believe that formal structures of organisations are adopted ceremoniously and function as myths:

Technologies are institutionalised and become myths binding on organizations. Technical procedures of production, accounting, personnel selection, or data processing become taken-for-granted means to accomplish organizational ends. Quite apart from their possible efficiency, such institutionalized techniques establish an organization as appropriate, rational, and modern. (Meyer & Rowan, 1977, p344)

Such technical procedures, for example accounting, are therefore seen to be adopted in order to establish organisations as 'modern' and therefore legitimate.

# 4.2.2 Isomorphism

Another aspect of the social construction of organisations is Isomorphism. Di Maggio and Powell describe isomorphism as a:

Constraining process that forces one unit in a population to resemble other units that face the same set of environmental conditions (Di Maggio and Powell, 1983, p. 149)

Furthermore, Meyer and Rowan (1977) argue that the structure of organisations reflects their institutional environments, where organisations become isomorphic with them. They draw on the work of Berger and Luckmann (1967) to explain that the reason for this is that organisations structurally reflect socially constructed reality.

In terms of isomorphism, Meyer and Rowan (1977) highlight the importance of organisational language:

A most important aspect of isomorphism with institutional environments is the evolution of organisational language (Meyer and Rowan, 1977, p. 349).

DiMaggio and Powell (1983) are concerned with the considerable homogeneity of both organisational forms and practices. They consider what drives structural change in organisations, they suggest that is it decreasingly driven by competition or by the need for efficiency, and instead attribute it to processes which make organisations more similar, without necessarily making them more efficient. They argue that similarities in organisational forms arise out of the structuration (Giddens, 1979) of organisational fields. The organizational field is defined as follows:

Those organizations that, in the aggregate, constitute a recognized area of institutional life: key suppliers, resource and product consumers, regulatory agencies, and other organizations that produce similar services or products (Meyer and Rowan, 1977, p. 148).

Di Maggio and Powell (1983) further explain that the:

Highly structured organizational field provide a context in which individual efforts to deal rationally with uncertainty and constraint often lead, in the aggregate, to homogeneity in structure, culture, and output (p. 147).

Di Maggio and Powell (1983) outline three different mechanisms which bring about isomorphism. The first of these is coercive isomorphism, which is a result of political influence and also legitimation. Di Maggio and Powell (1983) also discuss how this coercive isomorphism can result in legitimation (Meyer and Rowan, 1977). Here, both formal and informal pressures on one organization by other organizations which they are dependent and by cultural expectations in the society within which organizations function. Such pressures may take a variety of forms and may be forceful, persuasive, or in the form of an invitation. This organizational change can include responses to government mandates, for example Di Maggio and Powell (1983) describe a situation where nonprofits maintain accounts, and hire accountants to enable them to meet them requirements of tax law. Another issue which can affect the behaviour and structure of an organization include the presence of a common legal environment.

Meyer and Rowan (1977) discuss how the organisational environment, including the influence of the state, can result in organisations coming to mirror rules which have been institutionalized by and within the state. This results in organisations becoming increasingly similar within the sector, and conforming to the rituals of wider institutions. By structuring themselves according to their environments, these organisations become structured less around technical activities, and output controls. These pressures are not however limited to the Government, they can also come from outwith.

The second type of isomorphism discussed by Di Maggio and Powell (1983) is mimetic isomorphism. The cause of this form of mimicry is uncertainty, including situations where: there is a lack of understanding of organisational technologies; goal ambiguity; or when the environment creates symbolic uncertainty. One solution which organizations turn to in times of uncertainty is modelling, where one organization borrows practices from another. In terms of the diffusion of these models, this can be unintentional, or can occur indirectly through employee transfer or turnover, or openly by consulting firms or industry trade associations. Innovation can also be attributed to modelling. A considerable amount of the similarity of organizational structures is a result of the lack of variety of structures to be selected from. Under the above circumstances, organizations will tend to mimic similar organizations in their field which they believe to be more legitimate or successful. The universal nature of these mimetic processes is attributed to high presence of certain kinds of structural arrangements, rather than to any evidence that the adopted models enhance efficiency.

The final isomorphic pressures which are discussed are normative pressures which are predominantly the result of professionalization. Di Maggio and Powell (1983) identify two aspects of professionalization which result in isomorphism: the first is formal education; the second is the development of professional networks crossing organisations, facilitating the uptake of new models.

Meyer and Rowan (1977) describe three consequences of isomorphism:

Isomorphism with environmental institutions has some crucial consequences for organizations: (a) they incorporate elements which are legitimated externally, rather than in terms of efficiency; (b) they employ external or ceremonial assessment criteria to define the value of structural elements; and (c) dependence on externally fixed institutions reduces turbulence and maintains stability. As a result...institutional isomorphism promotes the success and survival of organizations (Meyer and Rowan, 1977, p. 348)

These consequences will now be further explained. The first is that organisations adopt structures which are legitimated externally, rather than in terms of efficiency, this helps the organization to secure its survival. Meyer and Rowan highlight the importance of organisational language as a key aspect of isomorphism within institutions, both the way in which the organizational chart is labelled and also the terms used to define the goals, policies and procedures of the organization equate to myths that help to explain and interpret the actions of individuals. Such language relating to structure, isomorphic with institutional rules provides legitimacy. The importance of language is also emphasised by Berger and Luckmann (1967) with regard to the process of social construction. They describe language as 'the most important sign system of human society' (p.51), as understanding the reality of everyday life. The ability of language to become detached from its origins means that it is capable of accumulating and storing meaning and experience which it can preserve and then pass it on to future generations. It allows the build up of symbolic representations and schemes of classification.

Isomorphism also results in organisations employing external or ceremonial assessment criteria to define the value of structural elements, such examples include endorsements by important people, or the prestige of personnel. In addition, they also adopt structures with a high ceremonial value, such as those that are viewed as prestigious or those which reflect the most up-to-date expert thinking.

Finally, isomorphism results in and depends on the way in which organisations conform to and also become legitimated by their external environments. When such isomorphism occurs, organisations gain the legitimacy and resources they need to

survive. Meyer and Rowan (1977) therefore argue that this institutional isomorphism promotes the success and survival of organisations.

Meyer and Rowan (1977) however highlight two issues facing organisations that are dependent on isomorphism with institutionalised rules. The first is a situation where there may be conflicts and inconsistencies between the demands of technical activities and efficiency, and conforming to ceremonial rules. The second issue is that ceremonial rules may conflict with each other, this is because such rules are transmitted by myths which arise from different parts of the environment.

This section has given an overview of the conceptual framework adopted in this thesis. A broad overview of social construction has been outlined and two specific ideas of legitimation and isomorphism have also been explored. The next two sections will consider how these ideas have been applied, firstly in relation to organisational research, and secondly in relation to governance in charities.

### 4.3 The Use of Social Construction in Organisational Research

As will become apparent, social construction has been widely used in research on organisations. Later in this chapter the application of social construction in research on nonprofits will be considered. However, firstly some wider examples of this approach within organisational research will be considered. These examples have been specifically selected due to the nature of how the authors use these ideas to interpret the results, as will become apparent in section 4.4, some studies are more limited in their application of social construction. This section will begin by considering a broad social construction approach (Perren & Grant, 2000), before focusing on more specific aspects of this approach, namely Legitimation (Arnaboldi & Lapsley, 2004; Meyer, 1994, and Mouritsen & Skaerbaek, 1995).

Perren and Grant (2000) use a social construction perspective to investigate the development of management accounting in small firms. The authors argue that

concepts developed by Berger and Luckmann (1967) provide a framework which allows the exploration of management accounting within the context of small businesses. Here, management accounting in small businesses is viewed as being socially constructed by individuals interacting to share understanding, interacting within or outwith the 'micro-world' of the owner-managers business. A framework is developed for analysis based on the key processes of social construction. The authors consider the life history of these owner-managers, from their prior experience of macro-level objectified management accounting knowledge before the start-up of the business, to the acceptance by the employees of the owner-managers habitualized routines as management accounting 'facts', and finally the development of the habitualized accounting routines of employees which may also be interpreted as management accounting facts.

There were three sources of macro-level objectified management accounting knowledge from outside the owner-managers 'micro-world', these were: the owner-managers previous personal experience, external management accounting expertise and accounting software, and management accounting knowledge gained from the previous experience of employees. The primary motivation for many of the management accounting actions was the owner-managers fear of failure. This led to externalization and some routines becoming habitualized prior to influencing the actions of the employees. There were only a few examples in the cases where an employee had taken action which resulted in an habitualized management accounting routine that then resulted in an objectified management accounting 'fact'.

The social process as described by Berger and Luckmann is found to be in existence in this research, where the owner-manager or employer takes actions which are 'externalized' and may become habitualized. The occurrence of the habitualization of management accounting actions are deemed to take place so, as Berger and Luckmann note they 'can be reproduced with an economy of effort' (p.71). The fear of the owner-managers of failure is seen as a key motivation for many of the management accounting actions which were personally instigated – these led to externalisation and the habitualization of routines. Some of the habitualized routines

appear to have become objectified into management accounting 'facts' within the 'micro-world'.

In contrast to the broad social construction approach from Perrin and Grant (2000) there is also a wide literature which considers more specific aspects of this approach. There are a number of studies which consider how organisations legitimate themselves. One such study is Arnaboldi and Lapsley (2004) who examine the introduction of activity based costing (ABC) – an accounting tool which eminates from the private sector - into a healthcare organisation which is undergoing change. The specific nature of this organisations activities are concerned with the transformation of donated blood into a range of healthcare products. The adoption of this accounting technique has been initiated by the organisation itself, this is specifically relevant within this study, as accounting practices are generally set by central government in the UK. The specific concepts drawn on in this paper are that of legitimacy and mimicry. This paper considers the pressures of the organisation to become what Brunsson and Sahlin-Anderson (2000) term the 'complete organisation', whereby public sector organisations begin to mimic organisations from the private sector. This paper considers the various stages of the implementation of ABC, and finds that it is closer to an exercise in legitimation than a practical tool which enhances efficiency.

Throughout this paper, examples are identified which link back to the conceptual framework. For example, the way in which the healthcare organisation portrays itself as a 'complete organisation'. One such instance is the use of business language where it identifies its core business the blood 'supply chain'. This business language is identified as an attempt by the organisation to portray itself as legitimate (Meyer and Rowan, 1977). The results further identify the pressures which the organisation faces to become akin to a business, and therefore present itself as a complete organisation. An example included the discovery of AIDs, the impact of which resulted in a reorganisation whereby support and analysis services were centralized at the headquarters, this was identified as being consistent with the construction of a

'complete organisation', where the public sector organization establishes an authorative centre (Brunsson & Sahlin-Anderson, 2000).

In the organisation, the interest in ABC has been recurrent since the 1980s, however, previous implementation attempts had failed. One such failure was due to the realization that the organisation was not structured in an appropriate way to allow the achievement of economies of scale in the manufacturing and testing of blood. This was linked to what Brunsson and Sahlin-Anderson (2000, p727) call the construction of rationality, whereby top management discovered that the structure of the organisation was not appropriate to the change in context. Most of the interviewees were not able to justify the choice of ABC. The selection of this technique was identified as being more related to the presence of a champion of ABC in the organisation, and the desire of management to use techniques that were 'modern' than to evaluate the costs and benefits of the approach.

The ideas in this paper by Arnaboldi & Lapsley (2004) are particularly relevant to this study. Charities face similar pressures to public sector organisations such as the NHS to portray themselves as modern. Here, the adoption of a private sector tool has been identified as a legitimating device that was 'adopted' to portray the organisation as 'modern' rather than implemented for reasons of technical efficiency. Also of interest in this paper is the idea of the 'complete organization', as outlined in chapter 3, charities are increasingly adopting private sector practices which may be a result of pressures to appear 'complete'.

Another study which has a specific focus on Legitimation, is that of Mouritsen and Skaerbaek (1995). This is a study of the interplay between accounting and management and art as part of civilisation. Mouritsen & Skaerbaek undertake a detailed study of the history of the Royal Theatre in Denmark which demonstrates the complexity of organisational forms. They show how management and rational practices have been deployed over the life of the theatre. Within this they particularly focus on accounting as a device for improving management, they demonstrate that accounting has a major legitimating function in ensuring the continuation of the

entity. This goes beyond management of resources to include the attraction of resources. The authors deploy issues of institutionalism in particular where the organisation is embedded in society. The environment is important in shaping behaviours of organisations. In this way the authors draw on key concepts of social construction – legitimation, symbolic behaviour and role of language (accounting). It develops the idea of accounting as a language and how accounting and art fit together in order to legitimate the organisation on two levels.

A further study by Meyer (1994) is concerned with the institutional and organizational rationalization in the mental health system. The study adopts a case study approach of a specific mental health organization, these are organizations which tend to be not for profit. The study focuses heavily on legitimation, but also considers institutionalization and decoupling. The study considers the impact of the environment of rationalized organizations and the function of legitimacy – a concept discussed by both Berger and Luckmann (1967) and Meyer and Rowan (1977). They explore the issue of legitimation both in terms of the issue of crisis in the mental health sector and the influence of organizational environment which provides legitimacy. The findings in this study show that mental health organisations in America are more internally decoupled than other systems. This follows on from Meyer and Rowan (1977) who describe decoupling as a function of legitimation where external activities and formal structures are separated.

In terms of mental health, Meyer states that:

'Formal structures must be adopted to a complex and changing environment...Yet at the same time reasonable coherent systems of internal activity must go on – there are obvious practical problems of daily activity. The organizational solution in such situations is decoupling (Meyer, 1994, p.225)

This study is particularly interesting as it focuses on a nonprofit organisation and identifies an example of how these organisations decouple their external façade from their core activities.

This section has considered studies which have drawn on social construction (Perren & Grant, 2000) and more specifically legitimation in organisational research (Arnaboldi & Lapsley, 2004; Meyer, 1994; Mouritsen & Skaerbaek, 1995). The next section will look at research which has drawn on these ideas in research on nonprofit organisations.

# 4.4 The Use of Social Construction in Research on Nonprofit Organisations

This section will consider some examples which are more specific to the issues investigated in this research with regard to nonprofit organisations. Papers will be drawn on which have a focus on nonprofit governance and which draw on social construction. The table below gives an overview of these papers, the key area of their investigation and the aspects of social construction which they draw on. These papers are split into five key themes: the role of the board; the composition of the board; board behaviour; board and organisational effectiveness; and organisational structure.

Table 4.1 The Use of Social Construction in Nonprofit Research

Study	Focus	Perspective
The Role of Board		
Cornforth & Edwards	Looks at contribution that	Institutionalization,
(1999)	boards make to	Isomorphism (Di Maggio
	organizational strategy in	& Powell, 1983)
	the public and nonprofit	
	sector.	
Steane & Christie (2001)	Investigates patterns of	Isomorphism (Di Maggio
	governance on Australian	& Powell, 1983)
	nonprofit boards	
Parker (2007)	Examines strategic	Legitimation &
	decision-making at the	Isomorphism (Meyer &
	boardroom level of two	Rowan, 1977; Di Maggio
	nonprofits.	& Powell, 1983)
The Composition of the Board		
Abzug & Galaskiewicz	Whether nonprofits	Legitimation (Meyer &
(2001)	legitimate their board	Rowan, 1977)
	through recruitment of	
	educated professionals or	Isomorphism (Di Maggio

	community	& Powell, 1983)
	representatives	
Board Behaviour		
Wood (1992)	Proposes a model of	Social Construction
	nonprofit board	(Berger & Luckmann,
	behaviour.	1967)
Board & Organisational Effectiveness		
Herman & Renz (1997)	Board & org effectiveness	Social construction
		(Berger & Luckmann,
		1967)
Herman et al (1997)	Board effectiveness	Social construction
		(Berger & Luckmann,
		1967)
Organisational Structure		
Leiter (2005)	Examines similarity of	Isomorphism (Di Maggio
	organizational structures	& Powell, 1983)
	of Australian nonprofits.	
Leiter (2008)	Examines organizational	Isomorphism (Di Maggio
	structures of nonprofit	& Powell, 1983)
	organizations in Australia	
	and the US, and the extent	
	to which they are similar.	

The first set of papers by Cornforth & Edwards (1999), Steane & Chistie (2001) and Parker (2007) focus on the specific role of the board. The idea of institutionalization is used in a study by Cornforth and Edwards (1999). They focus on the contribution that boards make to organisational strategy in the public and nonprofit sector and focus on four cases; a local voluntary organisation, a school, a national voluntary organisation, and a college. The authors state that their research draws on new institutional theory (Di Maggio and Powell, 1983; Scott & Meyer, 1994), they consider the impact of the institutional environment on board roles and how ideas become institutionalised, particularly through coercive, normative and mimetic pressures. In the cases, coercive, normative and mimetic pressures were identified which shaped the nature and the role of the board. For example, one of the cases was a Further Education College, the three isomorphic pressures were identified as follows (1) coercive pressures came in the form of government legislation requiring Colleges to recruit people with business experience to their boards, and also forcing FE Colleges into competition with other educational organisations (2) this results in

normative pressures, with a more managerial view of governance, and the idea of the College as a business (3) mimetic pressures were also apparent, where the senior management of the College wanted to mimic successful businesses that they viewed as role models. The findings of this study are particularly pertinent to this thesis. Of specific interest is the consideration of how these different isomorphic pressures shape the role and structure of charity boards, and also the findings surrounding the pressures for the organisations to become more business-like.

Isomorphism is also used to understand the nature of nonprofit governance and the role of the board by Steane and Christie (2001). This study investigates similarities in Governance of Australian nonprofits, via questionnaire to the chair of the board. The findings here suggest that although nonprofits do mimic some aspects of the forprofit, or 'shareholder' governance model, the majority of boards are more similar to the traditional nonprofit 'stakeholder' approach. At the beginning of the paper, Steane and Christie (2001) summarise their findings, and note that:

Australian nonprofits manifest features that are alike in appearance to practices of the corporate sector, but with some differences. Similarities can arise due to coercive or education forms of mimicking (Di Maggio and Powell, 1983), but some features remain distinct despite the influence of legislative and governmental changes on directors.

Although at the outset of their paper they make reference to Di Maggio and Powell (1983), and the idea of isomorphism, this concept is not developed throughout the paper to understand why similarities exist, this is in fact the only point that the term isomorphism is used, and reference is made to Di Maggio and Powell, 1983. For example, in the introduction, Steane and Christie highlight the pressures on Australian nonprofits from Government to become more like the corporate sector, however investigating this in terms of ideas of coercive isomorphism is not carried out. The findings show an uptake of corporate practices is typical of boards with members with management, law or marketing expertise, however these findings are not discussed in terms of normative isomorphism which is facilitated through formal education and networks of professionals which encourage the uptake of new models (Di Maggio and Powell, 1983). Although they do highlight that some nonprofits

'mimic' their for-profit counterparts, they do not explain these findings with reference to the work of Di Maggio and Powell (1983). Finally they do not explore the possibility that isomorphic tendencies towards the corporate model of governance could be a form of legitimation. This is surprising as the pressure for these organisations to appear more like the corporate sector is highlighted in the paper. Despite the limited use of the ideas of Di Maggio and Powell to explain the results of this study, the findings are still of interest in relation to this thesis. In particular, that a high presence of business expertise on nonprofit boards results in the mimicking of for-profit boards, particularly in relation to aspects such as the increased focus on policy and strategy over program management.

A further study which uses ideas of Legitimation and Isomorphism to explore the role of the board is Parker (2007). This paper investigates strategic decision-making at the boardroom level of two nonprofit organisations. They find the existence of normative, mimetic, and coercive isomorphic pressures. Normative pressures were found in the common educational and business background of board members. This business presence was found to result in mimetic isomorphism, with the documentation of the equivalent of private sector company missions, strategic plans, business plans and annual budgets. Coercive isomorphism came from the national office of the nonprofit in the form of a national business plan and budgetary control, where the branches came to be aligned with a more commercial philosophy as demanded nationally. As noted below, formal and strategic business plans were found to be adopted ceremoniously, legitimising the organisation:

Formal strategic and business plans largely served as a legitimising role of matching the expectations of major shareholders such as national board and membership in general. For both boards the formal strategic plan played no major role in boardroom strategizing but appeared to play a legitimising ceremonial role of presenting the rational planning and control imagery expected of corporations in the private sector (Parker, 2007, p. 1473).

This is relevant in highlighting various isomorphic pressures for nonprofits to become more business-like, particularly the recruitment of business people onto nonprofit boards and the resulting mimicking of private sector organisations. Also of interest is the finding that nonprofits adopt formal structures and blueprints, in this case in the form of formal plans, to legitimate the organisation.

Legitimation (Meyer & Rowan, 1977) and Isomorphism (Di Maggio & Powell, 1983) are used by Abzug and Galaskiewicz to explore board composition in nonprofit organisations. At the outset they highlight the importance of the board, and its composition as a legitimating device:

The boards of nonprofit organisations perform important governance and legitimation functions. They have the responsibility of ensuring that the organization lives up to its mission and expends resources in a fiscally responsible manner. Because of this oversight role, the board often comes to symbolize or represent the organization to the broader community. Given that stakeholders often cannot know if nonprofits are faithful to their mission or use funds wisely, they judge the organization by seeing who is on its board of trustees. Board members come to represent the organization and become a basis for its legitimacy claims. The composition of boards, then, is of central importance to nonprofits (Abzug and Galaskiewicz, 2001, p. 51).

In this study, two perspectives on legitimacy are considered: (1) the credentials and expertise of trustees which communicate the organisations commitment to efficiency norms and the business model (Di Maggio & Powell, 1983) (2) legitimacy in terms of constituent support, where an organisation is legitimate if it represents the identities or interests of different constituencies in the community – an aspect neglected by neo-institutional theory. This paper examines the composition of US nonprofits boards at three points in time – 1931, 1961, and 1991 – to consider (1) whether the percentage of directors with professional and managerial backgrounds has increased over time and (2) if boards have become representative of different community interest groups. The findings of this study produced mixed results. In support of neo-institutional theory, over the time period investigated, there was an increase in the proportion of board members who achieved high educational credentials, and also those with managerial backgrounds. However, the findings also suggest that there are still a select number of nonprofits who were found to legitimate their boards on the basis of community representation, rather than with the presence of highly educated managers. The idea of nonprofits legitimating their boards through the recruitment of individuals with a business/ management background is central to this thesis. However, Abzug and Galskiewicz (2001) highlight a limitation in neo-institutional theory, which is that it does not recognise that in addition to legitimacy from above, some nonprofits are required to show they represent or serve their community below.

Social construction is also used to explore board behaviour. Wood (1992) proposes a cyclical model of board behaviour. The focus of this study is human service agencies, and interviews were conducted with key individuals involved in governing the organizations. Prior to the development of this model and its various stages, Wood states that:

Implicit in my argument is the assumption that reality is a social construction (Berger and Luckmann, 1967). Empirically, this means that events internal and external to an agency are subjectively interpreted by individual members and by the board as a group (p. 143).

Wood then goes on to describe the various stages of the model, however for the rest of the paper no other references are made to Berger and Luckmann or to social construction. Opportunities are available in the paper to draw on ideas of Berger and Luckamann, or other work which develops similar ideas. For example, one of the stages described here is the 'Corporate Phase', during which 'an unambiguous signal is given to the executive director to assist the board in mimicking the Fortune 500 model of oversight' (p. 147). Mimicking behaviour such as this is outlined above as 'isomorphism' (Meyer and Rowan, 1977; Di Maggio and Powell, 1983), however this work is not drawn on in this paper. Di Maggio and Powell outline three isomorphic processes (see above), but these are not drawn on to understand this behaviour. Furthermore, it is also noted that during this stage 'the corporate style becomes entrenched and highly routinized', this routinization is another key aspect of social construction, however Wood does not develop this idea or explain its importance in social construction. Another stage in the board cycle is described as the ratifying stage where ritualization is key: 'Processes for decision-making are ritualized; the ritual itself becomes more important than its content; and staff recommendations are ratified without question, sometimes without comment' (p.

148). Ritualization is also discussed as a key element of social construction by Berger and Luckmann (1967). This paper is limited in its discussion of the use of social construction to explore board behaviour and would have benefited from drawing more extensively on the work of Berger and Luckmann to explain the behaviour of nonprofit boards.

Board and organisational effectiveness is also explored through social construction. The study by Herman, Renz and Heimovics (1997) is concerned with organizational effectiveness, in particular the effectiveness of the nonprofit board and is driven by social construction. This study takes the form of a survey of various stakeholders of two specific nonprofit organisations. The key concept employed in this study is that of myths. Under social construction, 'there is no independently real board (or organizational) effectiveness' (p.375). Another implication in the social constructionism approach is that it states that 'there is no true or valid interpretation' (Crotty, 1998, p.47), this therefore implies that there may be variations in the judgements of the stakeholders which were surveyed in terms of their views on effectiveness and also the different kinds of information which they use in informing their judgements.

A study based on a survey of stakeholders is by Herman and Renz (1997) this study however is concerned with overall effectiveness of nonprofit organisations, not just the board. This study is driven by social construction and the authors assume under this perspective that organizational effectiveness is a set of judgements by stakeholders this resulted in varied judgements by stakeholders of the same organisation. They find that all stakeholders use socially constructed evidence of board effectiveness when they form judgements of organizational effectiveness.

The final two papers explore organisational structure, isomorphism is the key focus of Leiter's (2005) study of Australian nonprofit organisational structures. In this paper the level of isomorphism in Australian nonprofits is considered, along with the specific factors which are deemed to cause isomorphism, as outlined by Di Maggio and Powell (1983). In terms of the analysis, emphasis is placed on the variations in

organizational characteristics, where the lower the variation is, the greater the isomorphism. The focus on variation is suggested by Di Maggio and Powell (1983) who note that 'the best indicator of isomorphic change is a decrease in variation and diversity, which could be measured by lower standard deviations of the values of selected indicators in a set of organizations' (p. 155). The findings do not show evidence of isomorphism among Australian nonprofits, instead, they suggest a variety across the sector. Furthermore, the findings do not show support for the application of Di Maggio and Powells explanations for isomorphism to Australian nonprofits as an organizational field. Despite these findings, Leiter recognises the influence of the work of Di Maggio and Powell (1983) in the organization studies literature, and notes that the failure of the results to follow the theory may not necessarily reflect the absence of isomorphism in nonprofits, but the difficulty of detecting it within the data. Due to the nature of the these unexpected findings, Leiter places emphasis on the limitations of the study and highlights the importance of exploring different specifications of the organizational field.

Another study which uses Isomorphism is that of Leiter (2008) which looks at the extent of isomorphism in the organizational structures of nonprofit organizations in Australia and the US. The findings report the level of isomorphism; with the extent of isomorphism almost the same for 6 out of the 8 characteristics investigated, namely: departmentalization, hierarchical levels, formalization, autonomy, work intensity, and organizational age. However, specific reasons for this isomorphism is not discussed, in particular the specific isomorphic process of coercive, mimetic, and normative isomorphism are not developed throughout this paper, despite being discussed in the introduction. This paper would have benefited from discussion of the possible reasons for such isomorphism, such as similarities in government legislation. The author also highlights another limitation of the paper in that the potential consequences of such isomorphism is not considered.

This section has provided an overview of research on nonprofit governance which draws on aspects on social construction. The following section will consider the relevance of these studies, and those in the preceding section in relation to this thesis.

### 4.5 Conclusion & The Relevance of Social Construction to this Study

This research investigates Governance in large Scottish charities. As outlined above, the conceptual framework of this study will draw on two specific aspects of social construction: legitimation and isomorphism. This chapter began with an overview of these ideas and then considered their broad application in organisational research, and then their use in nonprofit governance. This section will further consider the relevance of these studies in this research and also how the key concepts of legitimation and isomorphism will be used.

The use of social construction in organisational research was considered in section 4.3. These papers were identified for their in-depth use of key ideas of the conceptual framework. Arnaboldi and Lapley (2004) provided an insight into how a public sector organisation adopted a private sector accounting technique in order to appear 'modern' and therefore legitimate. They also introduced the idea of the 'complete organisation', where public sector organisations adopt private sector practices in order to resemble these organisations which are seen as 'complete'. This is particularly useful in relation to this study, where it has already been identified that charities face pressures to become more 'business-like'. The literature presented in chapter 3 gave examples of how charities have adopted practices from the private sector. Mouritsen and Skaerbaek (1995) developed the idea of accounting as a language and how it is used to legitimate the organisation. This is particularly useful in this thesis as the use of business language by charities will be considered. Finally, Meyer (1994) describes a situation where a nonprofit organisation as decoupled its external activities from its core in order to maintain legitimacy.

Section 4.4 considered specific studies which explored various aspects of governance through the use of social construction, these have provided useful insights for this thesis. They provided examples of the various isomorphic pressures which nonprofits face, and how these shape various aspects of nonprofit behaviour and more specifically, the behaviour of the board. These studies are particularly useful with regard to the aspect of this study which focuses on the impact of business expertise

on nonprofit boards. For example, Cornforth and Edwards (1999) and Steane and Christie (2001) find the existence of coercive pressures for charities to become more business-like resulting in them mimicking private sector organisations. In addition, Abzug & Galaskiewicz (2001) find evidence of the use of business people on nonprofit boards being used as a legitimating device.

It has been reported that pressures to become more business-like are apparent in the charity sector in Scotland. One example of such a pressure is for the increase of business skills, such as finance, on charity boards, for example from the Scottish Charity Regulator (OSCR, 2008) – this could be seen as coercive isomorphism (Di Maggio and Powell, 1983). The adoption of business people onto charity boards may also be a form of mimetic behaviour, where charities attempt to present themselves as 'modern' by copying the practices of other charities in the same 'organizational field' which they see as legitimate (Di Maggio and Powell, 1983).

Within the context of governance, this study will look at the specific isomorphic pressures faced by these organisations and the extent of isomorphism in the governance structures in these organisations within the charity field and the recruitment of business people onto charity boards. Governance is a crucial element of the legitimating function of charities. It sends signals to the external legitimating environment that all is well, that individuals with up-to date expertise have been recruited. Abzug and Galaskiewicz (2001) emphasise the importance of the board as a legitimating function. Also considered therefore, is whether the recruitment of business people onto the boards of charities is merely a legitimating exercise, as charities look to portray themselves as modern and up-to-date.

This study aims to build on the ideas and studies presented above. For example, although Abzug and Galaskiewicz (2001) consider whether boards recruit members with management backgrounds, this study is limited due to the historical nature of the data. This present study will gather primary data from Scottish charites, with a survey to investigate the incidence of these individuals on charity boards, and a

series of in-depth case studies which will allow investigation of whether the role of these individuals is merely symbolic.

This research also considers the importance of the language of business in the social construction of charitable organisations. The importance of language is highlighted by Berger & Luckman (1967), Meyer & Rowan (1977) and Di Maggio & Powell (1983). Meyer and Rowan (op cit.) note that language can function as myths that interpret and explain the actions of individuals. In the context of this study, such an example would be where the myths of accountants and other business people on boards explain organisational activities. In terms of business people such as accountants and management consultants, people will say that they will perform certain tasks or solve specific problems, without actually knowing the specific individuals or exactly what they will do. However, in such a case, both the speaker and the listener understand such statements to describe how certain responsibilities will be carried out.

Considerable research has been undertaken on public sector organisations with regard to the implementation of private sector practices, with numerous studies identifying the adoption of such practices as legitimating devices (see Arnaboldi & Lapsley, 2004). However, such a situation has not received significant attention in relation to charities (with the exception of Abzug & Galskiewicz). This situation is particularly pertinent in the context of charities, who face increasing pressure to become more business-like. Of particular interest is the recruitment of business people onto the boards of charities, this may be part of the social construction of the organisation, as charities seek to portray their organisations as modern and up-to-date. They may recruit business people in order to maintain legitimacy, with their role being more symbolic and ceremonial. In addition, chapter 3 presented examples of charities adopting private sector practices, however there is a lack of research which considers whether the adoption of such practices is merely a function of legitimation.

### **CHAPTER 5**

#### **METHODOLOGY**

#### 5.1 Introduction

In chapters two and three the research context was considered, it was evident from this that there has been a significant lack of research into charity governance in the UK, particularly in Scotland. It was also highlighted that this existing research has largely overlooked the phenomenon of business expertise on charity boards, and its associated impact. The general influence of business on charities has been considered in chapter 3, presenting itself as a contentious issue, with polarised views as to the appropriateness of charities engaging with and adopting business practices and expertise. When this research commenced, there was a clear lack of studies considering these issues in the Scottish context. A need was therefore identified to examine governance in charities in Scotland, particularly in terms of internal governance relating to the functioning of the board, and externally, to assess the effectiveness of new regulatory measures, such as OSCR to see if they overcome the defects of prior arrangements. Furthermore, a deeper understanding of the incidence and impact of business expertise on charity boards is required, particularly in the context of increasing pressure from the charity regulator for charities to become more 'business-like' (OSCR, 2008). The ability to adopt sophisticated practices – while preserving their distinctive ethos represents a fundamental challenge for these organisations. This research aims to address these gaps.

This chapter sets out the research questions of this study and the research approach employed. It will be structured as follows: (5.2) will outline the research questions this study seeks to address; (5.3) outlines the research methods and data collection; (5.4) considers how the research was structured with regard to the conceptual framework; (5.5) outlines the limitations of the research; and (5.6) finally this chapter will be concluded.

### **5.2 Research Questions**

The aim of this research is to examine Governance in Scottish charities. This study will contribute to our understanding of internal governance, particularly in relation to the functioning of the board, encompassing the role of the board, the relationship between the board and management, board and organisational effectiveness, and the composition of the board. Externally, the impact of the new charity regulator, OSCR, will be considered. In addition, the impact of business expertise on charities will also be investigated by considering the presence of business professionals on charity Boards. Specifically, this is a study of the largest section of Scottish charities on the basis of annual total income.

This study will seek to address the following research questions:

- 1. What kinds of governance issues do charities face in achieving their missions? Why is this the case?
- 2. What impact does external expertise, particularly business expertise, have on the governance and management of charities?

#### 5.3 Research Method

This research was undertaken using a multi-method approach – encompassing both quantitative and qualitative elements, with the use of both a survey and a series of case studies. Utilising a variety of methods to study a phenomenon has been seen to improve the validity and overall reliability of the research, as weaknesses of particular methods can be counter balanced by the strengths of others (Denzin, 1978); research such as this is espoused by Morgan (1998).

Such an approach is outlined by Yin (2003):

Multimethod studies can pose complimentary questions that are to be addressed by different methods. Most commonly, case studies are used to

gain insight into causal processes, whereas surveys provide an indication of the prevalence of a phenomenon (Yin, 2003, p. 151).

The benefits of using preliminary quantitative methods to contribute to a principally qualitative study are outlined by Morgan (op. cit.):

The knowledge provided by an initial small-scale use of quantitative methods helps to guide the decisions that the researcher makes in the larger qualitative research project. The classic example is a preliminary survey or census of a field setting either to guide the selection of sites and informants or to provide a context for understanding the contacts that one does make. Preliminary quantitative results can also help to focus the analysis on large amounts of qualitative data (Morgan, 1998, p.369).

Arnaboldi and Lapsley (2003) provide such an example of a preliminary quantitative study which then contributes to a principally qualitative study in their investigation of the implementation of a private sector accounting technique – activity based costing (ABC) – in local authorities in Scotland. Initially, a survey was conducted of all local authorities in Scotland, in order to establish the incidence of ABC use. Following this, case study sites were selected on the basis of the answers to the survey, allowing a more in-depth exploration of the use and rejection of ABC.

This thesis followed the two-stage approach outlined above. Initially, a survey was conducted of the largest section of Scottish charities, followed by a series of more indepth case studies. The design of this study allows for triangulation. Triangulation is advocated by Denzin (1978) who describes it as the combination of a range of methodologies to study the same phenomena. Denzin (op cit.) outlines triangulation as follows:

Triangulation, or the use of multiple methods, is a plan of action that will raise sociologists above the personalistic biases that stem from single methodologies. By combining methods and investigators in the same study, observers can partially overcome the deficiencies that flow from one investigator or one method (Denzin, 1978, p.294).

Denzin (op cit.) outlines four types of triangulation: data, investigator, theory, and methodological triangulation. This study employs methodological, or more specifically, between method triangulation through the use of a survey and case studies, utilising interviews and documentary evidence. With regards to this strategy, Denzin (op cit.) highlights:

The rationale for this strategy is that the flaws of one method are often the strengths of another; and by combining methods, observers can achieve the best of each while overcoming their unique deficiencies (Denzin, op cit. p. 302).

Further rationale for the choice of this approach and methods will be outlined later in this chapter. Firstly, however, attention will be turned to the specific context of the research – the Scottish charity sector – and on how the population was defined.

# 5.3.1 Defining the Population

As outlined above, this research employed both a survey and also case studies. Before defining the population for this study, it was necessary to consider the specific nature and context of the study, and the essential attributes the population must have. This research centred around the Governance and Management of charities and the interaction between the two. There was also a distinct focus on the impact of business expertise on these organisations which was investigated by considering the impact of those individuals with business expertise that sit on charity Boards, and challenges arising from this in these organisations where traditionally mission and values take priority.

When considering these factors in terms of defining the population for this study, it became clear that the essential characteristics these organisations were required to have made it unfeasible to include small charities in this research. Firstly, there needed to be a distinct boundary between the management/workers of the organisation and the Board in order to understand the relationship between the two. In smaller charities, the boundary between the Board and management/workers tends

to be blurred, with few, if any directors that are external to the organisation. Another condition which needed be met is that external directors must be present on the Board, and in particular business professionals – as noted above, smaller charities may not have external directors, and those which do may find it difficult to attract business professionals onto their Board. In addition, there was a requirement that organisations must be significant enough in size to have the ability to adopt sophisticated business practices. As a consequence of this, small charities were not included in this study.

The study focused on the largest section of Scottish charities; other studies which have targeted larger charities include Hyndman (1990), in this study of the information needs of contributors to UK charities, the population consisted of the largest 200 fundraising charities in the UK. It is expected that organisations that have a larger income will require/have in place more management expertise and also sophisticated management/ governance systems to run the organisation.

The set of organisations which the population for this study was drawn from was the Office of the Scottish Charity Regulators statutory register of charities. As previously outlined, OSCR was established in December 2006 and is the independent regulator and registrar of charities in Scotland, the statutory register was introduced on 1 April 2006. Table 5.1 below documents the number of organisations registered with OSCR and their annual income for 2005/06.

Table 5.1: Total Number of Charities Registered with OSCR as at 1/8/2006

Income Band (£)	Total Number of Registered Charities
0 – 24, 999	22116
25, 000 – 99, 999	3738
100, 000 – 249, 999	1460
250, 000 – 499, 999	519
500, 000 +	939
Total	28772

Source: OSCR (2006)

Table 5.1 shows that the majority of these charitable organisations are in the smallest income bracket £0 - £24,999, these were excluded due to the factors outlined above. As a result of the decision to focus on the larger section of Scottish charities, the top section of the OSCR register was interrogated further, OSCR defines a large charity as an organisation with an annual income of £500, 000 or more; there were 939 charities such as these. However, it was felt that 939 was still too large a number of organisations to survey, but further interrogation of the OSCR register showed that there were 211 charities in Scotland with an income of £5 million and more. These large charities make a significant contribution to the sector, with 97.3% of total income from the sector made by just 5.6% of charities (OSCR, 2008). Therefore, a decision was made to focus on the largest section of charities.

On closer inspection, it was discovered that a number of these charities were actually part of the same organisation; after removing these organisations from the study, the number of organisations in the population was reduced to 204. These organisations were then split into categories in terms of charitable purposes – as defined by OSCR. OSCR defines 15 charitable purposes – however, only 10 were used in this study to classify these organisations.

#### These charitable purposes were:

- 1. The Advancement of Education
- 2. The Advancement of Arts, Heritage, Culture or Science
- 3. The Advancement of Health
- 4. The Relief of Those in Need by Reason of Age, Ill Health, Disability, Financial Hardship or Other Disadvantage, Housing Associations
- 5. The Advancement of Environmental Protection or Improvement
- 6. The Advancement of Public Participation in Sport
- 7. The Advancement of Animal Welfare
- 8. The Advancement of Religion The Prevention of or Relief of Poverty
- 9. The Advancement of Citizenship or Community Development
- 10. More Than One Charitable Purpose

In order to assign the organisations to these groups, it was necessary to consider each organisation individually, taking into account the overall mission of the organisation and also the work in which they did. Due to the nature of this study (the investigation of governance and management in charities) an important aspect of these organisations was that they were actively engaged in providing services, rather than being simply grant awarding trusts. After considering the work of each organisation, it was discovered that a number did not meet these requirements. For example, (Emms Nazareth; Greater Glasgow Health Board Endownment Fund; Lothian Health Board Endownment Fund; Robertson Trust; Lloyds TSB Foundation for Scotland; Volant Charitable Trust). In addition, One Plus: One Parent Families went into liquidation on the 25<sup>th</sup> Jan 2007. This reduced the total number of organisations in the sample to 180. It was also felt that this would be a manageable number of organisations to include in a survey and obtain a high response rate – it would be practical to chase up individual organisations with a smaller sample of organisations. In addition, there are a number of surveys which sample high numbers of organisations which have had very low response rates (e.g, Cornforth, 2001). Case study sites were also selected from this group of the largest charities.

### 5.3.2 The Survey

The first stage of this research utilised a survey, through the use of a self-administered questionnaire. The questionnaire was distributed via email to a key informant – the secretary to the board – of the largest section of Scottish charities identified above. There were several reasons for the design of the research in this way. Methodologically, the use of the initial survey allowed data to be obtained across the complete 'organizational field' (Di Maggio & Powell, 1983) which was the largest section of Scottish charities. It was also set up to allow the analysis of the incidence of various norms and behaviours of charity boards, including the inclusion of business people, this enabled the consideration of whether isomorphism was apparent (Di Maggio & Powell, op. cit.). The survey was also used to guide decisions

in the second stage of the research, namely to aid the selection of specific case study sites and to highlight matters for further investigation. This section will now consider the use of the survey – and in particular of a self-completion questionnaire as a research method.

Quantitative research, such as the use of surveys has been viewed as the dominant form of research in nonprofit organisations (Pharoah, 2005; Alcock and Scott, 2005). Similarly, Bielefeld (2006), who considers this type of research in nonprofit organisations, highlights that in the journal *Nonprofit Management and Leadership*, during 2003 and 2004 70% of the research articles published could be classified as quantitative, during the same period, 81% of research papers published in *Nonprofit and Voluntary Sector Quarterly* were classed as quantitative. Bielefeld (op. cit.) calls for a continuation of this trend, raising concerns over nonprofit managers placing reliance on research into management practices in the for-profit sector, which is potentially incompatible with the nonprofit sector. In this paper, the need is highlighted for management research specific to the nonprofit sector, specifically research that is quantitative in nature. There are a number of studies which have used surveys to investigate issues of nonprofit governance - for example, Brudney & Murray, 1998; Cornforth, 2001 - these are largely international, with relatively few considering the UK, and none investigating the Scottish context.

In considering the use of the survey, there are a number of potential challenges which must be taken into account. Among the biggest challenge is that of nonresponse, this issue is highlighted by Kennedy and Vargus (2001) who consider this challenge in the context of philanthropic studies research. They note that there is a decline in survey participation which is set to continue, reasons given for this trend include the decrease in civic engagement as noted by Putnam (2000), and an increasing level of phonecalls and letters concerning issues such as fundraising which cause confusion among potential survey respondents. Despite this, Kennedy and Vargus (2001) emphasise the quality of the data collected via these surveys; increased knowledge and skill of researchers has resulted in the design of questionnaires which are clearer and which allow respondents to answer with increasing accuracy. Furthermore, they

also note that the decrease in mail survey response is significantly smaller than in interviewer administered surveys.

This issue of nonresponse is also considered by Smith (1997) who reflects on the results of a survey of NPOs working for Peace and evaluates whether the high response rates believed necessary for valid surveys of individuals are the same for organizational-level surveys. The findings here show that valid results can be obtained from organizational surveys with lower response rates – the decision not to respond to an organizational questionnaire may be the product of more random factors than a comparable individual decision. The author also comments on the work of Tomaskovic-Devey et al (1994) who highlight a number of possible factors which determine the respondent's decision to complete a questionnaire which seeks information on their organization: whether they have the authority, capacity, time, knowledge or access to provide information and whether they see an organizational interest in participating in the study. Smith applies this to NPOs and assumes that there will be a consistent level of motivation to participate in a survey across a population of NPOs, because these groups characteristically aim to be visible in their communities. A study that promises to provide information seen as useful to the organisation that is done by a researcher who is seen as legitimate and trustworthy by organisational leaders should appeal to their motivations.

This problem of nonresponse can result in error/bias in the results. Hager et al (2003) discuss nonreponse bias in the context of nonprofit organizations. They study various ways of improving returns to eliminate this effect, in particular they look at the influence of three factors of response rates: they find that the complexity of the survey and the use of small incentives has little or no influence on response rates. However, they find that response rates improve when FedEx is used over standard mail. These are interesting findings however, cost constraints mean that this would not be viable for this study.

Other limitations are highlighted by Bryman (1989), these include the lack of an interviewer to aid the respondent if they do not understand a specific question – this

was be addressed in this study by administering the questionnaire by email, this made it easier for the respondent to seek qualification of questions. Another issue with self-completion questionnaires is that of the lack of ability of collecting additional data through making observations and requesting documents which would be possible with interviews. However, because this questionnaire was distributed via email, directly to the respondent, key documents could also be requested electronically if necessary; there was also the opportunity to collect more in-depth data when conducting the case studies. The issue of nonresponse was highlighted previously with regard to surveys of charities, Bryman (1989) – who discusses the survey in a more general organisational context - also lists this as the most fundamental issue of self-administered questionnaires, however in this study, a number of steps were taken to combat this, which will be outlined in detail later in this chapter.

There have been a number of surveys conducted in the area of charity governance which have gained high rates of response, these include Preston and Brown (2004) who received a 73.6% response rate from Chief Executives and a 50.1% response rate from Board members for their survey of the commitment and performance of Board members, Holland and Jackson's survey of Board development achieved a 62% response rate and Hoye (2004) received a 66.7% response rate in a survey considering the relationship between chief executives, chairs and directors of boards. In addition, these high response rates concur with the views of Smith (1997), who believes that due to the nature of charities, valid research is likely to appeal to charity leaders, making them more likely to participate. Although it is apparent that a number of limitations exist with the use of the survey method via self-completion questionnaire, these issues have been considered in turn with respect to this study and appropriate steps were taken to alleviate any negative aspects of this method of research.

Despite the challenges outlined above, there are a number of potential advantages to be gained from using self-completion questionnaires. The rationale for this method was considered at the outset of this chapter. Further more, this section has already highlighted a number of studies which advocate the importance of survey research in charities (Bielfield (2006), Kennedy & Vargus (2001)), other observations about the advantages of this type of research are outlined by Bryman (2001). First comes the consideration of cost, the questionnaire is considerably cheaper to administer than, for example carrying out a large number of structured interviews, furthermore it allowed organisations from all over Scotland to be included in the study. Another positive aspect of a self-completion questionnaire is that it was also quick to administer, in the case of this study, via email; it would not be feasible in terms of time or money to carry out interviews with a large number of individuals from charities across Scotland. The use of this tool also eliminates any interviewer effects.

## 5.3.2.1 Designing the Questionnaire

As noted above, a self-completion questionnaire was utilised in this study. This section now considers the process of designing this research instrument which is presented in appendix 1.

Considerations which had to be made while designing the questionnaire were:

- The research questions; there are two research questions which were investigated in this research, the first focused on governance issues and the second focussed on the impact of business expertise on the board, therefore the questionnaire had to address both these issues.
- The key informant; as noted above, was the secretary to the board of directors, this person generally holds a senior management position within the charity or within an external organisation, therefore there may be a limit on how much time they can spend on filling in a questionnaire, as a consequence the questionnaire was required to be fairly short and concise.
- The population; this is organisations registered with OSCR who have an annual income of over £5 million.

• Method of administration; self-completion questionnaire distributed via email to the Board secretary.

# Rationale for the Questionnaire

Dillman (1978) highlights five types of questions that can be included in a questionnaire, these are; behaviours, beliefs, knowledge, attitudes and attributes. The two types of questions that were required in this survey are behaviour and attitude. Behavioural questions are important as they establish what people do; there was a need for behavioural questions in this survey due to the nature of the research questions, it was important to understand the behaviour of boards and see whether the behaviour of Scottish charities matches that of previous studies of board behaviour in other countries and to see how these organisations react to certain governance issues. It also allowed the investigation of similar behaviours, and allowed the investigation of isomorphism (Di Maggio & Powell) between these organisations. In addition, the behaviour of business people on charity boards needed to be established in order to answer the second research question which attempts to assess the impact that business people have on charity boards.

Attitude questions will be important in order to draw out the theoretical implications of this research. De Vaus (1991) highlights that attitude questions are important as they allow information to be gathered about what respondents think is desirable; it will be important in this research to understand how charities feel about governance in their organisation and whether the boards behaviour actually matches up to the way that charities present themselves. In terms of the second question which considers the impact of business people on charity boards, it will be important to consider the boards attitude towards these people and what they actually bring to the organisation; there has been some disagreement in previous literature on this as discussed in the review of the literature in chapter 3. Considering the attitudes of the organisation about business people (for example question 10) and the actual behaviour of business people (questions 4,5,6,8) will allow some of the key ideas of

social construction to be drawn out, such as whether business people on boards is simply a form of legitimation (Meyer & Rowan, 1977).

The questionnaire comprised of 11 substantive questions, all of which were open questions. Many experts in survey research favour the use of closed rather than open questions in self administered questionnaires as they feel that forced questions will be easier for the individual to answer and make it more likely they will complete the questionnaire. In addition, closed questions do not discriminate against those respondents who are less articulate (De Vaus, 1991; Newell, 2001), however this should not be an issue with this survey due to the high calibre of the respondent; the board secretary tends to be someone with a significant position within the charity such as the company secretary of chief executive or outwith and therefore should not have difficulty expressing their views via open questions. In addition, open questions allow respondents to answer in their own terms and can therefore result in a higher degree of accuracy (Bryman, 1989).

One of the main aims was to keep the length of the questionnaire short, ideally one page of A4, the main reason for this was the time constraints of the board secretary – it was felt that they would be more likely to fill in a shorter questionnaire which was one page in length and asked concise questions compared with a lengthy questionnaire. The questionnaire was sent to the board secretary as an email attachment, therefore a brief outline of the research and details on how to fill in and return the questionnaire were included in the main body of the email, they were also given the option of going through the questions on the phone if they did not feel they had the time to write answers to the questions themselves. The justification for sending the questionnaire via email was that it was quicker and cheaper that conventional mail, there was more guarantee that it would actually reach the desired respondent, it would also make it easier for the respondent to easily reply to the email if they wanted to ask any further questions about the study to clarify any questions included in the questionnaire it would also make it easier to go back to the respondent if any of the responses they gave were unclear or more information was needed.

#### 5.3.2.2 The Pilot Study

Before conducting the main survey for this investigation, it was decided to undertake a pilot study. Pilot studies refer to both smaller scale versions of the main study intended to be carried out (Polit et al, 2001) and also to the testing of a particular research instrument (Baker, 1994). In terms of this study on charity governance, the pilot study was used for both of these reasons. Several advantages of conducting a pilot study are highlighted by van Teijlingen & Hundley (2001), these include: advanced warning may be given by the pilot study about specific areas where the main research project could fail; it may pinpoint areas where research protocols may not be followed; whether proposed methods or instruments are too complex; poor response rates can also be identified and procedures can be put in place to counter this. Pilot studies have also been advocated by De Vaus (1991) and Prescott & Soeken (1989) who see them as a way of eliminating some of the risks involved in conducting research. However, Teiljlingen & Hundley (2001) also note some limitations in carrying out pilot studies, for example: they may result in inaccurate predictions being made as a result of the pilot data, such as response rate.

Governance and the phenomenon of external business expertise in charities was investigated in two ways via the pilot study. In the first instance the focus was placed on an important branch of charitable activity; namely science, arts, heritage and culture, secondly, with a focus on major charities in the sector (i.e. those with annual turnover in excess of £5 million) to explore scale effects, there are 20 such charities in Scotland. This survey is informed by the debates on how business expertise may or may not be deployed within charitable boards and is directed at key informants within charities, namely the secretary to the board of the charities.

Before the questionnaire could be distributed, the name and email address of each board secretary was required. Although many charities have a section on their website dedicated to information about the board, none of these organisations had the name and contact details of the board secretary, therefore this information had to be actively sought through contacting the organisations. The majority of these charities

had an email address on their website for people requiring general information about the organisation, for those charities that had this; they were emailed asking for this information. Organisations that did not reply to this email, or those that had a telephone number rather than an email address on their website were phoned for this information. The questionnaires were then emailed to the board secretaries, those that did not respond within two weeks were send a reminder email and if there was no response after this, they were contacted by phone. The results of the pilot study are outlined in appendix 4.

### Pilot Study Assessment

Of the 20 questionnaires that were sent out, 14 responses were received. This 70% response rate is very high for a study such as this. However the sample was very small compared with the proposed sample number for the main study (164 organisations); Teiljlingen & Hundley (2001) advise a cautious approach when using a pilot study to predict the response rate for the main study.

The rate of response for the pilot study was high at 70% and provided some interesting findings, as highlighted in appendix 4. However, it is necessary to consider further the rate and quality of responses to individual questions and decide whether it is necessary to change them for the main study. Appendix 2 outlines each question, the rate of response, any issues with the question and any changes required for inclusion in the final survey instrument.

Although the response rate was high and some interesting responses were received, there were issues with some of the questions in terms a lack of detail with some questions due to the tendency towards one word answers. Appendix 2 outlines changes that will be made to the main survey instrument, these changes are outlined below.

### Implications of Pilot Research on Main Study

After considering both the response rate and also the quality of answers from the pilot study, it was concluded that the study was successful and it would therefore be a viable option in taking this study forward to extend this approach for the main stage of data collection (for analysis of pilot questionnaire see appendix 2). There were however a number of issues which required consideration before doing so; in terms of the research instrument, the questionnaire used in the pilot study had quite a narrow focus and centred around the impact of people from business on nonprofit boards, therefore a broader focus was taken in the main questionnaire in order to obtain relevant data to answer research questions 1 and 2.

As outlined above, certain questions required re-wording due to a tendency towards 'yes/no' answers. In addition some broader questions were be added. Measures to enhance reliability were also incorporated into the revised questionnaire. The consistency of the questionnaire was enhanced by adding questions which would highlight inconsistencies in the answers.

Another consideration was concerned with the individual that the questionnaire should be directed to. In the pilot study the questionnaire was sent to the board secretary, who was seen as a key informant. The use of a key informant is considered by Bryman (1989), he highlights their use in providing quantifiable data on characteristics of their organisations and also notes they should be someone in a senior position who can speak for the organisation. The Board secretary fits this description: it is their job to service the Board, taking minutes, circulating board papers and information therefore they know the work of the Board as well as, if not better than any other individual. In addition, they do tend to hold senior positions within the charities or occasionally senior positions outwith the organisation.

Bryman (1989) considers some issues with the use of a single key informant, firstly that you cannot be sure that the person to whom the questionnaire was sent actually answered it. To overcome this problem, the name and contact details of the Board

secretary of each charity was collected and the questionnaire was sent via email to the Board secretary. This way, the Board secretary did directly receive the questionnaire. Another issue is the question of whether one individual has the ability to provide accurate information on the organisation as a whole. In the case of this study, information is being sought about the Board and the Governance of the organisation, it can therefore be argued that the Board Secretary is the person best placed to provide this information as it is their role to service the Board.

There are also concerns over whether the use of a single key informant is too narrow. However, in considering other papers which have conducted surveys of nonprofit governance, there are a considerable number which also use a single key informant, examples include: Brown and Iverson (2004); Callen et al (2003); Gill et al (2005); Herman et al (1997); Brudney and Murray (1998). These studies focus specifically on the Chief Executive as the key informant, other studies such as Steane and Christie (2001) have used the Chairman to inform their study. There are surprisingly few studies which do use the Board Secretray as the key informant in surveys of governance in charities, however two examples of UK studies which have used this approach are Corforth (2001) and Cornforth & Simpson (2002), the reason they give for focusing on this individual is that they are likely to have a good knowledge of the board and also be more dispassionate than others such as the chair or the chief executive. For the purposes of this study on nonprofit organisations and the types of questions asked, it is felt that the board secretary will be the best person to focus on. In terms of questions of narrowness of focus, these were addressed in the second phase of the research, where the views of other key personnel in the organisation were sought from the case studies sites.

#### 5.3.2.3 The Main Survey

The main survey was conducted in the same way as the Pilot study (see appendix 3 for main survey questionnaire). The name and contact details of the Board Secretary of each organisation were identified, and the questionnaire was distributed to the

Board secretary of each charity via email. The questionnaire was sent to 160 charities, with 107 useable responses received, this was a 67% response rate.

### Analysing the Survey Data

Once data was collected via the questionnaire it was inputted into SPSS. The data was analysed using a simple statistical tests. One of the specific purposes of the survey was to investigate the level of isomorphism in charity governance. Descriptive statistics, such as standard deviations are highlighted by Di Maggio and Powell (1983) as a key indicator of isomorphism:

Since the effect of institutional isomorphism is homogenisation, the best indicator of isomorphic change is a decrease in variation and diversity, which could be measured by lower standard deviations of the values of selected indicators in a set of organizations (Di Maggio and Powell (op cit. p. 155).

For the purposes of analysis, charities were split into three broad categories based on charitable purpose, these were: advancement of education, provision of housing, and those concerned with health, social care, and community development. Three organisations were excluded as they didn't fit these categories, bringing the total number of useable responses to 104. Due to the small size of the population, more sophisticated techniques may have been of limited value. In addition, emphasis was placed on depth of analysis gained from combining quantitative and qualitative findings, rather than sophisticated statistical analysis.

#### **5.3.3** Case Study Research

The second stage of this research involved conducting case studies. Where the first stage of the research – the survey – allowed for the investigation of the existence of isomorphism (Di Maggio & Powell, 1983), the nature of case study research allowed the in-depth investigation of the functioning of the board, and the role of business

people on charity boards, and was particularly useful in investigating whether they were used for legitimation (Meyer & Rowan, 1977). The case studies presented an opportunity to gather data on how governance works in the charity context, how boards function, and to develop a deeper understanding of the role of business people on charity boards – whether their ideas penetrated the organisation, or whether they had a more ceremonial or symbolic role (Meyer & Rowan, 1977). Also of interest were charities that don't have business people on their board in order to determine why they had resisted such isomorphic pressures. The conduct of case studies also allowed for the testing of the validity and reliability of the questionnaire in the first stage, as well as allowing the development of a deeper understanding of governance in Scottish charities.

In this research it was necessary to scrutinise the behaviour of organisations from internal/ external published data. It was also necessary to study the interactions between individuals within the organisation and the kinds of information flows which inform their actions, for example, between the board and management. In addition, as part of this investigation, it was necessary to study and understand the motivation and behaviour of key individuals and the impact that modern management methods and external pressures have on their behaviour. Given the sensitivity of the above topic and the complexity of the organisations, it was necessary to explore these phenomena in-depth at different study settings. Therefore, this research used a multiple-case study approach to investigate these different aspects of the governance of charities. In order to understand such dimensions, it was necessary to engage with key actors within these organisations, to ascertain what they know, see, and understand (Silverman, 2001). This stage of research gathered data by conducting semi-structured interviews with key governance actors in the organisation and also through available documents.

Case study sites were selected on the basis of the survey responses. As outlined in the survey analysis section, charities were grouped into three areas: advancement of education, provision of housing, and health, social and community care. In addition to these three groupings having the largest numbers of charities, they have the highest income, the OSCR publication *Scottish Charities 2008* identifies four purposes which received the greatest proportion of total sector income in 2005, these are: the advancement of education, which received the largest proportion; followed by the advancement of heritage, arts, culture and science; and advancement of health and provision of accommodation which received an equal share. As, those organisations pursuing the advancement of arts, heritage, science and culture were the focus of the pilot, they were excluded from the main study. This is a study of the largest charities in Scotland as defined by total annual sector income, one charity from each of these areas was studied in depth. This cross-sectoral analysis allowed a more detailed study of the issues within the charities sector as a whole rather than just one part of it and it allowed for comparisons to be made between sectors.

The previous section discussed quantitative research and highlighted the dominant position which it has held in voluntary sector research. In response to this trend, a report by the Charities Aid Foundation (2005) - which focuses on conducting research in the voluntary sector - calls for an increase in qualitative research to redress this balance. In this report Pharoah (2005) notes the following:

If the voluntary sector is to be fully illuminated, a range of research approaches is needed to redress what is currently an imbalance. Much of the most influential and best-funded voluntary sector research throughout the 1990s was...almost exclusively quantitative...The voluntary sector has an informal and complex nature, and is highly engaged with people in many different ways — as members, volunteers, beneficiaries, activists and donors. The use of qualitative research techniques will be increasingly essential if its social meaning and impact are to be grasped (Pharoah, (op cit.) p. viii).

A number of articles in this publication specifically call for an increased amount of case study research. For example, Glasby and Manthorpe (2005) suggest how the case study can be used to explore how the organisation has evolved and been shaped, helping to gain an insight into the complex nature of the organisation and the sector in which it sits. The in-depth study of a specific organisation will also give the potential for a close relationship to develop between the researcher and the researched in a case study organisation, Russell and Scott (2005) note that this may

allow truths and stories to emerge which otherwise may have been held back. Case study research permits investigation of the past and the present, it allows analysis of continuity and change over time, it allows the detailed study of specific phenomenon and the relationship and interaction between the phenomenon being studied and the context it is set in (Glasby & Manthorpe, 2005).

The case study approach allows in-depth studies of organisations to be carried out and also allows for triangulation (Denzin, 1978, Yin, 2003). Keen and Packwood (1997) also highlight the importance of triangulation in helping to guarantee validity in the research. The case study approach provides a richness of data and depth of understanding which is not apparent from other methods. Yin (2003) calls for in depth investigations of phenomena to fully understand them, and also advocates where appropriate, the use of multiple case studies.

One of the main issues of using case studies highlighted by Bryman (1989) is generalizability and whether this can actually be derived from just one case. However, this study will include 3 cases, this will act to improve generalizability and allow comparisons to be made which make it easier to identify specific features of the cases; the use of more than one case also improves validity. The variety of methods used also allows research questions to be answered through methods appropriate to each question. This study is strengthened by the use of three case studies which increases the depth of the study and also increases validity; comparisons will be made between sectors also.

As noted above, the case study approach allows a number of methods to be used to collect different types of evidence. This study will include the analysis of documentation, and semi-structured interviews. These are outlined below.

## Documentary Analysis

As part of the case study approach which will be employed, an analysis of both internal and external documentation will be undertaken for each case. Some of the problems of using this type of research method are highlighted by Glasby and

Manthorpe (2005) who note that formal records may not capture the difficulties and tensions within the organisation. However in this instance, the use of documents will be backed up by in-depth interviews with key individuals.

#### *Interviews*

Russell and Scott (2005) consider issues surrounding the selection of interviewees, and the assumption that tends to be made by the researcher that those selected for interview will have the ability to answer any questions which are posed to them. For example Russell and Scott (2005) highlight the importance of where the interviewee is located within the organisation, in particular, location may prevent them from admitting that they lack the required knowledge that we assume them to have, for example the CEO of an organisation may find it difficult that they lack knowledge about specific aspects of their organisation (see appendix 5 for interview schedule).

Although it has been noted above that examples of case study research in the voluntary sector are sparse, there are some examples in the academic literature which focus on governance, for example Cornforth and Edwards (1999). These highlight how governance issues can be investigated through the use of this research method.

### Conducting the Case Studies

Three case study sites were identified from the questionnaires on the basis that they displayed a commitment to governance and would potentially provide interesting examples of good practice, they also had some level of business expertise on the Board, the organisations chosen were; a housing association, a social care charity, and a Further Education College. Interviews were conducted with the chief executive, the chairman and also members of the Board, in particular those from business.

### Case Study - Data Analysis

Data was analysed according to the framework analysis method (Ritchie et al, 2003). All interviews were recorded and transcribed verbatim by the researcher in order to enhance familiarisation with the data. After familiarisation, recurring themes and ideas were identified, a coding index was then developed with themes then sorted into broader categories and key themes.

### 5.4 The Research Design and Social Construction

This section will consider the research design in relation to the conceptual framework of the study – social construction. This study adopts a multi-method research design with the use of a survey and a series of case studies, allowing for between-method triangulation. Denzin (1978) highlights that with the use of between-method triangulation, one of the main principles is that the methods must be selected 'with an eye to their theoretical relevance' (Denzin, op cit. p. 303).

At the outset of this chapter, the rationale for the choice of research methods employed was outlined, the use of a survey allowed the collection of data across the complete 'organisational field' (Di Maggio & Powell, 1983). In addition, it also allowed the consideration of similarities and differences between aspects of governance and the phenomenon of business expertise on charity boards, to uncover the extent to which isomorphism was apparent (Di Maggio & Powell, op. cit).

The case studies allowed further investigation of the reasons for isomorphism and for those resisting isomorphic pressures, for example, those with few business people on their boards. Case study research enabled a more in-depth understanding of these organisations, and was used to consider whether charities recruited business people to maintain legitimacy, or whether their ideas did actually penetrate the organisation.

The table below presents papers on charity governance which also draw on social construction, these papers were considered in chapter 4, however, the focus is now on the specific research methods employed in these studies.

**Table 5.2 The Use of Specific Research Methods to Explore Social Construction on Charity Governance** 

Study	Focus	Research Method						
The Role of the Board								
Cornforth & Edwards (1999)	Contribution boards make to organizational strategy in the public and nonprofit sector.	Powell, 1983);	4 case studies (observation of board meetings, interviews, documents)					
Steane & Christie (2001)	Investigates patterns of governance on Australian nonprofit boards	Isomorphism (Di Maggio & Powell, 1983)	Questionnaire					
Parker (2007)	Examines strategic decision-making at the boardroom level of two nonprofits	& Rowan, 1977; Di Maggio & Powell, 1983)	Complete member researcher participant observation					
	The Composition							
Abzug & Galaskiewicz (2001)	legitimate their board through recruitment of educated professionals or community representatives (USA)	Isomorphism (Di Maggio & Powell, 1983)	Secondary data					
W. 1 (1000)	Board Bel							
Wood (1992)	nonprofit board behaviour. (USA)	Social Construction (Berger & Luckmann, 1966)	Interviews (80 board members & 21 executive directors)					
Board and Organizational Effectiveness								
Herman & Renz (1997)	Multiple papers on board & org effectiveness (USA)	Social construction (Berger & Luckmann, 1967)	Document reviews; Questionnaire to key					

			informants (CEO, board members, senior managers, funders;	
			interviews with CEOs	
	Organisationa	al Structure		
Leiter (2005)	Examines similarity of organizational structures of Australian nonprofits.	Maggio & Powell,	Survey	
Leiter (2008)	Examines organizational structures of nonprofit organizations in Australia and the US, and the extent to which they are similar.	Isomorphism (Di Maggio & Powell, 1983)	Survey	

The above table shows that a mixture of quantitative and qualitative methods have been used to explore charity governance using a broad social construction perspective. With the exception of Herman & Renz (2007), there are few studies which combine both quantitative and qualitative data, with none of the above studies utilising both a survey and a series of case studies. This study therefore will offer a unique perspective by combing the using of survey and case study research in the investigation of charity governance.

#### **5.5 Limitations of Research**

This section will consider the limitations of this research. In terms of the survey, as the survey instrument was in the form of a self-completion questionnaire, a level of caution must be taken in interpreting the results. The questionnaire was completed by the secretary to the Board and although this individual was chosen as it was believed they would be more dispassionate than the Chair or the Chief Executive (see

Cornforth (2001); Cornforth and Simpson (2002)) these results still may suffer from some kind of 'halo effect' as those individuals will want to portray their organisation in the best possible light. In terms of the case studies, Yin (2003) highlights issues with generalising from case study research. Furthermore, this study only focuses on the largest section of Scottish charities, therefore the results will not be relevant for the sector as a whole.

#### 5.6 Conclusion

This chapter has considered the specific design of the research. It has outlined the research questions which will be addressed in this thesis, the research methods and data collection, and has considered how the conceptual framework is reflected in the research design. This study will employ a multi-method research design, utilising a survey and also a series of case studies. The following five chapters will outline the findings of this study. In the following section, the results from the survey will be outlined, followed by result from the case studies.

### **PART TWO: FINDINGS**

#### **SURVEY RESULTS**

As described in the methodology chapter, a survey was conducted which took into consideration both current Governance procedures in charities, and also the incidence and impact of business professionals in charity Governance. In terms of the conceptual framework, the survey was designed to consider the presence of isomorphism (Di Maggio and Powell, 1983) in Scottish charities, evidence of legitimacy practices were also considered (Meyer and Rowan, 1977). The survey was also used for identification of issues which required more in-depth investigation at case study sites, and also the selection of particular cases for further study.

A self-completion questionnaire was distributed to a key informant - the board secretary - of the largest section of Scottish charities; a 67% response rate was achieved. The survey was structured into two sections to reflect the two distinct research questions of the study. The first section considered governance issues which were highlighted in the literature in chapters 1 and 2, these included: the role of the board; the distinction between strategy and operations; board effectiveness; and the regulatory framework. The results from this section are presented in chapter 6. The second section considered the impact of business expertise on charity boards. The results from this section are outlined in chapter 7.

#### **CHAPTER 6**

#### GOVERNANCE IN SCOTTISH CHARITIES – SURVEY RESULTS

This chapter will report results from the first section of the survey, which took into consideration specific issues surrounding governance in charities as identified in both the introduction in chapter 1 and the review of existing literature in chapter 2, these were: the roles and responsibilities of the board; the division between strategic and operational issues; the effectiveness of charity governance; and the control and regulation of charities. This chapter will be structured as follows: 6.1 will consider the role of the board; 6.2 will consider the distinction between strategy and operations; 6.3 will look at the effectiveness of governance in charities; 6.4 will look at control and regulation of charities; and 6.5 will conclude.

### 6.1 The Roles and Responsibilities of the Board

The roles and responsibilities of the board were considered in chapter 2, where existing research on this topic was discussed. From this, it emerged that a large proportion of this literature is prescriptive, with a number of papers outlining what the key roles of a board should be (Axelrod, 2005; Houle, 1990; Ingram, 2002). However, studies which have looked at the way in which such roles are carried out, have found that the roles and responsibilities of the board tend to lack clarity and a number of charity trustees are unclear on what their role should be (see Taylor et al, 1996; Widmer, 1993). Findings have also suggested that this may stem from the fact that individuals are unclear about what to expect when they take on the role of charity trustee (Cornfoth & Edwards, 1999; Green et al, 2001; Harris, 1989; McAdam & Gies, 1985; Widmer, 1993). Significant gaps were also reported in relation to perceived role importance, and fulfilment (Inglis et al, 1999; Green et al, 2001).

To investigate whether a similar situation existed in Scottish Charities, in the survey, three aspects were considered: (1) how well defined the roles and responsibilities are in charities; (2) how roles and responsibilities are communicated to board members; and (3) what is done to make sure that board members are aware of what their role is. These findings are now presented in turn.

## 6.1.1 Defining the Roles and Responsibilities of the Board

This section will outline the results from the survey in relation to the definition of the roles and responsibilities of the board. Respondents were initially asked to consider how well defined the roles and responsibilities of their board were and to rate them on a 5 point scale. The scale and attached value for analysis purposes was as follows:

- □ Very well defined (4)
- □ Well defined (3)
- □ Not well defined (2)
- □ Not defined (1)
- $\Box$  Don't know (0)

A one-way between groups analysis of variance was conducted to explore the impact of sector on the definition of roles and responsibilities. The results are shown in the table below:

Table 6.1 Definition of Roles and Responsibilities of the Board

Charitable Purpose		
	Mean	Standard Deviation
Education	3.53	.550
Housing	3.43	.634
Health & Social Care	3.52	.508
Total Population	3.50	.558

There was no statistically significant difference between the groups. The mean score for the total population was 3.5, suggesting that roles and responsibilities of board members in large charitable organisations are between well and very well defined. The highest mean was found in the education sector (M = 3.54), followed by health (M = 3.52) and housing (M = 3.43).

In contrast to the available literature which suggests a lack of clarity surrounding the roles and responsibilities of charity boards, large Scottish charities report that they are between well and very well defined. However, the documentation of well-defined roles and responsibilities could be a form of legitimating behaviour, with these charitable organisations seeking to present themselves to the outside world as modern and up-to-date (Meyer and Rowan, 1977).

Additional open questions were asked in order to further understand these results, these are discussed below. These relate to where these roles were outlined and also what is done by the charity to ensure that the board members understand their role.

## 6.1.2 Communication of Roles and Responsibilities of the Board

In addition to asking respondents to consider how well defined the roles and responsibilities of the Board were in their organisation, they were also asked where these roles and responsibilities were outlined, in order to provide further clarification to the previous question regarding the definition of their role. The table below outlines these results:

**Table 6.2 Location of Roles and Responsibilities** 

	Educati	ion	Housing		Health		Total	
Governance	26	60%	19	68%	14	42%	53	51%
Documents								
Other Internal	5	12%	8	29%	10	30%	23	22%
Policies/								
Documents								
Governance	6	14%	3	11%	9	27%	18	17%
Handbook								
Induction	4	9%	-	-	7	21%	11	11%
Training	2	5%	-	-	1	3%	3	3%
External	11	26%	3	11%	4	12%	18	17%
Guidance								
No Response	1	2%	1	4%	-	-	2	2%
Total	43		28		33		104	

All organisations were able to report where the roles and responsibilities of their Boards were outlined, and 59% of these organisations gave more than one example. There were six broad categories where these roles and responsibilities were outlined, namely; governance documents, other internal polices/ documents, governance handbook, induction, training, and external guidance.

## Communication of Roles: Education

In terms of Educational organisations, 60% noted that the roles and responsibilities of the board were outlined in governance documents, these included; in the constitution and standing orders, articles of governance, scheme of delegation, memorandum and articles of association. Just over a quarter highlighted they referred to external guidance for roles and responsibilities of the board, for example; the association of governing bodies of independent schools guide for governors, the schedule to the 1992 further and higher Education act, the ASC guide for college members, and Scottish Funding Council guidelines. Other policies and internal documents were also noted as places where roles and responsibilities were outlined, for example; board and committee remits and board documents, 12% of educational organisations reported that this was where duties and responsibilities were outlined.

Other examples given included; in a governance/ board members handbook (14%), or outlined in induction packs (9%) and training sessions (5%).

### Communication of Roles: Housing Associations

Housing Associations had the highest proportion of organisations that outlined the board duties and responsibilities in governance documents (68%), these included; association rules, standing orders, scheme of delegation authority, code of governance. About 29% of housing associations also referenced other internal policies and documents such as; internal management policy, operations manual, and also in job descriptions for board members. Housing Associations did not note that induction and training were where roles and responsibilities were outlined, however 11% did highlight the use of governance manuals for this. External guidance (11%) was also used as a reference to duties and responsibilities, including; Scottish Federation of Housing Associations codes of conduct and code of governance.

## Communication of Roles: Health, Social & Community Care

In terms of health, social and community care, a considerably smaller proportion of organisations referred to governance documents as a place where duties and responsibilities were outlined; however, 42% of these organisations noted that they were outlined in such documents, including, governance code, standing orders and memorandum and articles of association. Regarding other internal policies and documents, 30% noted that this was where roles and responsibilities were outlined, including, job descriptions, board papers and agendas. These organisations used governance handbooks and inductions considerably more that the other sectors; 29% of organisations outlined duties and responsibilities in a governance manual, and 21% outlined them in the induction process, 3% noted they were outlined during training. In terms of external guidance, 12% used this, examples include; OSCR

guidance for charity trustees, Institute of Directors Standards for Directors, and Charity Commission guidelines.

### Communication of Roles: Conclusion

It is apparent from the results that almost all organisations were able to provide information about the documentation of the boards roles and responsibilities. Many organisations noted that they were located in more than one place – this may go some way to providing support for the findings from the previous question which suggested that duties and responsibilities were well defined. However, it is apparent that there is no one standard location for these roles to be set out. One concern is that documents that are listed may not necessarily be documents which board members would regularly refer to, if at all, particularly formal governance documents.

The majority of charities seem to refer to internal guidance, with relatively few organisations making a reference externally. All sectors reported that the main place that roles were outlined was in formal governance documents, these however, are not necessarily documents that board members regularly refer to. Meyer and Rowan (1977) suggest that organisations adopt formal structures and blueprints for legitimation. In this case the roles outlined in these formal governance documents may function as myths and have only a ceremonial value, merely serving to present an image of the organisation in line with that expected in relation to coercive regulatory and legislative requirements. The adoption of formal documentation by charities for legitimacy purposes is found by Parker (2007).

Differences were found between charities, this is perhaps surprising due to the common environment shared by these organisations - they are all regulated by OSCR – it may have been expected that such coercive pressures would result in a similar policy for the documentation of roles and responsibilities or reference to specific documents. This may be reflected in the specific nature of the organisations in this sector. Typically, similar organisations refer to guidance produced by related bodies

– for example, Further Education Colleges reported referring to guidance from the Association of Scotlands Colleges and the Scottish Funding Council, and Housing Associations referred to the Scottish Federation of Housing Associations codes of conduct and code of governance. The organisations grouped into health and social care referred to the general OSCR guidance, this may reflect the diverse nature of the organisations within this group. Cornforth and Edwards (1999) suggest that the role of charity boards varies widely, and can be due to a number of factors including sectoral traditions and norms.

## 6.1.3 Ensuring Board Members are Aware of Roles and Responsibilities

Organisations were also asked to give details about what is done to ensure Board members are made aware of their roles and responsibilities. The results are highlighted in the table below:

Table 6.3 What is Done to Make Board Members Aware of their Roles and Responsibilities in Charities

	Educati	ion	Housing		Health		Total	
Training	14	33%	15	54%	10	30%	39	38%
Briefings	11	26%	2	7%	9	27%	22	21%
Induction	31	72%	22	79%	15	45%	68	66%
Documentation	13	30%	7	25%	4	12%	24	23%
Handbook	6	14%	2	7%	5	15%	13	13%
Away Day	2	5%	-	-	1	3%	3	3%
Self	3	7%	1	4%	2	6%	6	6%
Evaluation/								
Appraisal								
None	1	2%	-	-	-	-	1	1%
Total	43		28		33		103	

# Awareness of Roles: Education

The majority of Educational organisations had an induction process for new members (72%), some organisations noted that this involved individual sessions with the Chair and the Board Secretary. On-going training was also highlighted by 33% of organisations as a way to increase awareness of the boards roles, this included both internal sessions and also external sessions, including those run by the Scottish Council for Independent Schools. Board members were also given briefings/ reminders on their duties, for example by the Chair before a Board meeting, 26% of organisations reported this activity. Roles and responsibilities were also laid out in various documents which were issued to board members, for example, various governance documents such as memorandum and articles of association, in addition, Governance handbooks were used by 14% of educational organisations. Other ways in which board members were informed included at away days, and through self evaluations and appraisals. Only one organisation noted that there are no mechanisms in place to make sure board members are aware of their duties and responsibilities.

## Awareness of Roles: Housing Associations

Housing Associations reported the biggest use of both induction and training for board members, with 79% using induction for board members and 54% having some kind of on-going training in place to make sure board members are aware of their duties and responsibilities, for example, one organisation noted that they held committee development sessions 8 times per year, another noted how they provided board members with opportunities to attend conferences and seminars to increase their knowledge and understanding on specific issues or topics and another noted that their initial training programme for new members was provided through the Scottish Federation of Housing Associations. Additional practices included briefings (7%), a Governance handbook (7%), and self evaluation and appraisal (4%).

## Awareness of Roles: Health, Social & Community Care

Health, Social and Community Care organisations made the least use of induction sessions as a way to make sure board members were aware of their roles and responsibilities, with only 45% reporting use of this process, compared to 71% and 79% for education and housing respectively. These organisations also made use of training, with 30% having some form of training, for example one organisation noted that they held two training evenings per year with further sessions organised as they are required, another highlighted that annual training sessions were held for all trustees. Briefings were also given to members in 27% of organisations; 15% of organisations issued members with a handbook and 12% used other documentation, for example one organisation noted that they give their trustees a subscription to a Governance magazine which is aimed at the charity sector which covers issues surrounding good governance practice. Others noted that duties were highlighted at away days and through self evaluation and appraisals.

# Awareness of Roles: Conclusion

Overall, in terms of the ways in which organisations made sure that board members were aware of their duties and responsibilities, some form of induction was the most popular way of doing this (65%), this was followed by training for Board members (38%). Other ways of informing board members included briefings (21%), outlining duties in documents which were distributed to board members (23%) and issuing members with a handbook (13%). In addition to this, some organisations used the board awayday as a way of informing the board, and some highlighted that they were reminded through self-evaluation and appraisal. Only one organisation noted that they did not yet have anything in place.

In terms of making sure that board members are aware of their responsibilities, there appears to be a significant reliance on the use of the induction to communicate these, however there appears to be considerably less ongoing methods which would

reiterate/ reinforce these roles. For example 38% of organisations have some sort of training on roles however we cannot be sure whether this is on-going or just an isolated event. In addition, only 13% of organisations refer to a handbook where these roles are outlined.

The importance of board development, particularly on-going development, is highlighted in the literature, and is particularly related to increased effectiveness of the board (see Brown, 2007; Holland & Jackson, 1998).

# **6.2** The Distinction Between Strategy and Operations

In chapter two, tensions were found to exist in organisations where there was a tendency for the board to spend too much time on operational activities at the expense of strategy (see Leatherwood and O'Neal (1996); Cornforth and Edwards (1999); Green et al, (2001); Widmer (1993)). However, it is widely acknowledged in both governance guidance and the academic literature that the role of the board should be predominantly strategic. In order to investigate this issue in Scottish charities, respondents were also asked to indicate how much time the board spent on strategic rather than operational activities and also to give examples of strategic and operational tasks carried out by the board. This section presents the findings from these questions.

# 6.2.1 Assessing the proportion of Board time spent on strategy and operations

Respondents were asked to estimate the proportion of time which the Board spent on Strategic rather than Operational activities. A chi-square test was conducted to consider charity sector and the proportion of time spent on strategic and operational activities, the proportion of time is outlined in the table below:

Table 6.4 Percentage time spent on strategic rather than operational activities

Charitable Purpose		
	0-49%	50-100%
Education	9.8%	90.2%
Housing	23.1%	76.9%
Health & Social Care	16.1%	83.9%
Total Population	15.3%	84.7%

The results from the Pearson Chi Square shows that these findings are not significant, meaning there is no significant difference in the proportion of board time spent on strategy between the three sectors. There are however, some interesting points to note. Housing Associations report the highest number of organisations spending below 50% of the boards time on strategy (23.1%). In Housing Associations, the high proportion of tenant members on Boards is seen as a contributing factor in the Boards involvement in activities that are operational, due to the difficulties they have in adapting to the strategic nature of the role. It has been reported that tenant members view their role as being that of a representative, tending to focus on operational housing issues, at the expense of strategic issues impacting on the entire organisation (Audit Commission, 2004). This is a potentially significant issue in Scottish Housing Associations, where around 45% of volunteer Board members in the sector are tenants (Scottish Housing Regulator, 2009).

Education has the highest number of organisations that spend more than half the boards time on strategy compared to operations (90.2%), followed by health (83.9%). It is suggested in the literature that boards vary considerably in terms of their strategic role (Steane & Christie, 2001; Cornforth & Edwards, 1999). There is some evidence of this from these findings.

# 6.2.2 Types of Strategic and Operational Activities Undertaken by the Board

In order to gain a deeper understanding of how organisations classify strategic and operational activities, and to get an insight into the types of activities the board undertakes, organisations were presented with an open question and asked to give examples of both strategic and operational tasks undertaken by the board.

# Strategy v Operations: Education

In terms of education, there were differing views on the boards involvement in strategy. Some organisations that noted that the Board should not become involved in operational matters, however, others highlighted a tension which existed here.

The nature of these tensions can be seen in the following example, where an organisation had a working group on governance which had undertaken a mapping exercise to work out what key policies were high level (i.e. Board related) and what were management. Differing views were found to exist during this exercise, particularly in relation to whether the board sees detailed operational plans, or should just be told that they have to be done. This organisation felt that this issue of the division of strategy and operations was one of the biggest issues in terms of governance, and noted that there should be training courses on the difference between governance and management. Also highlighted was that some board members are unsure of how involved they should be, in particular the issue was raised surrounding business people, there is a question here as to whether they understand the issues – they may feel they don't know enough to take part. This specific issue of business people on the Board is considered in greater detail in the next chapter.

The examples of strategic activities carried out by the Boards of educational organisations fitted mostly into three areas, namely; risk, strategic planning and monitoring, and finance. There were however, also a number of activities which were

outside of these areas. Examples given in terms of risk included risk management and strategic risk assessment. Examples given within the strategic planning and monitoring heading included approval of strategic plans, including annual and 3-5 year plans and setting the strategic direction for the organisation. There were also a number of examples regarding estates and building, such as setting/ approving the estates strategy. Some were very specific about the nature of these activities; one organisation noted that

in terms of strategic tasks there is the estates strategy, we are currently in a major estates development phase. Most of the boards tasks should be strategic, i.e. not designing the building, but approving strategies e.g. HR/finance should be key.

Another area of strategic activity was that of finance, for example approving the annual financial statements for the year end, approving budget and monitoring it against targets and also setting key performance indicators for the organisation and its faculties. Other examples of strategic activities within educational organisations include setting and approving HR strategies and ensuring compliance with the statutes, ordinances and provisions regulating the College, for example health and safety. Notably, only two organisations referred to strategic activities in terms of the future of the organisation, these were both independent schools who noted the importance of the future in terms of the shape and direction of the school.

Operational activities were more varied than the examples given for strategy. A number of organisations put examples of financial activities under the operational heading, for example setting and approving annual budgets and monitoring financial performance against budget, review of management accounts and the approval of financial statements. These activities were listed by other organisations above as strategic, which points to a lack of clarity within these organisations regarding the distinction between strategy and operations.

Risk management was another issue listed as operational by one organisation but strategic by others. Other examples of operational activities carried out by these educational organisations included: staff issues, such a issues surrounding salaries and the board acting as a final appeals panel for staff disciplinary and grievance

hearings; top level negotiations with the Funding Council and other Educational

Institutions with regard to funding, estates development and partnership with

stakeholders; management of senior management team – occasional prioritising of

operational issue such as development of Student Union premises; overview of

departmental operational plans; review of internal audit reports in relation to internal

control; and approval of senior management appointments. One organisation gave a

specific example of the divisions between strategy and operations, by highlighting

their new estates strategy which resulted in the acquisition of a new £20m building.

They note that although this was mainly strategic from the boards perspective, being

a small institution, some key Governors were involved with detailed operational

aspects such as putting the finance package in place.

Strategy v Operations: Housing Associations

As with educational organisations, there were differences in the interpretation of

strategy and operations and the types of strategic and operational tasks carried out.

The difference can be seen by the following, where HA 1 noted that:

As a HA operational matters have to be brought to the Board

however, HA 2 noted that:

No real operational tasks carried out – receiving regular reports on a variety of performance issues, new policy areas, annual policy reviews,

financial reporting etc. comprise most of business.

In terms of strategic activities undertaken by the Board, the majority of housing

associations referred to approval and review of business plans, from annual to 30

year plans, others also highlighted the approval and review of policies and strategies

and also the annual Board away day where strategy is reviewed and developed. Only

four organisations referred to finance, three highlighting the approval of the budget,

104

and one referring to financial planning; another four highlighted risk management as a strategic activity in which the Board undertook. Other areas of strategy in which the Board undertook included, rent reviews, pensions, training/ attendance at conferences, asset management and selecting the development funding proposal from a tender exercise.

The majority of operational activities referred to by housing associations were related to the measurement and management of performance, examples given included; reviewing reports on performance indicators and performance management. In terms of finance, as with education, the housing associations also had approval of budgets in the operational category as well as the strategic category. Another area which appeared as examples of both strategic and operational activities was the approval and review of policies. Other examples of operational tasks included; consideration of appeals in respect of formal complaints, approving evictions, interview/recruitment of senior staff and agreeing a job title for a post.

# Strategy v Operations: Health, Social & Community Care

Similar activities were also detailed by Health and Social Care organisations. In terms of strategy, the majority of organisations listed activities such as: approval and formulation of strategic plan and business planning. For example, one organisation noted that the:

'full organisational strategic plan is updated annually, one of the steps is to present it to Trustees for input and then final approval. Key areas are also discussed with Trustees when being reviewed e.g. fundraising approaches, business growth'.

Finance examples were given including approval of the annual budget and review of management accounts. Risk management was also noted as a strategic activity, for example one organisation highlighted the review of risk management register.

As with the other two sectors, there was a crossover in terms of certain activities; where some listed them as strategic, and others as operational. For example, approval of the annual budget was also listed as an operational activity. Other operational activities included: finance; HR issues, such as the board acting as an appeal court in staff disciplinary matters; and health and safety issues. Other examples included the size of food budgets, minor staffing issues and discussions on cleaning facilities. There were however, some organisations that noted that the Board do not carry out operational activities, for example, one noted how the 'board members do not carry out operational duties. They provide strategic direction for the management team who have operational responsibility'.

# Strategy v Operations: Conclusion

In terms of the results for the type of activity the Board undertakes, organisations reported that the work of the Board was predominantly strategic, with an average of 60-69% of Board time for the sector as a whole, and for housing and health being spent on strategy, and 70-79% for education. Of concern is the range of the results, with 15% of organisations reporting that less than 50% of the boards time is spent on strategic matters. Numerous studies report situations where the board becomes involved with operations at the expense of strategy. Leatherwood and O'Neal (1996) reported that boards were more responsible for service than for strategy or control, Steane and Christie (2001) and Taylor et al (1996) found that boards engage in operational matters of charities. Cornforth and Edwards (1999) found a significant variation in the boards role in strategy between organisations, in particular they noted it was dependent on size and sectoral norms, this is apparent here.

In order to clarify charities interpretation of strategy and operations, and to gain awareness of the types of activities the Board undertakes, respondents were asked to give examples of such activities undertaken by the Board. The results from this question were very diverse, with some organisations explicitly highlighting that the Board should not be involved in operations, with others listing low level tasks in

which the Board is involved. In terms of strategic activities, all three sectors reported widely the Boards involvement in 'strategic planning', this however seems vague, and lacks detail. In addition, there was a lack of any mention of long-term visions for the organisation and ensuring the future safeguarding of the organisation. There was also a lack of consensus surrounding a number of tasks as to whether they were strategic or operational. For example; approval of the annual budget, approval of the annual financial statements, and risk management, were listed by some organisations as strategic, and some as operational. It is widely reported in the literature on governance and also in governance guidance, that these are roles which the Board should undertake. Therefore, there is possibly a classification issue with some activities.

Other activities which appeared in both strategic and operational categories were HR issues and health and safety issues; these are quite vague descriptions and the specific nature of these tasks may be a matter of judgement as to whether the Board should be involved or not. For example; in terms of HR issues, one example was approval of senior management appointments, this was listed as operational, however, it is an activity which is appropriate for the Board to undertake and could have a strategic impact. Another issue was concerning tasks which were listed as operational activities the Board undertook, however they seemed to be lower level activities which could have been dealt with by management rather than taking up Board time. For example, activities such as agreeing a job title for a post, cheque approval, minor staffing issues and discussions on cleaning facilities could be dealt with away from the Board.

The findings here surrounding the lack of clarity on strategic and operational issues may have implications in terms of the results for the previous question concerning the amount of time spent on strategic rather than operational activities. In addition, the results here also help interpret the questions surrounding the role of the Board. The earlier section reported that the duties and responsibilities of the Board were well defined, however, the findings here would point towards either an issue with the definition of duties and responsibilities or an existence of a gap between the duties of

the board and what the Board actually does. This is consistent with the findings of Inglis et al (1999) who found a gap between role importance and role fulfilment. Also Widmer (1993) who highlights differences in the way that board members perceive or carry out their roles. This would also support the idea that formal structures such as documents outlining the role of the board are decoupled from the actual activities the board carries out (Meyer and Rowan, 1977; Parker, 2007)

The next section will now look at the effectiveness of governance in these charitable organisations.

#### **6.3** The Effectiveness of Governance in Charities

This section considers the effectiveness of governance in charities, by asking respondents to rate the effectiveness of governance in their charities, and also to consider what could be done to improve governance in their organisation.

# 6.3.1 How Effective Governance is in Charities

Respondents were also asked to rate the effectiveness of their charity, on a scale, which was as follows:

- □ very effective (4)
- □ effective (3)
- □ ineffective (2)
- □ very ineffective (1)
- $\Box$  don't know (0)

A one-way between groups ANOVA was carried out, the results are shown in the table below:

Table 6.5 Effectiveness of Governance in Charities.

Charitable Purpose		
	Mean	Standard Deviation
Education	3.57	.501
Housing	3.56	.506
Health & Social Care	3.23	.762
Total Population	3.46	.610

A one-way between-groups analysis of variance was conducted to explore the impact of sector on charities perception of how effective governance is in their organisation. The total mean value was 3.46, which means that charities feel that governance is between effective and very effective in their organisation. No statistically significant difference was found between the groups.

# 6.3.2 Improving Effectiveness of Charity Governance

These organisations were also asked whether anything could be done to improve governance in their charity. These results are outlined below.

# Improving Governance Effectiveness: Education

Of the Educational organisations only three noted that there was nothing that needed to be done to improve governance. One organisation felt that governance would be improved if the burden on charities was reduced by external bodies:

The greatest challenge is to keep governors informed without overwhelming them with detail. The scale of regulation of Schools, child care and charity finance makes informal decision making very demanding. The consequences for trustees of making serious mistakes is demoralising and puts off people from becoming trustees. Less regulation would help.

A number of other organisations highlighted that there was always room for improvement, however this was dependent on a range of factors, such as the time and expertise of trustees and regulation. About one-third of these education charities highlighted specific areas where governance could be improved in their organisation, these tended to be internally focused as opposed to dependent on external factors such as regulation. These improvements were varied in nature, examples included: outside audit of governance by a third party; formal business planning and risk mapping; better budgetary control; more regular self-evaluation; benchmarking; another highlighted the earlier issue of trying to clarify management and governance. Four organisations highlighted the need for increased training for board members, and another noted that they were currently going through an exercise to improve and update governance particulars for recruiting board members. In addition, one third noted that they were either currently or planning to undertake a governance review. Those that had undertaken such a review noted they were acting on recommendations, however not all noted what these recommendations were. Of those that did, one noted recent improvements had included the appointment of an independent clerk to the board, raising the profile of board members such as their attendance at events and increased openness and transparency such as information on the intranet and web.

# Improving Governance Effectiveness: Housing

With regards to housing associations, just over 20% felt that no improvements were needed to the governance of their organisation. Others indicated improvements which could be enacted by regulatory bodies through less regulation. The majority of Housing Associations did list some kind of improvement which could be made to their organisation, these tended to fall into two categories; training and the structure and composition of the Board. In terms of training, examples included induction training and training on roles and responsibilities of trustees. With regards to the structure, one organisation noted that:

we are considering revising our sub-committee structure to give more of a business focus on the day-to-day management, and to clearly set time aside for strategic issues, probably through a dedicated Strategy subcommittee.

A number of Housing Associations also expressed concerns over the composition of their Board and highlighted the problems they face in recruiting new Board members, for example, one organisation noted that:

Attracting voluntary Board members is becoming increasingly difficult and if we could find a way of making meetings shorter and papers more succinct this might help. It would also help if some of the requirements imposed on us by our main regulator — Communities Scotland — were lessened. We spend quite a lot of Board time ensuring compliance.

Other organisations expressed concerns over the composition of their Board, for example HA 1 noted:

There is an issue with the make-up of the governing body in terms of equality. We have the wrong gender balance and the ethnic minorities are poorly represented. There are also not enough tenants.

#### And HA 2 noted:

In common with many other organisations there could be improved governance. As an example we do not have a Board member with a legal or property development experience which would bring enhanced levels of skills to the decision making forum of the Association.

Other Associations noted that self-appraisal and internal audits helped to ensure good governance, for example, one noted that 'a recent internal audit report on Governance suggested a formal letter of appointment for Board members, which we will implement'.

# Improving Governance Effectiveness: Health, Social & Community Care

Finally, improvements to governance in health and social care organisations are considered. The proportion of these organisations which explicitly stated that no improvements were needed was about 12%, for example, one responded 'no, since our Trustees discharge their responsibilities and duties conscientiously'. In contrast, another set of organisations felt that there was always room for improvement, however this could depend on factors such as the time in which individual Trustees can dedicate to the charity and external factors such as regulation.

In terms of specific improvements, there were a number surrounding the division between strategy and operations. For example, one organisation noted how 'council members could be better informed and a small minority reminded from time to time that they are not responsible for operational decisions', another noted how governance could be improved by 'improving the definitions on the strategic role of the Board' and similarly, 'directors have limited time, especially if not retired, so we cannot overdo meetings, training, visits to services etc. But also important not to impede direct management and decision making within agreed parameters'.

In common with the other two sectors, improved training for board members was recognised as a way of improving governance and also changes to the composition and structure of the board and committees. For example, one organisation noted how governance could be improved with 'easier access to suitably qualified and experienced Trustees – difficult to find/ attract right people who can commit required time to organisation', another highlighted that a 'change in the board make-up to allow more business members and less council influence may help', others would like to see a reduction in Board size. In terms of the committees, one organisation noted how 'beefing up the Audit Committee this summer with high powered company finance directors/ managers will focus on financial risk factors'. Other organisations highlighted recent governance reviews to identify issues, although they failed to give detail as to what they were.

# Improving Governance Effectiveness: Conclusion

The majority of organisations in all sectors reported that some improvements could be made to governance in their organisations. Although a small number referred to the impact of external factors such as regulation the majority of desired improvements were internal, categorised broadly into board structure and composition; and training.

These findings can be related to a previous question which reported that roles and responsibilities were well defined and outlined and also activities such as training were undertaken to ensure that Board members were aware of what was expected of them. However, in light of the confusion over the division of strategic and operational activities expressed in these sectors and the large proportion of time spent on operational tasks, this would point to less than effective governance.

In addition, the majority of organisations did give examples of some ways in which governance could be improved in their organisation, some examples were quite significant; such as issues surrounding the structure and composition of the Board. Previous studies, such as Cornforth (2001) suggest that board effectiveness is dependent on factors such as whether board members have the time, skills and experience to do the job, clear roles and responsibilities, some of which were found not to exist in these organisations. For example, it has already been noted there may be an issue surrounding the clarity of roles and responsibilities and in addition, some organisations highlighted the difficulties they have in recruiting board members.

A number of factors highlighted in the literature as impacting on the effectiveness of nonprofit boards were not investigated in the survey, but will however be discussed in the second part of the survey, which considers the composition of the board, identified by Provan (1980) and Callen et al (2003) as having an impact on board effectiveness. In addition, the case study will consider the impact of board staff

relations (Herman and Tulipana, 1985), and board development practices (Holland and Jackson, 1998; and Brown, 2007).

# **6.4 Control and Regulation in Charities**

Another significant issue which was identified in the introduction as particularly pertinent to the Scottish context was that of the control and regulation of charities.

# 6.4.1. Charity Views Concerning the Nature of Regulation

These organisations were asked to rate the nature of regulation using a scale:

- □ very excessive (4)
- $\Box$  excessive (3)
- $\Box$  not very excessive (2)
- $\Box$  not excessive (1)
- $\Box$  don't know (0)

A one-way between groups ANOVA was conducted, and findings are shown in the table below:

**Table 6.6 Control/ Regulation of Charities** 

Charitable Purpose		
	Mean	Standard Deviation
Education	1.76	1.165
Housing	2.19	.895
Health & Social Care	1.37	.809
Total Population	1.76	1.036

A one-way between-groups analysis of variance was conducted to explore the impact of charitable purpose on charities views on regulation. The population was divided into three groups according to sector. There was a statistically significant difference

at the p<.05 level for the three groups (p=.011). The main difference was between

housing, and health, the mean for housing was 2.19 meaning that housing charities

viewed regulation as being between not very excessive and excessive, this was the

highest mean for all three sectors, suggesting that housing charities felt that

regulation of charities is more excessive than the other two sectors. The mean for

health was 1.37, indicating that these organisations felt that regulation was between

not excessive and not very excessive.

6.4.2 Charities Justification for Views on Nature of Regulation

In order to gain an understanding as to why organisations had a certain view of

control/ regulation, they were also asked to explain why they had given the rating

that they did.

Views on Regulation: Education

Of the charities with an educational purpose, there were 5% which viewed regulation

as very excessive. The reasons given here were firstly; that there were too many

audits from Government Departments such as OSCR, SFC, Audit Scotland and

HMIE. Secondly, was the issue of recent regulation, for example:

...the detail has been badly thought through and the bureaucratic process and blind obedience to rules set out in highly prescriptive

legislation means that genuine organisations have an increased burden – with of course no guarantee that loss of charitable status would prevent

the financial abuses within private trusts. In other words the baby has

been thrown out of the bath water.

One quarter of respondents from the education sector felt that regulation was

excessive in nature, mainly due to the requirements they were expected to meet.

Examples include: 'requires duplicate efforts in accounting, reporting, often extra

115

hurdle for no gain'; another noted 'generally the level of bureaucracy that the College has to engage with creates a significant burden on cost which means funding is not always directed to front line services', others felt that not all was necessary, for example the best practice guidance by OSCR is not appropriate to Colleges, another felt that many external regulatory bodies often had conflicting requirements.

In contrast to these views, over half of these organisations actually felt that control/regulation was either not very excessive or not excessive. Of those who rated regulation as not very excessive, many felt that requirements were not overly burdensome, for example, one noted that the 'charities regulator requires an annual return which can easily be completed form College annual accounts' and another noted that 'we are reporting on most of the issues and providing data anyway'. Another organisation highlighted the importance of regulation:

while it is another layer of bureaucracy and legislation for Colleges to comply with, and in some respects this differentiates Colleges from some of the other charities operating, it is important that there is a proper regulatory authority and regime in place.

Reasons given by educational charities as to why control/ regulation of charities is not excessive were mainly due to the acknowledgement of the importance of this in order to ensure proper conduct of charities, for example one organisation stated that increased regulation of Scottish charities was long overdue, another also noted:

I think that charities play a key contribution to society in Scotland (and elsewhere in the UK) and it is crucial that they are properly led and managed and that their conduct is beyond reproach. For that reason, regulation is essential and a help to proper management.

Of those organisations who answered that they were unsure as to whether regulation was excessive or not, some felt that it may be too early to tell in terms of OSCR.

Views on Regulation: Housing Associations

Earlier results show that the highest sector mean was for housing associations,

indicating that these organisations felt that regulation was the most excessive. One

organisation felt that it was very excessive, noting that:

As well as OSCR, Registered Social Landlords are supervised very

closely by Communities Scotland in their role as regulator. We also provide information to the Inland Revenue, the Financial Services

Authority. Many Housing Associations are also supervised by the Care

Commission (among others).

The proportion of those who rated regulation as excessive was 37%. The main reason

for this was indicated as the number of bodies these organisations are regulated by.

For example, one noted that issue of 'too many regulators many of whom have

conflicting views. OSCR, Communities Scotland, FSA, HMRC'. This view was

echoed by other organisations.

Just over 55% of associations felt that that control/ regulation was either not very

excessive or not excessive. Of these 41% felt that it was not very excessive, reasons

included the fact that associations were dealing with public money and personal

donations and therefore needed to be accountable, some also noted that regulation

was not onerous. Others stated that it was not excessive, for example, one noted 'I

believe that the level of regulation by OSCR is proportionate and generally does not

overlap/ conflict with the role of Communities Scotland'.

Views on Regulation: Health, Social and Community Care

Health and Social care organisations had the lowest mean score in terms of their

views on regulations, indicating that they feel it is less excessive than the other two

sectors. None of these organisations reported that it was very excessive, and only

10% rated it excessive.

117

Almost three quarters of these organisations felt that control/ regulation was either

not very excessive or not excessive. The reasons given were similar for both, and

centred around the need for regulation to ensure charities remained accountable for

their actions, for example:

controls and regulation are necessary to maintain public confidence and

compliance with them is not particularly onerous;

As we utilise public funds we need to be accountable for how we use

them and be seen to operate within our rules and objects;

the charities sector has historically been under regulated and it is

important to give the public confidence in the level of regulation and

management;

Given some of the cases of abuse of funds that have arisen over the past

few years additional measures were required. As a larger organisation we have been able to cope with these measures – probably smaller

organisations may struggle with some of the requirements and the cost

involved.

The next section considers specifically the impact of OSCR on the governance of

Scottish charities.

6.4.3 The Impact of OSCR on Charity Governance

This section specifically considers the impact of OSCR on charity governance.

Impact of OSCR: Education

Respondents were also specifically asked the impact in which the advent of OSCR

has had on the way in which their charity was governed. One of the biggest issues

has been in the Further Education sector, where, after a review of John Wheatley

College, OSCR concluded that Colleges did not pass the charity test, and a question

118

mark was cast over the future of their charitable status; although there are now steps being taken to resolve this issue. However, a number of FE Colleges highlighted this issue, one organisation commented on OSCR:

The impact is serious on the FE sector across the Board. Following a decision involving Wheatley College – a genuine College operating for the public good and whose financial dealings were impeccable – it was decided that the College did not meet the statutory criteria. The VAT implications were profound as loss of charitable status casts the financial viability of the College... The wheatley decision means that all FE Colleges in Scotland now stand to lose their Charitable Status ... The issue remains unresolved but loss of VAT exemption would be catastrophic for organisations whose public funding barely keeps pace with inflation

Other consequences of OSCR included increased responsibilities being placed on Boards, which can be duplicatory, increased time required to ensure compliance and more 'form filling'. One University felt that OSCR doesn't map well into the University setting, one issue here is duplication, where these institutions may have to produce annual reports for both the Funding Council and OSCR. Other examples include additional reporting and making sure that the organisation will pass the 'charity test', another highlighted increased bureaucracy, however, noted that this was acceptable provided that public confidence was maintained. Another noted that:

It has led board members to review their obligations as Charity Trustees and it has led me as Chief Executive to review the manner in which I present material and matters for decision to them. The Scottish Council for Independent Schools has provided excellent support in understanding the new legislation.

Another organisation which highlighted the positive impact of OSCR, noted:

OSCR has had a good impact as all the members of the Board are now aware that we are a charity and also about Governance issues. So impact has been positive overall.

Almost 28% of respondents felt that OSCR had little or no impact on the way in which their charity was governed.

## Impact of OSCR: Housing Associations

In terms of Housing Associations, 58% of these organisations felt that OSCR had had little or no impact on the way in which their organisation was governed, some attributed this to the limited amount of contact they had had with OSCR so far, others highlighted the agreement OSCR has in place with Communities Scotland which meant they would be more light touch. Others noted they were required to make additional returns and additional duties had been placed on trustees such as the need to ensure the organisations activities do not infringe on its charitable status. Also noted was the requirement in some cases to set up separate undertakings for non-charitable activity such as shared equity housing. Another association noted how they had felt the impact more severely through accounting measures, noting 'more stringent rules on accounts – we had trouble with these'.

# Impact of OSCR: Health, Social and Community Care

In terms of health and social care organisations, over half felt that OSCR had little or no impact. The main reason for this appears to be due to these organisations feeling that they already had strong governance systems in place before the advent of OSCR, therefore little has needed to be changed, for example, one noted 'minimal impact as governance was already strong before OSCR was instituted'. Just below half of these organisations noted the positive impact in which OSCR had had, for example 'it has emphasised the need for clear transparent and robust governance arrangements', another stated:

whilst we remain focused on our charitable objectives, OSCR provides a useful framework to check our focus is in the right place, to measure the public benefit impact we have an to make sure governance is more than adequate and effective. The main impact has therefore been to highlight the importance of good governance and to make sure it continues to be strong.

Only two of these organisations seemed to have a negative attitude towards OSCR,

one noted 'we waste time doing OSCRs return as well as our accounts' and another

noted that it had 'doubled the reporting to regulatory bodies'.

Control and Regulation in Charities: Conclusion

With regards to control and regulation of charities, in contrast to the expected results,

charities actually felt that this was between not excessive and not very excessive. The

main reasons given for this were that they felt it was not particularly burdensome and

was important to ensure proper conduct of charities. Of those that felt that it was

excessive in nature, they noted the existence of multiple regulators and increasing

requirements, such as those relating to accounting. The range of answers for this

question ranged from not excessive to very excessive.

Organisations were also asked specifically to comment on the impact in which

OSCR had on governance in their organisation. The impact of OSCR seemed to be

felt mostly in the education sector, this was mainly with FE Colleges, who expressed

concerns over the possibility of losing their charitable status, and also duplications in

reporting. Many reported little or no impact, 28% of educational organisations

reported that OSCR had little or no impact, in the other sectors this was significantly

higher, with 58% of housing associations and over 50% of healthcare organisations

reporting similar views.

**6.5 Conclusion: Governance Survey Results** 

On the surface the results appear to show the existence of strong governance

frameworks; with organisations reporting that the duties and responsibilities are

clearly defined, the board concerns itself with predominantly strategic issues and

governance is effective in their organisations. However, looking beyond this, it is

apparent that this may not be the case that it is not quite as simple as this.

121

Respondents indicated that roles and responsibilities were well defined, this is in contrast to the literature. Explanations for this may be that charities document their roles in order to present themselves as legitimate to the outside world. Communication of roles was predominantly through formal governance documentation, it could be suggested that these are not necessarily regularly referred to by board members, providing further evidence for the documentation of roles for legitimacy purposes. Meyer and Rowan (1977) suggest that organisations adopt formal structures ceremoniously, with such formal guidance functioning as myths, merely serving to present an image of the organisation in line with regulatory expectations (Parker, 2007).

A lack of clarity was found to exist in terms of strategy and operations, and low level activities were also shown to be carried out by the Board. In addition, organisations highlighted a number of ways governance could be improved and also issues adhering to regulations were raised, suggesting that governance in these organisations may not be as effective as reported. These findings point towards the existence of loosely coupled organisations (Meyer and Rowan, 1977), where the external façade of the organisation, namely the formal rules of the organisation such as the documented duties and responsibilities conflict with the inner core of the organisation and the work that is actually being carried out. Organisations document and report adhering to prescribed governance practices in order to maintain legitimacy (Meyer and Rowan, 1977).

The findings from this section of the survey highlight a number of issues which Scottish charities face in governing their organisations, this is in line with the current literature. However, as outlined in chapter 5, there are limitations with regard to the nature of survey research. Although this survey has allowed for the initial investigation and identification of these issues, it will be necessary to considers these in the more in-depth setting offered by case studies. However, before this is considered, the results from the second part of the survey will be presented.

#### **CHAPTER 7**

# THE IMPACT OF BUSINESS EXPERTISE ON CHARITY BOARDS – SURVEY RESULTS

The previous chapter reported partial findings from a survey of governance in large Scottish charities. This section will focus on the remaining set of results, which consider specifically the impact of business expertise on charity boards. The key areas investigated here are: (7.1) the proportion of business people serving on charity Boards (Abzug & Galaskiewicz, 2001) and the satisfaction of charities with these numbers; (7.2) examples of business practices which have been introduced to the organisation by business members and how they were received (Landsberg, 2004; Dees, 1998); (7.3) the motivations of business people serving on charity boards (Austin, 1998) and their understanding of the specific context of the charity (Bowen, 2004; McFarlan, 1999); and (7.4) whether charities feel the inclusion of business people enhances public confidence in charities.

#### 7.1 The Presence of Business People on Charity Boards

This section considers the composition of the Boards of the largest section of Scottish charities, focusing specifically on the proportion of business people sitting on charity Boards.

# 7.1.1 Proportion of Business People on Charity Boards

The first set of findings which will be reported surround the composition of charitable boards, of particular interest is the proportion of business people who sit on these boards. This is outlined in the table below:

**Table 7.1 Percentage of Business People on Charity Boards** 

	Percentage of Business People on Charity Boards				
	0	1-25%	26-50%	51-75%	76-100%
Education	0	12%	49%	37%	2%
Housing	17%	25%	17%	25%	17%
Health	7%	18%	61%	7%	7%
Total	6%	17%	44%	25%	7%

The above table sets out the percentage of business people who sit on charity boards. In terms of the total population, 96% of charities reported having at least one business person on their board; with 76% of charities having their board composed of at least one-quarter business people. Turning to the three specific sectors in the study, in terms of Education, all organisations reported the presence of business people on their boards. Almost half noted that the proportion of business people on their board was between 26% and 50%, with only 2% having a board made up of between 76-100% business people. Housing associations had the highest percentage of organisations with no business people on their boards. The majority of healthcare organisations also had boards which had between 25 and 50% business people.

This tendency for charities to recruit business people to their board is what Di Maggio and Powell (1983) term isomorphism, where organisations become similar to each other. This similarity of charity boards having large proportions of business people on them can be explained through specific isomorphic pressures, here, coercive isomorphism, where the external influence of the charity regulator, OSCR can be linked to this uptake of business people, with its pressures on charities to be run in a more 'business-like' way. An example of such pressures can be seen in an OSCR report which investigates the failure of the Scottish charity one-plus in January 2007 (OSCR, 2008), here we can clearly see the emphasis on the importance of charities being run in a business-like way;

The board did not appear to contain adequate skills and independence of thinking to reflect the needs of a multi-million-pound business... While all of the directors were enthusiastic about the charity and the services it

provided, they did not seem to pay sufficient attention to the business aspects and support structures required to keep expanding services operating more effectively. (OSCR, 2008 p.6)

This focus on the charity as a 'business' and on 'operating more effectively' could explain the widespread recruitment of business people to charity boards. However, there is also the possibility that charities will recruit business people in order to appear legitimate (Meyer and Rowan, 1977), here, the presence of these business people on the board makes the charity appear to the outside world as modern and upto-date.

Abzug and Galskiewicz (2001) consider this issue of the use of individuals with highly educated/ management backgrounds to provide legitimacy on nonprofit boards. They find high levels of isomorphism between nonprofits over a period of time and suggest that nonprofits increasingly recruit these individuals to legitimate their boards. However, they also find a selection of organisations that legitimate their board through community representation. Such a finding may help to interpret the differences in the above results between the three sectors, for example, in terms of housing associations. These organisations had the highest proportion of organisations with no business people on their boards - they face significant pressure from their primary regulator – the Scottish Housing Regulator to recruit tenant members onto their boards, at present 45% of the individuals on housing association boards are tenant members (Scottish Housing Regulator, 2009). This may explain why housing associations have the highest percentage of organisations with no business people on their boards.

# 7.1.2 Charities satisfaction with the Proportion of Business People on their Boards

Charities were also asked how they felt about the proportion of business people which they had serving on their board. The table below outlines these responses:

Table 7.2 Charities Satisfaction with the Proportion of Business People Serving on their Board

	Satisfaction with business people		
Sector	yes	no	
Education	93%	7%	
Health	75%	25%	
Housing	83%	17%	
Total	85%	15%	

The overwhelming majority of respondents reported that they were satisfied with the proportion of business people, with 85% of the total population expressing that they were satisfied. Such findings are consistent with Austin (1998) who considers the presence of business people on US boards and finds that their presence is positively received. This was especially true in education, with 93% expressing this view and only 7% of educational organisations noting they were not satisfied with the number of business people on their board. As noted above, education was the only section in which every organisation had business people on their board. Of those organisations that were not satisfied, one had 37% of their Board made up of business people, one had 55%, and another had 33%. Interestingly, the latter organisation was able to specifically identify an area of business expertise which it believed would enhance its board, this organisation specifically noted that they would like an individual with risk management expertise. This clearly highlights their desire for more business people on their board.

With regards to healthcare, 75% were satisfied with the proportion of business people on their board. Of those that were happy, one noted that 'we have the right balance – we look for a range of skills' and another noted the importance of specific sector knowledge over business expertise:

we have problems getting people to sit on the board due to technicalities of the organisation. 3/9 people have worked in business. However, they do not sit on the board for this reason and business experience is not essential. Knowledge of the specific care sector is better to have. Don't see the need for any more business people. However we would like a lawyer who specialises in contract law – but this is for our specific purpose.

Of those that were not satisfied, one organisation specifically highlighted a reason for low numbers of business people as a reluctance to openly advertise for board members as part of their recruitment. Another specifically highlighted concerns over the ability of business people to adapt to the specific nature of the sector, noting that 'although it would be good to have more people on our board with business skills (finance, HR etc) they also need to have the ability to know how to apply them in a charitable business'. Another charity noted the benefit they felt that business people could bring to the charity; 'I would like more business people on our Board as I think it aids quality decision making. The idea is to have family members who are also business people'! However, the fact that they specify the desire for business people who are family members (of beneficiaries) may point to concerns that a business person who is not attached to the charity in this way may have difficulty understanding its specific nature.

In terms of the proportion of business people on these dissatisfied boards, this ranged from 0 to 44%. The two latter responses points to issues with these individuals understanding the specific nature of the charities whose boards they sit. This is consistent with the views of Bowen (1994) and Landsberg (2004) who suggest that business people lack an understanding of charitable organisations; although concerns regarding this issue were raised, these views are clearly in the minority as such a large number of organisations reported their satisfaction with the high proportions of business people on their boards. This issue is discussed later in this chapter.

Eighty three percent of Housing Associations reported satisfaction with the proportion of business people on their boards, with 17% expressing dissatisfaction with the composition; the range of business people on these boards was between 33% and 90%. Of those that were not satisfied, reasons for this included:

not enough unfortunately. We also have two sub-committees and require more members for these. We have just finished a recruitment for voluntary committee members which has been very successful and we will be interviewing quite soon, so that should replenish our numbers quite well, hopefully.

Another highlight the rules regarding the make-up of the board which were restrictive for this:

there is a maximum limit of 6 set for non tenant members with another set aside for a representative from x council.

These findings highlight the sector-wide recruitment of business people onto charity boards, and the significant numbers on these boards. These results, coupled with the high levels of satisfaction of charities with the numbers of business people on their boards appear to support the apparent desire of charities to have business people on their boards. This is supported further by the comments from those organisations who reported a dissatisfaction with the numbers of business people on their board, in that, they generally stayed that they wished for a larger number. An interesting finding was also reported from an educational organisation, who had specifically pinpointed an area of business expertise they required in terms of risk management; this raises an interesting question as to whether these organisations recruit business people to enhance specific areas, or whether they are recruited for their general expertise.

However, there are also organisations who express concerns about the intrusion of business people and practices into charities. This is consistent with the work of Myers and Sacks (2003) who problematize this issue, and state that such influences should be met with caution.

# 7.2 Impact of Business People

This section considers practices introduced to charities by Board members with business expertise, and how they are received.

# 7.2.1 Practices Introduced by Business People

After establishing the proportion of business people sitting on nonprofit boards, organisations were then asked to give examples of specific practices which they had introduced. These are summarised in the table blow.

**Table 7.3 Examples of Business Practices Introduced by Business Members** 

	Proportion of Organisations			
	Housing	Education	Health	Total
			& Social	
Type of Examples				
General Expertise	17%	21%	14%	18%
Specific Practices	63%	63%	69%	65%
No Examples	17%	16%	10%	14%
other	4%	-	7%	3%

Examples of practices which members of the board with business experience introduced into organisations tended to fall into two categories; the first was general expertise, which were reported in 18% of organisations, and the second, more functional expertise in areas such as accounting, HR, and risk management, these examples were reported in 65% of the organisations.

## Housing: Introduction of Business Practices

With regard to Housing Associations, only one organisation gave a negative example of the impact of business people, noting 'we have had people with a business background on our Board, they find it difficult to understand the not-for-profit bit', views such as this are also noted by Bowen (1994) and Landsberg (2004). Of those organisations that said that business people have not introduced practices, this was 17%. The proportion of organisations which felt that business people offered a general business expertise to the Board was 17%, for example, one noted how:

Board members bring knowledge and expertise to the Governing body. As a result of their constructive comments and questions that influences the Policy and Strategic direction of the Association rather than a direct impact on which we follow.

In terms of more specific business practices, 63% of charities reported these examples, which fell into a number of categories, including; accounting and finance, audit, performance management and reporting, risk management, human resources, and governance. Examples of the use of accounting and finance expertise included the 'ability to understand and if necessary challenge the Associations financial statements and the views of the internal and external auditors'. In terms of audit, one noted how business people had 'a more focused and outcome-centred approach to internal audit. A tendency to want quicker decisions and cut through red tape'. A general enhancement of performance reporting was also noted in a number of organisations including 'clearer provision of information' and reporting to business standards', another highlighted the use of traffic light systems in reporting (R(ed) A(mber) G(reen) reporting) and also the development of some KPIs. Risk management was another area which was influenced by business people, one organisation noted that 'they have contributed immensely to developing risk assessment', another risk management had been improved and was based on a model at the Health service. Improvements to HR practices and general Governance process were also given as examples.

## Education – Introduction of Business Practices

Similarly to the Housing sector, Education organisations gave examples which were mostly focused on the functional expertise of business people. About 21% of these organisations highlighted the general expertise which these business people brought with them. For example, one organisation noted:

No specific examples. However, they do bring an awareness of how to become more business-like, for example, how to take more controlled risks, e.g. risk management. They are also more financially literate, for example, they have aided in constructions of KPIs.

Another noted 'it is more a question of approach, rather than change in practices, i.e. how we deal with external partners/ other organisations (more businesslike)'. Another noted;

They do not tend to introduce specific practices as policies and procedures would tend to be more of an operational aspect of the University's work. However, governors do offer advice and also bring to bear an external perspective as to the way businesses in the private sector operate. This can result in the University adapting some of its practices. For example, we are currently updating our business continuity plan and have drawn quite extensively on the private sector experience of one of our governors.

Within Education, practices and improvements in functional areas were more common, with 63% of educational organisations reporting this type of board member input than at a more general level. These included; accounting and finance, performance measurement, audit, risk management, strategy, property, marketing and governance.

Within accounting and finance, both management and financial accounting expertise was noted, for example, one organisation highlighted the 'enhanced quality of monthly management accounts' and another noted that 'the advice of those with a financial background have facilitated best practice in financial management'. More specifically, in terms of management accounting, one organisation noted that a

business person had introduced 'balanced scorecard in terms of strategic planning but we have been moving away from that towards KPIs', another also noted their 'influence in the introduction of Key Performance Indicators'. With regard to audit, one organisation noted that the establishment of an audit committee was something which was influenced by someone with a business background.

Other practices were described as strategic, for example strategic planning, and the way that business people highlighted 'the importance of the financial impacts from strategic decisions and the practice of 'away' days to think clearly about the latter'. Risk management was another area in which business people contributed, for example, one noted that 'increased risk assessment is now in evidence'. Property development experience was also apparent, an example being with one College who noted that 'Members with specialist experience in law and property development were very helpful in giving advice when the College undertook a major campus redevelopment'. HR examples included 'the way the principals salary review is conducted', 'Director of one firm assists in appeals' and 'HR professionals have been able to suggest recruitment practices'. Marketing expertise was also noted, for example one organisation noted how the 'Marketing director of one local firm assisted in logo positioning'.

Other examples of increasing commercialisation were also given. For example, one organisation noted how 'Members of the Board continually highlight to us commercial and industrial examples of best practice which have been implemented'. Another noted the introduction of commercial practices, in this instance a number of incentives had been introduced to the College commercial company. However, the board secretary of this organisation noted that the public sector is slow moving, which makes it difficult for board members to push through innovative changes, an example given here was the Colleges new campus project, the secretary felt that had a board member tried to introduce something in relation to this then they probably wouldn't have been listened to. Governance improvements were also listed as being influenced by business people. For example, one organisation noted that a business

person was responsible for the adoption of the Carver Governance Model within the organisation.

#### Health, Social & Community Care – Introduction of Business Practices

Examples given by healthcare organisations followed a similar pattern to the other two sectors; with the dominance of examples of specific functional expertise in practices such as accounting, risk management, marketing and strategy, this was apparent in 69% of these organisations. There were also two organisations that specifically noted the potential for the private sector to learn from charities, for example, one noted how the 'evolution of new practices have been led by non-business members and the CEO' and another felt that 'business has more to learn from the voluntary sector than the other way round'.

There were a small number of organisations, 14%, who felt that business people did not introduce specific practices, but contributed more with their general expertise. For example, one organisation noted they had 'No specific examples. They act as a good sounding Board, as searching questions and provide suggested solutions where appropriate', similarly, another noted 'cannot think of any immediately but the discipline of regular attendance at meetings, properly tended minutes etc. has helped our charity establish itself'.

With regard to more functional expertise and practices, accountancy was mentioned more frequently, for example 'streamlining of financial reporting', 'improved financial reporting methods', introduction of KPIs and another also noted how an 'Accountant was first Treasurer, so helped with systems. No other special practices by 'business experienced' directors'. Another area documented was expertise in risk management, one organisation noted a practice introduced by a business member was the creation of a risk register. Marketing examples included one organisation who noted that a business person had highlighted the 'need for a marketing strategy in advance of a fundraising strategy'. Strategy was also mentioned, this example was

the introduction of the Rolling Annual Review of the Strategic Plan, 'rather than waiting until near end of the 3 year life plan and then reviewing'.

A number of other specific examples were given, for example an initiative of one business person was 'inviting members of staff to attend the board meetings to give a presentation on a specific key aspect of the service delivery, to ensure that the Board not only meet the staff at the centre of delivering the service, but to keep up to date with the organisations work and progress'. Other examples included 'performance related pay', 'improved tendering procedures for capital projects', 'better commercial appraisal of proposed service provision', and the 'introduction of annual appraisals for senior management'.

In contrast to the above, in response to the question of business people introducing practices, one noted 'no, because they aren't there to do this; they are there to monitor – and our control/ monitoring processes have improved as members have pointed at weaknesses in current practice'. This could be an example of legitimating behaviour.

#### Conclusion: Introduction of Business Practices

As has been previously highlighted, charities face increasing pressures to become more 'business-like' (OSCR, 2008). This section has considered the impact that business people have on charity boards, particularly in terms of the introduction of business practices into these organisations. The results were shown to highlight that this impact came in one of two forms; functional expertise, general expertise, or in some cases, both. There were some instances where specific practices were identified that a business person had introduced, such examples included: a traffic light reporting system, KPIs, a model of risk management adopted from the public sector, and the introduction of a balanced scorecard. The majority of responses pointed towards a more general contribution, rather than the introduction of specific practices and ideas.

Some interesting observations were made by one organisation from the education sector, who felt that the slow moving nature of the sector made it difficult for board members with a business background to push through innovative changes. In addition, two healthcare organisations specifically noted the lack of impact of these individuals, with one noting that new practices were led by non-business members and the CEO, and another organisation suggesting that business could learn more from the voluntary sector than the other way round.

The nature of these results are difficult to understand, although on the surface practices and expertise of business people have been identified as having an impact on the board, the lack of depth of the survey data makes it difficult to interpret in relation to the theoretical framework. A number of studies in the public sector have considered the adoption of private sector practices and have found these organisations use them for legitimacy (e.g. Arnaboldi & Lapsley, 2003). In order to ascertain whether these practices have an instrumental role in the organisation, or are merely symbolic will require in-depth investigation within case study settings.

# 7.2.2 Response to Practices Introduced by Business People

The above section outlined examples of practices which were introduced to charities by board members with business experience. However, as outlined in chapter 3, where the existing literature in this area was considered, differences were reported with regard to how these practices were received. Respondents were asked to rate how practices were received on the following scale:

- □ very well received (4)
- □ well received (3)
- □ badly received (2)
- very badly received (1)
- $\Box$  don't know (0)

A one-way between-groups ANOVA was conducted for the reception of these practices for the three groups. An overview of these results is presented in the table below:

**Table 7.4 Response to Practices Introduced by Business People** 

Charitable Purpose		
	Mean	Standard Deviation
Education	3.61	.803
Housing	3.61	.502
Health & Social Care	3.21	.419
Total Population	3.50	.658

A one-way between groups analysis of variance was conducted to explore the impact of sector on how well business practices were received in charities. Charities were divided into groups according to their charitable purpose. There was no statistically significant difference between the mean scores. Education and Housing reported the same mean score (3.61) which was between well and very well received. The mean score for health was slightly lower (3.21), suggesting that business practices were least well received in healthcare charities.

It is apparent from chapter 2 that the adoption of for-profit practices in charities yields a variety of views as to their appropriateness within the sector. The results from this survey show that the practices which business people introduced to charities were well received, this is supported by the work of Austin (1998) who reports a positive impact of the presence of business people on nonprofit boards in the US. However, this contradicts views from authors such as Bowen (2004) who suggests that professional staff in charities may be dismissive of business professionals due to the belief that they will not understand the unique qualities and professional norms of the organisation, or may be insensitive towards them.

Although there is a lack of studies which consider the impact of business professionals on charity boards, other authors have suggested specific issues with regard to the adoption of for-profit practices within the organisation, in particular, that they do not fit with the specific purposes of the charity (see Landsberg, 2004; Alexander & Weiner, 1998; Foster & Bradach, 2005; Dees, 1998).

Due to the possible tensions between business professionals and members of the organisation, and the suggested incompatibility of for-profit practices in nonprofit organisations, it may be surprising that the results of this study show that business practices were either well received or very well received in these organisations. One possible reason for this may be due to the findings from the previous question, which may suggest that these practices did not penetrate the organisation, therefore concerns about the impact of these perceived inappropriate practices would not be relevant. Such findings highlight the need for a more in-depth understanding of the impact of business members on charity boards, this will be enabled through the case studies presented in the following section.

Furthermore, taking into consideration some of the responses from the previous question, respondents suggested difficulties business people may have in the first place in terms of actually introducing such practices, for example comments from one organisation that noted how the slow moving nature of the sector made it difficult for board members with a business background to push through innovative changes. There may also be issues here with regard to self-reporting, whereby in order to appear more up-to-date, these charities will want to show their willingness to adopt practices from the for-profit sector. This points towards a desire for charities to present themselves as modern and therefore legitimate (Meyer and Rowan, 1977).

# 7.3 The Motivations of Business People and their Understanding of Charity Context

This section will focus on the main reasons why business people sit on charity Boards and the extent to which they understand the specific context of the organisation in which they serve.

# 7.3.1 The Motivation of Business People Sitting on Charity Boards

Another issue which was investigated was the main motivations of business people taking on these unpaid roles as charity trustees. The majority of the reasons given for business people joining the board were linked to the idea of 'giving something back', for example categories of responses around this included; giving something back to the community, to use their expertise to aid the organisation, and because they have a specific interest or connection with the organisation. A much smaller set of organisations reported board members sitting on their board to enhance their own personal development or in connection with their job.

Similar findings were reported by Austin (1998), where in a study of Harvard graduates who occupied nonprofit board positions, it was found that intrinsic motivation was the biggest factor in them joining compared with extrinsic motivation; here 60% noted they wanted to 'give something back' and 26% sat on the board for personal development reasons.

# 7.3.2 The Extent to which Business People Understand Charity Context

The issue of the understanding which business people have regarding charities was investigated; with charities asked to identify the level of understanding they felt their board members with a business background had of their organisation. Respondents

were asked to highlight the level of understanding of business people on their board on the following scale:

- □ very well (4)
- □ well (3)
- $\Box$  not well (2)
- $\Box$  not at all (1)
- $\Box$  don't know (0)

A one-way between groups ANOVA was conducted to explore the impact of sector on business people's understanding of the context of the charity. The results relating to this are outlined in the table below:

**Table 7.5 Level of Understanding of Charity Context** 

	Mean	Standard Deviation
Education	3.43	.931
Housing	3.63	.875
Health	3.36	.559
Total	3.46	.818

A one-way between groups analysis of variance was conducted to explore the impact of sector on how well business people understand the context of the charities on whose board they serve. Charities were divided into groups according to their charitable purpose. There was no statistically significant difference between the mean scores. The overall mean score was 3.46, suggesting that the level of understanding business people have of charity context is between well and very well.

Contrasting views have been reported within the literature on nonprofit governance as to whether these business people do actually understand the context of the organisation with concerns surrounding an over-emphasis on finance over mission (Siciliano, 1996; Landsberg, 2004; Miller, 2002).

#### 7.3.3 Ensuring Business People Understand the Context of the Charity

Charities were also asked whether they specifically did anything to ensure that business people understood the context of their charity. The overwhelming majority did not report any specific practices to aid business people in understanding the complex nature of the charitable sector. Nearly all organisations noted the same practices they used in order to inform boards members of their roles and responsibilities. These findings are surprising due to the vast difference between the for-profit sector and the charitable sector. The nonprofit literature highlights major differences between these two sectors, and some authors have previously highlighted difficulties with business people understanding the charity context (Bowen, 1994; Landsberg, 2004).

#### 7.4 The Impact of Business People on Public Confidence

The perceived impact of business people on the board in terms of improving public confidence was also investigated. Board secretaries were asked how much they agreed or disagreed with the following statement:

The inclusion of business people as board members in charities increases public confidence in the organisation in comparison with the inclusion of other individuals.

Respondents were asked to use the following scale:

- □ strongly agree (4)
- $\Box$  agree (3)
- □ disagree (2)
- □ strongly disagree (1)
- $\Box$  don't know (0)

A one-way between-groups ANOVA was conducted to compare the impact of charitable purpose on charities perception of whether business people increased the level of public confidence in the organisation. The results are outlined in the table below:

Table 7.6 Impact of Business People on Public Confidence

Charitable Purpose		
	Mean	Standard Deviation
Education	2.86	1.207
Housing	2.64	1.311
Health & Social	2.39	1.116
Care		
Total Population	2.66	1.215

A one-way between-groups analysis of variance was conducted to explore the impact of charitable purpose on charities perception of whether the presence of business people increased public confidence in the organisation. There was no statistically significant difference found between the mean scores. The mean score for the total population was 2.66, which was between 'agree' and 'disagree'. The highest proportion of charities 'agreed' or 'strongly agreed' with the statement, this was 48% and 22% respectively. In terms of those that disagreed, this was 13.5%, with a further 4% strongly disagreeing, the rest noted that they 'didn't know'. This apparent view that business people increase confidence in the organisation can lead us to believe that charities recruit business people in order to make them appear legitimate (Meyer and Rowan, 1977). As previously outlined, this idea of charities recruiting business people onto their boards as a legitimating function will be considered in more depth in the case studies.

Further evidence for this can be seen when considering the results in section 7.2 with these findings. It was reported earlier that business people do not necessarily bring their business expertise to bear within the charity, with relatively few examples of business practices being introduced into these organisations. This could be an

example of decoupling (Meyer and Rowan, 1977), where the organisations legitimate themselves by recruiting business people to present themselves to the outside world that they are operating in a more business-like way, and are therefore legitimate. However, this external façade is detached from the inner core of the organisation, where these business people have not made an impact, and the organisation is not necessarily being run in a business-like way.

In terms of the specific areas, Education had the highest mean, which was 2.86. This sector also had the highest percentage of organisations which either 'agreed' or 'strongly agreed', with a combined total of 79% of organisations expressing these views. This high regard for business people is also apparent from the findings of the composition of the board, whereby educational organisations were the only group where all organisations had business people on their board. These organisations also had the lowest percentage of organisation whose board was composed of only 1-25% business people, only 12% of organisations had less than a quarter of their board composed of business people, 88% had a quarter or more.

# 7.5 Conclusion: Business Expertise Survey Results

This chapter set out to specifically investigate the impact of business professionals on charity boards. The findings have highlighted the recruitment of these individuals across the entire sector, with 96% of charities found to have at least one business person on their board. This was attributed to coercive isomorphism (Di Maggio & Powell, 1983), in particular, pressures from the charity regulator to appear more business-like (OSCR, 2008). Such a finding would suggest the recruitment of these individuals as a form of legitimation (Meyer and Rowan, 1977). However, results for Housing Associations were lower than for other sectors, this was attributed to pressures from the Housing regulators to recruit tenant members onto the board, this was explain by Abzug and Galaskiewicz (2001) as legitimation from below.

Charities reported a general satisfaction with the proportion of business people on their boards, with those organisations that were not satisfied wishing for more business people. There were relatively few organisations that expressed the concerns outlined in the literature that these individuals may not fully understand the nature of the organisation.

The specific impact of business people was found to come in the form of both general and functional expertise. There were some examples of specific practices that business people had introduced, such as a balanced scorecard, KPIs, and a model of risk management. However the majority of responses pointed towards a more general contribution, rather than the introduction of specific practices. The nature of this data – responses from self-reporting questionnaires – was viewed as being difficult to understand. Although on the surface, practices and expertise of business people have been identified as having an impact, the lack of depth in the survey data makes it difficult to fully interpret these results. Numerous studies of the adoption of private sector practices by public sector organisations have found that they use such practices for legitimation (see Arnaboldi & Lapsley, 2003). This may be true of the charities in this study who may recruit business people and adopt these practices to maintain legitimacy. To fully understand this it will be necessary to study these ideas in a more in depth case study setting.

In the literature in this area (e.g. Landsberg, 2004) there is a view that the adoption of business practices undermines the charities fundamental ethos, however these results point towards a passive acceptance of the adoption of business practices. This acceptance may be due to the lack of impact that these 'business practices' have had on the organisation and the way it functions and that their role is merely symbolic. In terms of the motivations of business people serving on nonprofit boards, the main reasons were personal. This raises the question as to whether these business people are intent on bringing business practices and fundamental changes to the organisation, or whether they are there for reasons of personal fulfilment.

It was also found that charities view business people as having a role in increasing public confidence in the organisation. Such a finding would suggest that charities recruit business people recruit business people for legitimation – in order to appear modern and up-to-date (Meyer and Rowan, 1977).

In conclusion, the main findings from this section of the survey are in relation to isomorphism of charity boards regarding the recruitment of business people. There also appears to be a positive view of business people on charity boards, however the extent of legitimacy claims are unclear. In order to investigate whether their role is instrumental or symbolic, it is necessary to conduct further research in a case study setting. This case study research will be outlined in the following three chapters.

#### CASE STUDY RESEARCH

This section will present findings from a series of case studies conducted on Scottish charities. Cases were selected from the largest areas of charitable activity based on annual total income, these were: health, social and community care; housing provision; and education (OSCR, 2008). The specific organisations selected were a Housing Association, a Social Care charity, and a Further Education College.

The case studies presented an opportunity to undertake a more in-depth study of governance and the impact of business people on the board than the survey had allowed. It also allowed further investigation of specific issues identified from the survey, including the role of the board, the relationship between the board and management, particularly in relation to the role of the board in strategic and operational activities, and the composition of the board – particularly the extent to which individuals with specific expertise, such as business expertise use their skills on the board.

The choice of this type of method allowed for the mobilisation of aspects of the conceptual framework, as outlined in chapter 4, two specific concepts were drawn on: legitimation and isomorphism. The case studies allowed for further consideration of the specific isomorphic pressures faced by these organisations and the extent of isomorphism in the governance structures in these organisations within the charity field. Also considered, was whether the recruitment of business people onto the boards of charities was merely a legitimating exercise, as charities look to portray themselves as modern and up-to-date. The importance of the language of business in the social construction of charitable organisations was also considered.

This section is structured into three chapters, with each chapter reporting findings from a case study. Chapter 8 will consider a case study which focuses specifically on a social care organisation, Chapter 9 will consider a Housing Association, and Chapter 10 will focus on a Further Education College. Each chapter will be split into

two sections, reflecting the two main areas of investigation in this research: governance, and the impact of business expertise on the boards of charities.

# **CHAPTER 8**

# GOVERNANCE IN A SOCIAL CARE CHARITY: RESULTS FROM A CASE STUDY

#### 8.1 Introduction

This chapter presents the findings from a case study of a social care organisation. In order to maintain confidentiality, this case has been renamed and will be referred to as Robertson social care organisation. Interviews were conducted with key governance actors, and where possible documents were obtained and analysed. The table below sets out the four interviews which were conducted:

**Table 8.1: Interviews Conducted at Robertson** 

Interviewee Title/ Position	Background/ Day Job
Chief Executive	
Board Member 1	Private Sector Accountant
Board Member 2	Public Sector Accountant
Board Member 3	Management Consultant

This chapter will be structured as follows: section 8.2 will outline the background of the organisation, 8.3 will consider specifically the governance of the organisation, 8.4 will look the impact of those with business expertise on the board, and finally section 8.5 will conclude.

# 8.2 Background

Robertson was formed in response to an issue which was affecting its local community, regarding care arrangements for adults who required long term support. In particular, this focus was in relation to difficulties moving people from specific local institutions - two adult hostels, and one long stay hospital unit – back to living

and being supported in the community. Frustration with the situation at the time resulted in a group of local people making an application for a small grant to set up an independent local agency. They initially constituted themselves as a steering group, and put together the constitution, the memorandum and articles for a not for profit company limited by guarantee which then applied for charitable status – this became Robertson. The organisation was founded in 1989, with the current chief executive being appointed. The initial aim was to provide support services to adults with a learning disability in the local area. However, this was later broadened to providing support to anyone, of any age, requiring paid support. The organisation now operates over four local authority areas in Scotland.

The charity currently employs 300 staff and has a membership base of approximately 100. The services they provide range from a couple of hours of support per week to 24 hour support to 140 people, with ages ranging from pre-teens to individuals over 80 years of age. They also run a short stay (residential respite) service for adults with a learning disability living in the local area. They currently have a turnover of £6 million. The mission statement of the organisation is as follows:

(Robertson) believes that people should have the support they need, delivered in the way they want, to live their chosen lives. We work in partnership with people and their families, to help them to identify and plan for the kind of life they want to live, and then to support them to achieve this. We also work with local communities to strengthen their capacity to include people with disabilities and other support needs as valued citizens.

# 8.3 Governance

The following section outlines the governance of the organisation: firstly, the governance structure will be considered, followed by a discussion of the specific model of governance in the organisation – the board cycle, and finally the composition of the board will be detailed.

#### 8.3.1 Governance Structure

Governance arrangements of the organisation have undergone significant changes since its inception in 1989. The early arrangements were a reflection of the initial setup of the organisation. Robertson was established by a group of local people, and the initial management committee was set up in a way that allowed them considerable input into decision-making. The initial management committee had two major components; one was statutory, and one was non-statutory. The statutory component comprised of three people each from health, housing and social work, this was a total of nine individuals. The non-statutory component had eight individuals, these included people from housing associations, local voluntary organisations and family members. The organisation was membership based, with around 100 members, and the management committee was answerable to these members at the annual general meeting (AGM) and had to submit themselves for election or re-election at the AGM. This structure of the membership base, and the management committee of 17 with the statutory and non-statutory component was in place from 1989 to 1996. The Chief Executive and Committee deemed this an effective structure for the initial activities of the organisation – where there was a clear project, which was to move people on from the two hostels and the hospital.

However, as the organisation was moving through this project, difficulties were highlighted with this structure, including the requirement that any new proposals had to be taken to the committee for the statutory and non-statutory components to agree or disagree. Furthermore, because there were individuals there who represented other components, not all members primary commitment was to Robertson. In addition, it was noted by the chief executive that at this time, there was a shift in both the understanding of what governance was, and also the responsibility of the board, specifically towards an understanding that the primary commitment of a board member is to the organisation on whose board they serve.

The change in governance structure was also as a response to the external environment, with changes in the structure of local government changing from

regional to unitary authorities, there were also changes in the health service. These pressures led to a change in the governance structure of Robertson. The organisation began to look outside the governance of their own organisation, and began to think about how they could, as the chief executive described it, 'get some kind of sense and order into the activities of the board consciously and sensibly'. The changes that resulted were the removal of board representation from external bodies, as a result individuals became members of Robertson and were elected by the membership and the board number was reduced to 12.

# 8.3.2 The Board Cycle

Fundamental to the governance of this organisation is the adoption of a governance model – the Board Cycle. One of the board members who works as a management consultant was instrumental in the introduction of this model. The model was drawn from Bob Garratts (2003) book, *the Fish Rots from the Head* and was the result of work between the chief executive and the management consultant. At the time they were looking at how the board operated and what it needed to do, particularly in terms of making the most of time. The management consultant – who works with boards as part of his day job – suggested some ideas to the Chief Executive, and gave the Chief Executive some work written by Bob Garratt on boards. Although the main impetus for this move came from the management consultant, he noted that after these initial discussions, it then became a board decision, with the Chief Executive being described as the 'key mover' behind the adoption of this model.

The Chief Executive and the management consultant worked together on what became a shared vision to change the governance of the organisation. This shared approach is in contrast to some of the literaure in this area which suggests tensions exist between the board and management (Cornforth, 2003; Taylor et al, 1996). The relationship here is in line with what is advocated by Taylor et al (1996) and McClusky (2002) as a 'partnership' approach to governance between the board and

management. The existence of such a common shared vision is also found to have considerable impact on the effectiveness of the board (Cornforth, 2001).

The Chief Executive described this annual board cycle as a process whereby the main responsibilities of the board are addressed: direction setting; strategic thinking; monitoring and evaluation; and accountability at the end. The Chief Executive specifically highlighted that direction setting was about board, and not operational policy, that it was about looking and taking the longer view and saying which direction the organisation has to be going in.

The process of strategic thinking was undertaken once the strategic direction had been set, and involved consideration of what were the big milestones that the board needed to think about, and how to make sure, as an organisation, that the resources were in place to allow the organisation to move forward in that direction. When describing the strategic thinking process, the Chief Executive was quick to clarify the distinction between strategic thinking and strategic planning - the Chief Executive noted that Garratt labelled strategic planning as a contradiction in terms as planning is for managers, and strategy is for the board. Monitoring was described as the regular oversight of what managers are actually doing to achieve the goals the board has set for them. In terms of accountability, the Chief Executive emphasised the importance of being in a position to report back to the various stakeholders of the organisation on what they have done. A board self-evaluation was also introduced in order for the board to consider how they functioned, and what they must learn to improve as a board.

All members of the board who were interviewed were positive about the board cycle, for example, the private sector accountant noted that 'it's one of the best structured pieces I've seen'. The board cycle illustrates a commitment to clearly distinguish between strategy and operations, an area of considerable tension in charities (see Cornforth & Edwards, 1999; Steane & Christie, 2001; Leatherwood & O'Neal, 1996). An issue identified in the literature is lack of role clarity (Parker, 2003; Harris,

1989; Widmer, 1993), however the board cycle is structured to specifically reflect the key roles of the board, placing clear emphasis on what these key roles are.

The key roles of the board, as reflected in the board cycle, are clearly laid out in the board member handbook as follows:

- Establish what (Robertson) stands for and what it seeks to achieve.
- Determine how it plans to get there and that it has the resources it needs to achieve its goals.
- Delegate to the paid managers responsibility for implementing the plans.
- Monitor the use of resources in the interests of those (Robertson) exists to serve.
- Account to the members of the company, and to the stakeholders for those resources through the annual report, and at the annual general meeting.

In terms of the meetings structure of the board, they meet monthly, which is seen as an important element of the governance structure, as it enables members to keep upto-date more easily than if meetings were quarterly. In terms of keeping people up to date, the Chief Executive notes:

You still have to package reports like a story: the first paragraph is reminding people what it was all about and how we got to where we are and where we are now, the next bit is what happened since we last looked at it and the last bit is a kind of summary and what we need to do next. But you actually have to do that because you can't expect people just to pick up the thread with a month in between, they've got their own lives and their own commitments and their own interests'.

The structure of governance around the board cycle was also deemed to keep the board focussed on their role as each meeting is monthly, and is 'themed' and therefore agendas are designed to reflect this. The application of such a framework when designing board agendas has been shown to keep Boards more focused on their role (see Inglis et al, 1999; Inglis & Weaver, 2001).

With regard to the way that board members are kept up-to date with arrangements, they are issued with a number of key documents. In terms of policy documents, each policy document is lengthy, and therefore has a summary policy statement which states its key aspects – the board has a copy of all the policy statements and these are reviewed each year. The board also gets a workplan for each area that the organisation operates in, which details what has been accomplished to date, and what will be happening next – this is what is reported against at the quarterly monitoring meetings of the board, where the area managers come to the board and present their own area report. The budget for the year is also issued to board members in their pack. Alongside this, there is information about the structure of the organisation and its staff, and also the people they support – the level of support they receive, what it costs, where the money comes from.

The flexibility of the cycle was also noted. It was highlighted by the management consultant, that the cycle does not tend to work if a big issue comes up. He noted that there had been a recent instance where a major issue had arisen which needed a considerable amount of discussion, it was necessary to have that discussion in the board meeting – resulting in a two hour board meeting, meaning that the agenda that they were supposed to be dealing with could not be discussed. Therefore, at times, things must be moved over into the next again meeting. The structure of the cycle allows for this, as the management consultant explained:

I mean things like monitoring, for example, is fairly straightforward; there's a set of reports and you go through the reports and it's a fairly standard process. So in fact, you can do a board monitoring session probably in the hour and then that gives you another hour to discuss a topic.

It is apparent that the board cycle has become part of the organisations formal structure. Meyer and Rowan (1977) would suggest that such institutional rules are adopted ceremoniously and function as myths. However, far from being merely an act of symbolism, this cycle appears embedded into the organisation and is fundamental to the way in which organisation is governed. The academic literature contains numerous examples that suggest the adoption of tools used in the private

sector undermine the values of charitable organisations (Bush, 1992; Backman & Smith, 2000; Eikenberry & Kluver, 2004; Foster & Bradach, 2005; Harris, 2001; Landsberg, 2004). With regard to specific models of governance, Alexander & Weiner (1998) consider the adoption of corporate governance models in charities and conclude that they are not viable for the sector. In contrast, this does not appear to be the case at Robertson. This can be highlighted from the following statement from the management consultant, who brought the board cycle to the organisation:

so I think one of the things that makes it work as a board is the sort of common shared view, you know, everybody subscribes to a very key set of values that are put into operation and we do review the values every year and often in this session here, and that session there we'll look at the values. In fact we use the board development session this year to look at values, to revisit them, you know, why independent living, what's that about, so yeah we spent some time doing that. And I think that's what keeps it together as an organisation, there's not, I don't think I have ever, ever been involved with a board where there's never been any falling out, but not because they're all into group think, but just because everybody kind of has the same view. I mean there's still strong views expressed, and people will disagree but it's never got to sort of acrimonious debate and discussion, it's just usually very harmonious really, and I don't think I've ever come across a board that has such a common shared set of values. That'd be quite an interesting one to look at...about how values might drive the effectiveness of the board because I think it does in (Robertson).

Such a view is in contrast with reports from the literature which suggest that business people do not always fully understand the values underlying the organisation (Landsberg, 2004; Bowen, 2004; McFarlan, 1999).

#### 8.3.3 Board Composition

The Board currently has 15 members, and is serviced by the Chief Executive. In terms of it's composition, there are four service users on the Board, seven individuals that can be described as non-business members who come from predominantly the public and nonprofit sector. Finally, there are four members of the board who can be described as 'business people'.

The composition of the board is detailed below:

**Table 8.2: Board Composition of Robertson** 

Board Member	Background
Chair	MSP
Board Member 1	Private Sector Accountant
Board Member 2	Public Sector Accountant
Board Member 3	Management Consultant
Board Member 4	Service User
Board Member 5	Retired Social Worker/ Relative of Service User
Board Member 6	Retired Nurse Manager/ Relative of Service User
Board Member 7	Works in Community Services/ Community Development
Board Member 8	Founding Member/ Retired Social Worker/ Set up other Charities
Board Member 9	Runs Local Business
Board Member 10	Solicitor
Board Member 11	Service User
Board Member 12	Relative of Service User
Board Member 13	Service User
Board Member 14	Service User

The following statement from the organisations website sets out its view on the composition of the board:

Board membership needs to span a range of skills and experience, from using services (as a person with support needs or the relative of a person with support needs), to professional expertise about disability issues, and expertise related to running a medium sized company.

With regard to the composition of the board, the organisation has a grid which identifies the needs of the organisation, and what the current individuals on the board bring to it. The boxes of the grid are checked to ensure that between all the board members, they meet the requirements. Should any gaps be found to exist, then this highlights the areas that need strengthened, and points to who they should be

recruiting next. Therefore when the organisation recruits new board members, they specify the exact type of expertise they are looking for.

The essential elements of the grid – and essentially what the board must cover are: people who can understand the finances of the organisation, therefore someone who can understand the accounts and read a balance sheet. Someone with an understanding of law, particularly company law, charity law, and community care legislation. People who understand community development, people with PR skills, people who have used the services themselves, or who have strong connections to people who use the services, so they can see it from the inside out. Such a system appears to be instrumental in the governance of the organisation, this is in contrast to the view of Meyer and Rowan (1977) who suggest that such formal structures function as myths.

Another key element of the board is that it currently has four service users on it. The recruitment of service users on the board has had implications for the board, for example, previously, board papers were sent out one week in advance, however they are now sent out two weeks in advance. Board papers have also had to be made more accessible. In order to make board papers accessible to the service users, layouts are standardised, and symbols are used, so for example, with regard to the proposed transfer of properties to a housing associations, the icon of a house is placed next to this so that it is instantly recognisable. In terms of the monitoring reports, graphs and symbols are used to allow for a picture of what has been done against what still needs done, which the chief executive notes makes it easier to talk through the process. They also use photographs of the individuals whose report it is. This improvement in accessibility has also had an impact on the other board members, who noted that this actually made the finance papers much easier for them to understand. At board meetings, flip charts are also used and the board is broken down into smaller group discussions. During board meetings, all individuals have cards which they must hold up if they want to speak – this is so that everyone gets a fair chance to talk and so people don't cut each other off. Abzug & Galaskiewicz (2001) suggest boards are legitimated from below, the recruitment of service users

here could be seen to reflect this, however the presence of these individuals appears to be instrumental, with significant steps taken to involve these individuals and hear their views.

### 8.4 The Impact of Business People on the Board

The following section specifically considers the impact of people with business skills on the board. There are four members of the board who can be described as 'business people'; the first of these members works in local business, another of these members is managing director of a firm of management consultants, this member had previously worked as a clinical psychologist for the NHS. The remaining two members are both accountants, one is a private sector account, working for a large bank, the other is a public sector accountant, although has previously worked in the private sector. The latter three were interviewed for the study. The following section considers the motivations of these business people, their understanding of the organisation, the use of business expertise on the board, and pressure for the organisation to become more business-like.

#### 8.4.1 The Motivation of Business People

The three business people interviewed for this study were asked how they came involved with the organisation, and what their primary motivations were for joining the Board. The two accountants were relatively new to the board, the public sector accountant had been on the Board for approximately a year. This member joined the Board after responding to an advert in the local press: they lived in the local area, and after doing some background research on the organisation, felt that it would be a very interesting opportunity. The private sector accountant had been looking to become involved on a Board for some time and also saw an advertisement for Board members in the newspaper, the primary motivation of this member for joining the Board was 'to give something back to the community' and also for personal development. The third business person, the management consultant, had recently

moved to the local area and also saw the post advertised, he had previously worked in the NHS as a psychologist and had worked with Robertson in this role a number of years ago. The motivations here were also to 'give something back'. These reasons for involvement are similar to those uncovered by Austin (1998) who considers the motivations of Harvard MBA graduates involvement on nonprofit boards 'to give something back' was the dominant reason.

# 8.4.2 Business People's Understanding of the Organisation

Although these three individuals were labelled 'business people', two of them had previous experience with work which was related to the organisation. The Management Consultant had previously worked as a psychologist for the NHS, during this time he had worked with service users of Robertson, he also had previous Board experience as a member of the Executive Committee of a Housing Association during the late 1980s/ early 1990s, and also on the Board of the company he works for. The public sector accountant had previously worked in a high secure hospital setting and noted that they had a lot of experience around clinical governance and social care issues and felt able to bring this experience to the Board. The third Board member, the private sector accountant, had no previous experience of work related to the sector, and noted some difficulties with understanding some of the specific issues related to social care.

The difficulties of understanding the complexity of the sector was highlighted by the management consultant, who noted that although he had previous experience within such a setting, it was a number of years ago, and he had found it difficult to come back in:

I found it really hard because I'd been out of it for the best part of 12/13 years, and coming back in was quite hard. And I think it's quite hard in this specific sector because it's not just knowing something about the issues surrounding people with learning disabilities but also around the whole social care sector...and the provision for housing, supporting people, legislation and so on.

However, the private sector accountant shows an awareness of the differences between voluntary and the private sector:

As an accountant on the board, I try to be very mindful of what our purpose is in being there as an organisation, and it's not the same as running a commercial organisation.

# 8.4.3 The Use of Business Expertise on the Board

In terms of the external expertise which these three business people brought to the board, there appears to be a significant distinction in terms of the impact in which they have had on the organisation. The management consultant appears to have been instrumental in terms of the organisations adoption of its board cycle, which was based on work that he did with boards as part of his day job. In addition this model does appear to be embedded within the organisation and is used positively. This finding is in contrast to propositions made by Meyer and Rowan (1977), who state that organisations will adopt practices that will make them appear modern and up-to-date, however such practices do not contribute towards the technical efficiency of the organisation. In contrast, it appears the board cycle has been effective, and the management consultant now notes how he uses Robertson as an example when working with organisations from the private sector:

I do use it as an example because I do work with boards in the private sector and I have sort of drawn on the experience to say 'look, when you're struggling about how you...the process you use, you know I'm on a board, it sits, we meet for two hours a month, we've got a yearly cycle of activities, and it's quite an eye opener to some people, that you can actually get through and run a relatively biggish organisation, 5/6 million pound a year organisation on that basis, with people who are volunteers, and not getting paid for it.

Other examples of what this individual brought to the organisation were also apparent, for example, part of the work in which he was involved with outside the organisation was helping organisations to recruit people for senior positions,

something which he also been involved with in Robertson. The management consultant also noted that one of the areas he worked on in his consultancy company was around collaborative working, and he had been pushing this within Robertson about getting into partnership and collaborative working with local authorities where, he noted, there had historically been some tension. This is further evidence of the instrumental value of this business person on the board.

Another way that the specific expertise of these business people was used was through the Audit Committee – this was the only sub-committee - and through participation in ad hoc working groups set up to deal with specific issues. Both accountants were on the Audit sub-committee, with the private sector account Chairing the meetings.

In contrast to the management consultant, who was instrumental in the adoption of the board cycle, the two accountants on the board had less of an impact. The main contribution of these two individuals was to give clarification and reassurance on the financials and to oversee specific decisions. For example, they had made contributions on small working groups of 2 or 3 board members which were set up to deal with specific issues, for example reviewing rents. There was no evidence that these individuals brought in their external expertise other than in an oversight capacity. Another example of this was noted by the public sector accountant when describing what he brought to the board:

Just experience as well, when you see things coming across the board table that you've probably dealt with in the past and that helps a lot because you can then say whether it's reasonable or not. For example, setting rent levels; so we'll be looking at previous years comparisons, and what other organisations are doing. The management team are very good at doing that, they do all the predatory work so when it comes to the board table you can look at it and say 'well does that seem reasonable' that's really helpful. Again, the management team, they're very good at providing information for you.

The presence of these two accountants on the Board may seem to imply that the accounting function of the organisation will be rigorous, however, after discussion with the management consultant, we can see that this is not the case:

we've got a couple of people who are accountants for example Public Sector Accountant and Private Sector Accountant, and it's quite interesting because Private Sector Accountant particularly I think finds it hard that she's in a different accounting environment, and she sometimes looks for some of the rigour that you might find in X bank etc. but it doesn't happen at Robertson.

The presence of these two accountants on the Board, coupled with their apparent lack of any instrumental contributions and the 'lack of rigor' in terms of accounting can be seen as a possible example of legitimation (Meyer and Rowan, 1977) whereby the organisation presents itself as modern and up to date.

There is a desire for more individuals with business-related expertise, and an acknowledgement that their knowledge would have an impact on the organisation, the management consultant noted:

We miss somebody who is in marketing...We miss probably also somebody who's in sort of related area of PR, you know if we had somebody who was really working with a PR agency lets say, I think we could do all sorts of things with (Robertson), particularly in terms of the (Robertson) Trust and fundraising.

#### 8.4.4 Pressure to Become more business-like

The pressure for the organisation to operate in a more business-like way was apparent. The private sector accountant discussed the issues faced with regard competitive tendering, where private companies are coming in who want to provide social care services. One of the issues with this is that they will be looking to make a profit in a nonprofit sector. The accountant voiced concerns over the quality of service that would be provided by these organisations, and noted how Robertson was committed to the needs of the service users. She notes the impact of this:

As an accountant on the board, I try to be very mindful of what our purpose is in being there as an organisation, and it's not the same as running a commercial organisation. You know, (Robertson) and another organisation just didn't know if they were going to receive funding so how can you even start doing a budget if you don't know if you're going to have any funding, so you can't really get too hung up on 'but we must know what our funding is' because you don't know because the local authorities won't commit to it. Or they commit to it for a year or two years, so until we get into a cycle of three-year funding you have all that management time tied up, continually working on the deals – whereas if it was three years, you've got that platform to work on.

The above example represents a specific isomorphic pressure. When such uncertainty is apparent, Di Maggio and Powell (1983) suggest that it will encourage mimicking of other organisations which they believe to be more successful, perhaps the private sector organisations that they are in direct competition with. Such financial uncertainty may explain the symbolic role of accountants on the board.

#### 8.5 Conclusion: Robertson Social Care

At the centre of this case study is the adoption of a private sector governance model, with a board member from the private sector being instrumental in its adoption. The board cycle was brought into the organisation by a board member with business expertise, in this case, a management consultant. The Board Cycle is at the heart of the governance structure is this organisation. Meyer and Rowan (1977) suggest that the adoption of such practices and recruitment of such people are adopted ceremoniously, having a merely symbolic role, in order to present the organisation as legitimate to its external environment. However, it is apparent that this model has become embedded within the organisation. It has a clear impact on governance, where on an annual basis, the main roles of the board are addressed: direction setting, strategic thinking, monitoring and evaluation, accountability. This translates into the monthly board meetings, which are themed around these areas and therefore focusing the board and their role, and keeping them from straying into more operational matters.

Fundamental to the successful implementation of the model also appears to be the shared vision of the Chief Executive and the management consultant who worked together to implement the board cycle within the organisation, and also the rest of the board who were receptive to the change. It has been reported that tensions can exist between the board and management (Chait & Taylor, 1989; Taylor et al 1996; Cornforth, 2003), however in this situation a partnership approach appears to have been adopted between the Chief Executive and a Board Member in order implement of new model of governance.

Such an example of the adoption of a model from the private sector, is in contrast to a growing literature which highlights concerns over charitable organisations importing models used in the private sector (Bush, 1992; Backman & Smith, 2000; Eikenberry & Kluver, 2004; Foster & Bradach, 2005; Harris, 2001; Landsberg, 2004). Such research suggests that business practices can undermine the values of the charity, however in this case it is made clear by the management consultant, that governance is driven by the values of the organisation.

In contrast, the other two board members with business expertise – a public and private sector accountant – did not appear to be so instrumental in contributing to the effective governance of the organisation. Their role appeared to consist of a more general oversight role. In particular, the private sector accountant was noted as having difficulty in adapting to the different context of the charity sector. The lack of any instrumental contribution can be viewed as a possible example of legitimation (Meyer and Rowan, 1977), where the recruitment of these individuals reflects a desire of the organisation to present itself as modern and up-to-date.

#### **CHAPTER 9**

# GOVERNANCE IN A HOUSING ASSOCIATION: RESULTS FROM A CASE STUDY

#### 9.1 Introduction

Housing Associations are charitable organisations which provide low cost social housing for those in housing need, including those on a low income, older people, and those with disabilities. Social housing has become an increasingly important issue, in 2008-09 57,304 households made homeless applications to their local councils in Scotland, this was a 21% increase since 1998-99 (Shelter, 2010). Both the decline in council housing as a result of the 1979 right to buy legislation, the transfer of Local Authority housing stock to Housing Associations, and the decline of private housing ownership (National Statistics, 2009) has resulted in an increasing reliance on Housing Associations as a provider of affordable housing, with one in nine properties in Scotland now Housing Association stock (Neary & Gibb, 2010).

These organisations face significant challenges, particularly in relation to the current economic, housing market, and credit crisis which is likely to result in longer-term public funding reductions (Neary & Gibb, 2010). Additionally, new legislation in the form of the Housing (Scotland) Bill will bring about changes to the regulatory system and the right to buy legislation. Within this context, Governance has evolved as a key concern of the sector. A 2009 report by the Scottish Housing Regulator highlights the importance of good governance, but reports significant variance in the quality of governance in the sector. The report suggests that many governing bodies have only a basic understanding of their role, and can become involved in low-level operational matters at the expense of strategy (Scottish Housing Regulator, 2009). This is illustrated below:

The quality of governance in the sector is variable. Some governing bodies are excellent, with a clear understanding of what their role is and how this differs and complements that of their senior officers. However,

other governing bodies have only a basic grasp of their role or become embroiled in low level operational issues, leaving their organisations to be led by their senior officers (Scottish Housing Regulator, 2009, p. 42)

This chapter presents the findings from a case study of a Scottish Housing Association, to protect its identity, it will be referred to as McPherson. Interviews were conducted with key actors, and where possible documents were obtained and analysed.

The table below sets out the four interviews which were conducted:

**Table 9.1 Interviews Conducted at McPherson** 

Interviewee Title	Position
Chief Executive	
Chairman	Retired Accountant
Board Member 1	Assistant Director of
	Finance in large public
	sector organisation
Board Member 2	Physical Disability
	Planning Officer with
	Local Authority
Board Member 3	Retired public sector
	manager

The chapter is centred around the two main themes of this research: governance issues, and the impact of business expertise on the board. It is structured as follows: (9.2) will provide some background to the case, (9.3) will consider governance, (9.4) will consider the impact of business expertise on the board, and (9.5) will conclude.

# 9.2 Background

McPherson Housing Association describes itself as 'a charitable organisation supporting local communities by providing and developing high quality affordable homes and housing related services'. It was formed in 1995 by a group consisting of both tenants and staff from the local area office of Scottish Homes, a previous housing regulator. McPherson have 35 employees and are governed by a management committee of 12 members made up of individuals from a variety of

backgrounds and experience. At present McPherson Housing Association own more than 17, 000 properties across east and central Scotland. The core members of this organisation have been involved prior to its inception in 1995, and were members of the initial steering group that formed the organisation in 1992. The current Chief Executive was on this initial steering group, becoming the Director of Finance when the organisation was established, and then progressing to his present post. The current Chair, along with another two board members were also involved on the initial steering group.

#### 9.3 Governance

The following section considers the governance of the organisation, particularly: governance structure, the role of the board, the distinction between strategy and operations, the relationship between the board and management, and the composition of the board.

# 9.3.1 Governance Structure

At the time of the study, a review of Governance had just been undertaken and changes were being implemented as a result. A significant focus of the review was centred around appraisal type meetings with the board members which were conducted by the chair and the chief executive. These discussions focused on the views of board members as to the practices and procedures that they perceived to be working well and those which could be improved upon, their own individual performance and the performance of the committee as a whole, and any training that board members felt would be beneficial. Structural issues surrounding the suitability of various sub-committees were also considered, along with meeting frequency. The structure prior to the review was a meeting of the full board six times per year, with sub-committees dealing with specific issues and bringing reports back to the full

management committee. Sub-committees encompassed: housing and development, finance and corporate services, and audit.

As a result of the Governance review, a number of issues were uncovered surrounding the sub-committee structure. Concerns were raised by individuals who did not sit on certain committees and felt that they were missing out on making important decisions. Such an example included individuals who did not sit on the housing and development sub-committee who felt that they were missing out on important development decisions and large maintenance contracts. To address this situation, the committee took the decision to feed everything through the full committee so it gave everybody the same information at the same time and they weren't just relying on minutes or reports from the sub-committees to catch up with what the sub-committees had decided. An issue with this was surrounding duplication of effort, things that had been agreed at subcommittees were then discussed again at the full meeting. There was a feeling from members that if they were not on a particular sub-committee, although they got the minutes of it, it was really just a résumé of the conclusions to the discussion of the decision taken. People not on that committee were not part of the discussion, they didn't know all the things that were dealt with there and the was a feeling of lack of ownership – they felt this would be better addressed by the management committee meeting monthly and doing away with all the sub-committees apart from the audit sub-committee which meets three times a year. The audit committee has the remit for looking at the risk map, also for looking at reports from internal and external auditors. In place of formal subcommittees, small working groups were created for specific issues, one such example was a recent meeting to discuss salary structure.

Major changes implemented as a result of this review included a change in the committee structure of the organisation. The meetings of the full board were changed from every second month, to every month, and both the housing and development committee, and the finance and corporate services committee were scrapped with their remits now being put through the full board. The only remaining sub-committee was the audit committee.

Another aspect of this restructure was the plan to have a clearer focus in the meeting of the full board. The chief executive noted his aim to introduce themed meetings, for example, looking at performance every quarter and finance every quarter and arrears every quarter and looking at certain subjects once a year. In contrast board member 1 and 2 were not aware of the plans to introduce such a structure to board meetings. There was also a desire to keep agendas light to allow more strategic subjects to come onto the agendas and to move the committee away from a lot of the day to day operational matters that the board has previously been more involved in.

The move towards monthly themed board meetings is reminiscent of the board cycle adopted by Robertson, however, in this instance rather than demonstrating that such a structure has been adopted and has improved the strategic focus of the board, the following comment from the chief executive points towards the adoption of such a structure as a legitimating device:

But it's always this dilemma – every association goes through this cycle of: they look at their governance, and either create sub-committees or they do away with sub-committees, have a full committee and then form short-line working groups to do various things and I think that's where we'll head more in the future.

The appears to be no rational justification for the adoption of this structure.

#### 9.3.2 The Role of the Board

Lack of clarity in terms of the role of the board presented itself as a significant issue within the nonprofit literature (Harris, 1989; McAdam & Gies, 1985; Parker, 2003; Widmer, 1993). At McPherson, the chief executive noted that new members are offered new committee member training through the SFHA, which is a structured package talking about their various roles and the expectations of them in Governance. Board members who work full-time however, highlighted difficulties with attending events such as these. Internal training in a less formal way was also offered, such as inviting board members to go through some policy training with

other members of staff. Despite these efforts, the issue of a lack of role clarity is also apparent within McPherson. For example, concerns were raised by board member 1 over lack of role definition and the level to which other board members actually understand the extent of their role:

sometimes it does feel a bit cosy, sometimes I don't think they realise, you know, at the end of the day, if something went wrong where does the buck stop; and it stops with us.

This comment was particularly in relation to a specific incident at a meeting relating to banding and structure, where the organisation was restructured around an individual who wanted to move from full-time to part-time:

what you get quite clearly coming over is 'oh aye, he's a nice guy, aye he should be paid...'. And what (board member 11) had said, you know, was there had been a previous meeting which I hadn't attended which was around structure, and there was a proposal because of somebody has not been well and had requested to go part-time and the rest of it. So basically they rebuilt the structure around that person going part-time. Now sitting there objectively, it should be about posts, not people, and sitting there objectively, you think 'a housing association the size of McPherson, with everything that's going on needs a full-time one of these.

Further issues over role clarity were highlighted by board member 1, it was apparent that although board members were issued with an induction pack which outlined their role, the onus was very much on the individual board member to read and interpret this role:

we get a big folder, so as part of your induction – and I haven't read the big folder – but you get a big folder which takes you through lots of different things and what your role is...I think from memory if you go through the big folder it'll give you kind of a job description in terms of what they expect from you as a committee member, whether everyone round that table's like me and has read that folder or not, I don't know...

The above examples highlight that although the role of the board appears to be formally documented and formal training is offered, there is a sense that this has a symbolic role (Meyer and Rowan, 1977) with board members not always attending

the training or actually reading or adhering to the formal roles which are documented.

# 9.3.3 The Distinction Between Strategy and Operations

As outlined in the research context, the distinction between strategic and operational activities presents itself as a fundamental challenge in organisations (Baysinger & Hoskisson, 1990), particularly in nonprofits where boards engage in operational activities (Leatherwoord & O'Neal, 1996; Steane & Christie, 2001; Cornforth & Edwards, 1999), this can cause significant tensions between the board and management (Cornforth, 2003).

At McPherson, the Chief Executive highlights that between 70 - 80% of what is discussed as board meetings is operational. He highlights the difficulty of staying strategically focussed:

we are trying, at least in theory, to create more space on the agenda and move committee a wee bit away from pure operational matters and trying to think more about strategy, because... it's very easy for committee to get in overly involved in operational and management matters and trying to get that dividing line is always a bit of a balancing act.

There are also conflicting views as to whether there is actually a clear distinction between strategic and operational activities. For example, With regard to these activities, board member 1 highlights 'I don't think there is a clear distinction'. This board member continues to explain that the board is too operationally focussed:

The standard financial instructions, all that kind of stuff comes to us for final ratification and I don't think, you know I wouldn't...I wouldn't take that to a board decision in here of non-execs, do you know what I mean, necessarily. And as I say, stuff like management structure, I would have expected just (chief executive) to deal with that. Things like salaries, yes, that would have to come, because he can't make a decision on his own salary, so yes...And everybody knows in finite detail where the double-glazings going in, and how it's getting on.

Additionally, board member 2 also notes:

I think there is, there are certainly things that come to management committee and I think 'why's that coming to management committee? Why does staffing structure have to go to the management committee, because to me that's something the Chief Executive would make a decision on.

In contrast, however, the chair of the board signals that there is a clear distinction between strategy and operations and the division of these tasks between the board and management:

I don't feel that the management committee if you like, interferes in the running of the organisation. There is a clear delineation, we're responsible for the strategic decisions: which way are we heading? How will we do it, or do we want to do it? Yes. How will we do it? We decide to a certain degree, but then it's up to (chief executive) and his management team to implement that for the management committee. So I don't personally feel that there's any difficulty in that.

In Housing Associations, the high proportion of tenant members on Boards is seen as a contributing factor in the Boards involvement in activities that are operational, due to the difficulties they have in adapting to the strategic nature of the role. It has been reported that tenant members view their role as being that of a representative, tending to focus on operational housing issues, at the expense of strategic issues impacting on the entire organisation (Audit Commission, 2004). This is a potentially significant issue in Scottish Housing Associations, where around 45% of volunteer Board members in the sector are tenants (Scottish Housing Regulator, 2009). This situation may go some way to explaining the operational focus of the board reported by the chief executive and board member 1 and 2, as this organisation has a high proportion of tenant members – with six of these individuals serving on the board.

## 9.3.4 Board and Management Relations

As outlined in chapter 2, the involvement of boards in operational activities is seen to be a fundamental cause of tensions between the board and management (Chait & Taylor, 1989; Taylor et al, 1996; Cornforth, 2003), it may be expected that tensions

would exist in McPherson, given the apparently heavy involvement of the board in operational activities. However, in contrast a shared approach to managing and governing the organisation appears to be adopted. For example, as a result of the restructuring, all the senior management team go to every board meeting, and the chief executive has meetings with the chair once or twice per week. In considering this revised structure, the chief executive notes:

it just gives you a broader ownership really, for both the committee and for the senior staff. We can actually cover for each other, we're part of discussions much more collaboratively. But from my own point of view, I also, you know, (chair) was in today, (chair) is in regularly, I see him once or twice a week.

The chief executive gives the following example of this shared approach suggested above, when describing a project he was working on with the chair:

I try to keep him up to date of anything that I think is going to be significant, we are actually jointly working on some PR stuff at the moment because – which is not something we've done very much of, and again that's about how we move forward collaboratively. Some of that's strategically, some of that's really about operational. Strategically about how we are placed in the hierarchy of housing associations, operationally, to make sure that our profile's high enough that opportunities – development opportunities, or others that might come our way don't bypass us because people don't know about us.

In the following quote the chief executive does outline a desire to make his mark on the organisation, however it appears as if needs to seek approval from the board:

if I think it is a strategic decision...and I've been staff restructuring as well, and I involved (chair) in that discussion, so you've got that, we then took it to a — we've got a convenors group which is a standing subcommittee of some of the more senior members of the committee if you like — it tends to be the Treasurer, Secretary, Chair, Vice Chair, and the Chair of our subsidiary company as well...as well as speaking to (chair), I talk through my plans with them, so they obviously give their feelings on various things. And so before it's gone to committee for actual decision, it's actually gone through a couple of opportunities for them to influence and shape really, my plans, because they were very much my plans, well, the management team, but only up to a certain point. Because this is really my chance to stamp where I thought we needed to go in the future in the organisation...But, again, we're a small enough organisation, that if something doesn't work, in a year or twos time we can say 'och, this isn't really working, we'll change it again' so we're

not actually putting things in place that are necessarily going to have to stay for the next year.

A further example of the chief executive seeking approval is outlined below:

But having the open dialogue with (chair) is a big help because then I can test the water really. If I've got an idea I can run it past him and if his face screws up in a big grimace then I think 'ok, well maybe I won't. But it allows him to know where I'm heading with these things, and shows that I'm actually doing some work as well, it's not just a case of hiding away in my wee room and churning away on the keyboards, it does allow you to kind of keep things bubbling along. And I much prefer that to having an absent Chair if you like, that you would only see at meetings.

This shared approach is also noted by board member 1:

I remember the last, a couple of meetings ago, there's some kind of costing database thing that they use to forecast over the next hundred years kind of idea, and that's what (chief executive) did, (chief executive) gave a high level overview of what that costing...well I'm not interested in what that costing thing does, I just need to know that you're discharging your duties properly and I can ...you know, my role as I see it is about, is a governance role in so far as satisfying myself and therefore everyone else that's involved that the staff are doing what they're paid to do. I don't actually need to see what they do on a day to day basis so, yes, absolutely, I think it's just that shared collective.

#### 9.3.5 Board Composition

The board currently has 12 members, including the chief executive, with the maximum number at 15. The composition of the board is outlined below:

**Table 9.2 Board Composition of McPherson** 

<b>Board Member</b>	Background
Chair	Retired Accountant
Chief Executive	
Board Member 1	Assistant Director of Finance in large public sector organisation
Board Member 2	Physical Disability Planning Officer with Local Authority
Board Member 3	Retired public sector manager
Board Member 4	Tenant Member
Board Member 5	Local Councillor
Board Member 6	Tenant Member
Board Member 7	Tenant Member
Board Member 8	Tenant Member
Board Member 9	Tenant Member
Board Member 10	Tenant Member
Board Member 11	Head of Housing Services at another HA

As a result of the absence of any limit on the length of time board members are allowed to serve, many of the current board members have been on the board for a significant period of time. It was noted by the chief executive that throughout the first ten to twelve years, the committee was made up of the same people. Three of the current members, including the chair were on the original steering group which was set up to form the organisation in 1992, another three members have been on the board for periods of over 13 years. Such a situation is in contrast to the Association Rules which state:

McPherson Housing Association recognises that in order to demonstrate openness and accountability it must aim for a reasonable turnover of the membership of the Management Committee over time.

Despite the above policy, a number of individuals have served for lengthy period, including some who were on the original steering group which set up the organisation. It was apparent that this was an issue of concern to the Chief Executive:

it's something that the committee themselves have talked about, that in the interest of good Governance having open-ended, never ending kind of membership isn't necessarily the best thing, because it does mean that we get into a comfort zone, that people just come along out of habit, that there's not this kind of automatic freshening up of the committee. The current chair of the board is a retired accountant, who used to run his own accountancy firm close to the housing association offices. The vice-chair has been with the organisation since 2003 and is a retired public sector manager, also serving on other public and nonprofit boards. In addition to the chair and the vice-chair, there are four additional non-tenant members. There is an occupational therapist who works as a physical disability planning officer with the local council, there is a member who works as head of housing services for another housing association, another member who is an associate director of finance with a large public sector organisation, and finally another member who is a local councillor. The remainder of the board is composed of six tenant members. It was noted specifically by both the chair and the chief executive that the organisations has experienced difficulties in recruiting members to the board.

One of the biggest issues that arose was the pressure from the regulators to recruit tenant members onto the board, as outlined below:

The regulatory, development funding and policy frameworks in Scotland have encouraged tenant membership of governing bodies. Around 45% of governing body members are currently tenants (Scottish Housing Regulator, 2009, p. 44)

The commitment to tenant involvement is laid out in the Associations *Recruitment of Management Committee Members Policy*:

As a matter of policy, McPherson HA is committed to ensuring that McPherson HA tenants have the opportunity to be actively involved in the running of the organisation. In pursuit of this principle, McPherson HA will aim for at least a third of all Management Committee members at any time to be tenants. McPherson HA will provide information, support and training to tenants who are interested in becoming involved in the Management Committee.

Despite this policy of 'active involvement', the practical impact of this was noted by the chief executive who identifies the mismatch in skills between the tenant members and the running of the organisation, he particularly notes issues such as understanding complex loan finance. The chief executive considers the recruitment of tenant members below:

seen as good practice and I'm never quite sure why that's the case because they are actually responsible for a £5 million business and while tenants may well have some input to it, you really are looking at business skills first and foremost, and I think there's also always a bit of a mismatch; I think sometimes tenants are kind of conned into coming onto Boards like ours in that there's a push for us to have representation, and they can be very useful but within a fairly limited way because they aren't used to running businesses generally.

This conformity to expectations from the regulator has impacted on the composition of the board, providing legitimacy via the recruitment of tenant members to the board. This situation is viewed as a significant issue in the governance of the organisation, for example Board Member 1 (finance director) gave the following response when asked how governance of the association could be improved:

The fact that there's this kind of encouragement to have tenants, and lots of them sitting round there rather than people who are actually there for their skills and what they can actually bring to the board.

The level of participation from tenants members was of particular concern, Board Member 1 noted:

there's somebody younger than me that sits round that table -I think he sits there because his dad sits there if I'm honest, until recently, I'd never heard the guy speak, and there has been a deliberate move by (Chair & Chief Executive).

The Chairman also consider the participation level of tenant member, and comments specifically on one member:

I suppose on most committees you get most of the participation from a small number of the members and sometimes not so much from others and we have two family members on our committee the younger one until fairly recently had contributed absolutely zilch in the way of questions, comments...As a result of the interviews I referred to earlier, and thinking about this particular member of committee, we arranged that my predecessor as chair, who's still on the committee would act as a kind of training officer and buddy and whatever else you like to call it – and it has been successful because he'll meet with anybody, but particularly this younger member before a meeting, and identify questions, points that might be made, questions that might be asked, we're now getting something out of him...

The commitment of the organisation in the recruitment of tenant members represents a specific coercive isomorphic pressure from the regulators. However, it is clear that these members to not make a significant contribution to the governance of the organisation. Such a situation where boards legitimate themselves through community representation is found by Abzug and Galaskiewicz (2001), they find a situation where select organisations are found to legitimate themselves from below through community representation, such a situation is apparent here.

A further example of such legitimation is the presence of a formal skills audit of the committee. The chief executive notes the following:

we do have a formal skills audit of the committee -I say we have it, it's built into standing orders, last year...earlier this year was the first year we'd actually formally done it'

This is a clear example of legitimation, where formal structures are adopted ceremoniously, and function as myths (Meyer and Rowan, 1977).

#### 9.4 The Impact of Business People on the Board

The following section will consider specifically the impact of business on the board of McPherson, by first considering pressures that the organisation faces to become more business-like, and secondly the use of business expertise on the board.

## 9.4.1 Pressure to Become more Business-like

The use of business language is identified in this case, such language as suggested by Meyer and Rowan (1977) to be symbolic, and adopted in order to maintain legitimacy. As has been identified, the charity sector is under increasing pressure to become more business-like, and the adoption of such language could be viewed as a response to this. Such language is used in the annual report, where the organisation is described as a 'business' A further example of the use of such language is used by

the chief executive below, in discussing the pressure to recruit tenant members to the board:

seen as good practice and I'm never quite sure why that's the case because they are actually responsible for a £5 million business and while tenants may well have some input to it, you really are looking at business skills first and foremost, and I think there's also always a bit of a mismatch; I thin sometimes tenants are kind of conned into coming onto Boards like ours in that there's a push for us to have representation, and they can be very useful but within a fairly limited way because they aren't used to running businesses generally.

Such organisational language is discussed by Meyer and Rowan (1977). Here, the isomorphism of business language with institutional rules provides rationality, prudence and legitimacy.

#### 9.4.2 The Use of Business Expertise on the Board

There was a sense that business members on the board had a broad role in line with other board members, specific use of their functional expertise was not apparent. For example when asked about their role of the board, board member 1 (finance manager) noted:

I suppose to give assurance as a board, to give assurance that these guys are doing the job that they're being paid to do, and that they're not just, willy nilly, going away, you know. That they're actually making a difference. And to challenge some of the stuff that they do bring to us.

This broad role of business people was also outlined by the chief executive, who noted that business people have more of an:

overseeing and monitoring role, I suppose refinancing exercises are a case in part, we would be involved in looking at the options, discussing with the consultants and then preparing a report jointly with the consultants to the committee and then make sure the committee get enough time to look at it...'

One difference which board member 1 feels they have made since coming on to the board is to challenge decisions and ask questions:

it's interesting, because the feedback I got when I first started from the actual people who work there, the people who are on the payroll, they'd say 'thank goodness you're here because you're asking questions', because it was really demoralising for them spending time putting papers and stuff together and nobody asked a question about it. But sometimes it gets to the stage where I almost kind of apologise now because I think it's me that's asking questions all the time and nobody else.

The importance of challenging management is highlighted by the Scottish Housing Regulator:

In a number of organisations governing bodies are not providing sufficient challenge to management. Often staff are seen as the experts and it is difficult for volunteer governing body members to get beyond this. In our experience, the key ingredient in this regard is independent thinking – to ask questions, to get the right information and advice, and to step back and make their own decisions in light of this. But governing bodies are not always good at insisting on the provision of proper information to fulfil their responsibilities. And they need to know when they have been given an adequate answer (Scottish Housing Regulator, 2009, p. 44)

Within McPherson, providing challenge and asking questions is not always welcomed, as board member 1 notes:

but sometimes for those older people, or longer serving people sitting round that table, you almost think you're an irritation

A significant distinction is made between board members who have recently been recruited, and those who have served for a considerable length of time:

And you kind of look at who's asking the pertinent questions around the table, I mean since me, BM 11 came on at the same time, presumably I came on because I had a finance background, BM 11 came on because she had a housing background, I can't remember the other guy that started, but he's a councillor or ex-councillor that's come on; and you start to hear different kinds of questions being asked compared to the other people; and the other that have been there forever are still quite happy to just sit there and have half a dozen biscuits and have a cup of coffee, do you know what I mean. But there will need to be more of that, I don't think that it can continue as it is. Because apart from anything else,

you run the risk, the fact BM 11s only been there a year, and that actually made her consider her position.

#### 9.5 Conclusion: McPherson Housing Association

This case confirms the existence of a number of issues which are apparent in the literature on nonprofit governance. There appears to be a lack of clarity in the role of the board, with a perceived lack of understanding of the full extent of the role. In addition, the board appears significantly operationally focused. This confirms findings from the Scottish Housing Regulator (2009) that boards become caught up in low-level operational issues. This could be due in part to the high level of tenant members on the board, this is seen as a contributing factor in the Boards involvement in activities that are operational, due to the difficulties they have in adapting to the strategic nature of the role. It has been reported that tenant members view their role as being that of a representative, tending to focus on operational housing issues, at the expense of strategic issues impacting on the entire organisation (Audit Commission, 2004). This is a potentially significant issue in Scottish Housing Associations, where around 45% of volunteer Board members in the sector are tenants (Scottish Housing Regulator, 2009).

This apparent tendency for the board to stray into operational matters is in apparent contrast to the literature which suggests tensions exist when boards stray into operational matters. In McPherson, the chief executive appears to actually take operational matters to the board, possibly to seek approval. This points towards a shared approach to both the governance and management of the organisation. There is also a lack of a clear governance structure, although a recent review has brought about structural changes in terms of meeting numbers and committee structure these appeared to be for reasons of legitimacy rather than reflecting increased effectiveness.

In terms of the presence of business people on the board, they do not appear to have any instrumental value. Although there appears to be a desire from board member 1 to have more of an impact, the current structure does not seem to allow for this, meaning this role is limited and symbolic (Meyer and Rowan, 1977).

#### **CHAPTER 10**

# GOVERNANCE IN A FURTHER EDUCATION COLLEGE: RESULTS FROM A CASE STUDY

#### 10.1 Introduction

This chapter presents findings from a case study undertaken within a large Further Education College in Scotland. In this case, governance arrangements had become of particular concern in the organisation: it was felt that governance had been overshadowed in recent times due to the development and move to a new campus, this had tended to dominate board and committee meetings; there had also been certain issues which had brought about industrial action. The data presented in this chapter, in line with the rest of this study will focus on two specific aspects of Governance: specific Governance issues which the College faces in achieving its primary purpose of the advancement of education, and the impact of individuals with business expertise on the Board.

This chapter presents the findings from a case study of a Further Education College, which will be referred to as Mackay. Interviews were conducted with key actors, and where possible documents were obtained and analysed. The table below sets out the thirteen interviews which were conducted:

**Table 10.1: Interviews Conducted at Mackay** 

Interviewee Title	Position
Chief Executive	
(Principal)	
Vice-Principal (Finance)	
Board Secretary	External – from legal firm
Chairman	Retired Civil Servant
Board Member 1	Business background
Board Member 2	Support staff rep
Board Member 3	Academic staff rep
Board Member 4	IT specialist
Board Member 5	Local Government – Economic Development

Board Member 6	Managing Director of Consultancy Business
Board Member 7	Investment Banker
Board Member 8	Director of HR in large Public Sector Organisation
Board Member 9	Director of Finance in Large Charity

#### 10.2 Governance Issues

This first section will focus on the main issues which the board faced in governing the College. Specific issues from the existing Governance literature and also from various reports on College Governance were investigated. These issues – along with others which emerged - fell into a number of specific categories, namely: the role of the board; the distinction between strategy and operations, and how it impacts on the relationship between the board and the senior management team, and the composition of the board.

#### 10.2.1 The Role of the Board

The specific role of the nonprofit board has been highlighted as an important issue in governing nonprofit organisations (OSCR, 2006, Brown & Guo, 2010). Within this case the key role of the board was identified by all board members to some extent as a strategic/ visionary role, however within these definitions, a wide range of interpretations of the nature of this role were apparent. Two members in particular viewed this role in terms of compliance with Government policy rather than determining the overall direction of the College. The remaining members did describe the Boards role in terms of setting the strategic direction of the College, with varying levels of detail of what this actually entails. These responses may suggest that there is some acknowledgement and understanding that the Board role is strategic, but a lack of clarity exists over what this actually means (Harris, 1989; McAdam & Gies, 1985; Parker, 2003; Widmer, 1993). There were however descriptions which closely matched the various definitions of the Boards role outlined in the guidance literature on College Governance (see Association of Scotland's Colleges, *Guide for Board Members* (ASC, 2006)), for example, with

regard to setting the long term direction of the College and delegating to management.

In addition to the specific role of setting the strategic direction, some members described additional roles, many of which highlighted the emphasis the Board placed on the oversight role. Additional roles outlined included: ensuring that the management is effective in carrying out its functions and meeting its objectives; monitoring whether the College is meeting its objectives; ensure strategy for the College meets needs of all its stakeholders; ensuring financial stability for the College; ensuring quality of staff and student experience; ensuring value for money for the expenditure of public funds provided by the Scottish Funding Council; ensure provision of good education to students. Other key roles outlined included: bringing individual experience and expertise to bear for the benefit of the College; compliance with financial memorandum; financial responsibility in terms of the use of public funds; appointment of the Principal and overseeing their performance.

These findings highlight the tendency for board members to discuss their roles in terms of strategy, however the way that this term was conveyed varied between members. A plethora of literature and guidance exists on what the role of the Board should be (Houle, 1990; Axelrod, 1994), in addition, such examples which relate more specifically to Colleges include; the *Good Governance Standard for Public Services* (2004), and the Association of Scotland's Colleges *Guide for Board Members* (ASC, 2006). The general consensus in such publications is that this role should have a strategic focus, this is acknowledged by the board members, however, findings raise concerns as to whether there is a full understanding of what this role actually entails. There may also be an issue over how board members view the scope of their role. In addition, the role of the board in Colleges is perhaps somewhat confused by the title 'Board of Management', and the role outlined in the Further and Higher Education (Scotland) Act 1992 whereby the Board is charged with the responsibility for 'managing and conducting the College', however no definition of what is meant by this is provided.

This issue of the role of the Board will now be investigated in more detail below, focusing on key aspects of the role. These aspects include: the Boards role in terms of setting the strategic direction of the College, target setting and monitoring, and monitoring and controlling finance.

### 10.2.1.1 The Role of the Board in Setting the Strategic Direction of the College

The extent to which Board members were involved in setting the strategic direction of the College was investigated. This role was initially defined by many members as a key role of the Board, however, it became apparent that there was a lack of clarity and detail surrounding the Boards role in this process. Concerns were apparent from some members who highlighted the lack of time given to strategy, with one member noting that the only time the Board meets to look at the bigger picture is at the annual residential. The two main ways in which board member involvement was facilitated was the annual board residential, and through membership of the committees.

The board residential, or away day as it was also referred to was an event for board members which took place annually over two days. Many members described how attendance at this event allowed time for reflection, and gave board members the opportunity to focus on the 'bigger issues' affecting the College. In addition to the Board awayday, the other activities Board members felt enabled them to be actively involved in determining the strategic direction of the College was membership of the committees. Each board member was appointed to two committees, which were matched closely to their own specific expertise. The way in which the committees were structured allowed for board members to use their expertise to contribute both individually and collectively to specific areas.

Further activities which enhanced the strategic contribution of board members included board development sessions which were run prior to Board meetings, these generally took the form of a presentation from a member of the senior management team, or an external expert. The way in which meetings were structured was also

highlighted, whereby specific items which were forward looking, rather than backward looking were allocated a space within the meeting.

## 10.2.1.2 The Role of the Board in Target Setting and Monitoring

Board members were also asked about their role with regard to setting targets and monitoring performance against those targets. There was a divergence of opinion on the Boards involvement with this role. Three members noted that they felt the actual setting of these targets was operational, and should/ was done by management, board member 1 felt that monitoring was also an operational activity for example:

we're not directly involved in target setting and monitoring performance against targets it's really an operational matter, though this is perhaps one of the areas where there is some confusion and overlap. We should not – in my view – be setting targets but what we should be doing is approving or not, target setting and the realistic-ness of the targets that are set. We should be receiving reports on the monitoring, not monitoring it ourselves.

There appeared to be a debate with regard to the Boards actual role in target setting and monitoring, with some feeling that target setting in particular was an operational role. However, other member discussed the setting, refinement and monitoring of the world class performance indicators. The key tool here was identified as the WPIs, of which the Board members were very positive about the contributions they made.

Findings from a study by Miller (2002) of twelve nonprofit Boards show that Board members carry out the monitoring function according to their personal or professional competencies rather than considering measures that would show progress in achieving mission related goals and objectives, examples given include accountants concerning themselves with organising financial documents and lawyers focusing on legal or contract issues. This was not apparent from the responses to this particular question, however findings which will be highlighted in more depth later in this chapter do support this claim. In particular it was highlighted that those with accounting expertise played a key role in the challenge of accounting information.

#### 10.2.1.3 The Board Role in Monitoring and Controlling Finance

Both the interpretation and response to the question surrounding the effectiveness of the Board in controlling and monitoring the finances of the College varied widely. There was a divergence of opinion as to whether this was actually a board role.

Many members responded to this question in terms of how this role was carried out and identified issues, leading to the assumption that they did feel that this was a role the Board should undertake. Others however, commented specifically on the nature of this role in terms of the Board. For example, one Board member felt that the Board should not be monitoring and controlling the finances as this was operational and the Boards involvement should be when difficulties occur or approval is needed. Another member partly agreed with this, in that they felt the Board could not control the finances as this would be operational, however they felt the Board should monitor the Finances. Two other members highlighted the significant role that the Finance Committee played in this, and another member noted that in terms of the Board this role was 'clearly an absolutely key responsibility'.

There was a general agreement that the Board is effective in controlling and monitoring the Colleges finances, however, issues were identified which impacted on this, in particular around provision of information to the Board.

Another member commented on the reliance on management information and trust in the Executive:

The Board can only judge on what it's presented with; there's got to be a high level of trust in that what you're being presented with is accurate information, that's across any of the aspects, and finance probably being the most important one. As a member of the audit committee we get quarterly updates of income against expenditure and we get the final years accounts and I think that's as much as we can ask as a Board.

Also:

I think our ability to do that is continually improving through the improved management, through the improved reporting, through the improved regular accounts, and from my perspective, I see the Senior Management Team responding to the Boards requests, any requests the Board have had in relation to the financial management of the College.

Two members also highlighted the key role of Board members with an accounting/ financial background in the role of monitoring and controlling the Colleges finances, this is evident below:

There are individuals on the Board that are more inclined to scrutinise the figures than I might – I give them a courtesy glance and then say, they balance, that's fine – whereas there are other members of the Board that maybe have a background as being accountants and that's what they're happy doing. And I think the skills of the people on the Board, there is a good match of skills, there are a couple of number crunchers here, and question askers that like to crunch numbers and ask questions.

### Similarly:

I think that currently we have a couple of very effective Board members who are accountants – one's an accountant, one works in the finance industry, financial sector – and, you know, we as Board members rely on them to do a lot of the scrutiny.

Overall, although two members felt that there was an operational element to this role, there was a general feeling that controlling and monitoring of finances was a Board role and it was carried out effectively at the College. Issues were identified which did have an impact on this, namely: inefficient IT systems resulting in out of date information and reliance on management information/ trust in the SMT. In addition, the key role of Board members with an accounting/ financial background was highlighted in terms of being able to scrutinise and challenge the information presented.

Similar findings were documented in the Review of Scotland's Colleges Report (Scottish Executive, 2006). The majority of respondents in this study (33 of 36) felt that they had effective monitoring and control of their finances, the rigour of this

process was also found to be aided by Board members with a business background. In addition, many Colleges felt that improvements could be made with regards to real-time accuracy of financial management accounting information.

#### 10.2.2 The Distinction between Strategy and Operations

The above section investigated the key roles of the Board of the College, where a number of areas were subject to debate as to whether the Board should undertake the or not. This is a contentious issue - the distinction between strategy and operations, and the level the Board felt their involvement in each should be. There was a general feeling that there was a distinction between strategy and operations, however it was felt that this was not absolute and the existence of a grey area was perceived by many. Within this, some opinions varied, for example the following Board member was clear of the distinction between the two and that the Board should not be involved in operations:

I think the Board definitely should not be involved in operations and so there should be a clear distinction. The Board should be aware of what's going on in operations, particularly if any issues are likely to crop up, but they shouldn't go looking to be involved in those things and in an ideal world they wouldn't be involved at all, unless they're asked to bring specific expertise that they have by the management, but it's that management who should be responsible for operations.

Other Board members, however saw this as being more complex:

I think there's a clear understanding that we should have that division. I think when it comes to practice, drawing that line is a bit difficult, and as I think it varies with circumstance as well. I don't believe it's an absolute thing that can be pre-determined in advance and reduced to writing and we can read it and understand what we're meant to do. I think there is a bit of sucking it and see in that regard - for both the leadership team and the board.

Such a distinction between strategic and operational activities presents itself as a challenge in all organisations (Baysinger & Hoskisson, 1990). When boards stray

into operations, this is often a fundamental cause of tensions between the board and management (Cornforth, 2004; Chait & Taylor, 1989; Taylor et al, 1996). Two examples of such tensions caused by determining whether activities are strategic or operational are apparent in this case. The first considers a situation when the senior management team decided to hire part-time lecturing staff from and agency, and the second when the principal restructured the senior management team.

#### Recruitment of Agency Staff

This situation came about when the senior management team attempted to source temporary lecturing staff through an agency. The Principal was quite certain that this was a totally operational decision. The Board were not consulted on the decision and instead, were presented with a proposal with a decision that had already been taken, which was that the SMT were going to use the agency to recruit temporary lecturing staff. The Chair of the HR Committee noted that he was immediately concerned about the impact this would have on employee relations. However, by the time he found out about the proposal, it had already been communicated to the outside. The result of this proposed move caused issues with the teaching trade unions. It was at this point that both the Chair of the Board and the Chair of the HR Committee became involved, they were assured by the SMT that the discussions with the teaching unions were going well and that any issues would be worked out. This turned out not to be the case and resulted in strike action by staff; two one-day strikes.

The Chair of the HR Committee believed that as a Board, they had 'went out on a limb' to support the SMT. Although the Chair of the HR Committee noted that both himself and the Chair of the Board had made it clear to the Principal and Vice-Principal that they had no choice but to support them after they had been put in such a difficult position. Due to the fact that they had already made the decision public the Chair of the HR Committee noted that as a Board they either had to support them, or come out and say that they didn't, which he highlighted would have been a huge loss of face for the leadership team. The decision was made by the Board to support the management team, they informed them of the importance of successful negotiations

with the trade unions. When these discussions failed they decided that they would take the criticism resulting from the strike action and that they would see it through – which they highlighted was very important. However, after the second one-day strike, the Chair of the HR Committee then noted how he received an email that had been sent around the College from the Principal saying that they were scrapping the proposal to use the agency: they had collapsed in the face of the strike. The Chair of the HR Committee had previously made it clear to the Principal that it was imperative that they did not collapse and if they were going to collapse part way through, they shouldn't do it.

As a result of this, the Chair of the Board and the Chair of the HR Committee had a number of separate meetings with the Principal and Vice-Principal to review their actions, because the Board felt that that what they did in this situation was make a strategic decision – which was about using the agency – and they didn't involve the Board it in at all. Instead, the SMT saw it as an operational decision. However, the Board viewed this decision differently, as it was a huge issue for the College, there were questions asked in the Scottish Parliament about it and MPs wrote to the Chair of the Board about it. The Boards view was that the management team misjudged the situation in terms of their decision not to come to the Board with it first to get them onside. The outcome of this was a increased awareness of this issue and it was decided that the issue of strategic and operational decisions is a judgement call, with the key to avoiding similar situations being keeping both sides informed, specifically with regular meetings between the Chair and the Principal.

## Restructuring of the SMT

Another example of this tension between strategy and operations can be seen with regard to a restructuring of the senior management team. The Principal felt that this was an operational decision, as can be seen below:

I do believe there's a grey area, and I often see this at Board meetings. I mean I've just...we've got 12 in our senior management team and 3 are leaving and what we've done is we've restructured our work at the senior management team level, people taking on different responsibilities. Now

fortunately the Chair of the HR Committee is very clear that that is operational, that's my responsibility, on the other hand I know at the next Board meeting there will be questions about why have I done this why have I done that, why haven't I got a Director of Human Resources anymore. So that for me is a very grey area where I sit because Board members see it in black and white and I end up with the grey in the middle.

The acknowledgement by the Principal that Board members would questions the nature of this decision was correct, with the Chair of the Finance Committee frustrated that he was not informed of this decision:

I don't know if you know but they've just done a reorganisation of the senior management team, that was announced this week, Monday, I learnt about it on Friday last week, a week ago, simply because the Director of Finance happened to be in for another meeting and mentioned it to me, and part of that involves a change in his role and his position, and I personally have great difficulty understanding how that has happened with not even a conversation with me. Now having said that, I have perfect understanding that the Principal has his prerogative to organise the staff as he thinks appropriate, however, I do find it odd if we are Board members, in my case we have specific responsibilities in terms of the Finance, if something is effecting the chief financial officer and I'm not involved in any way —I'm not suggesting that I have a veto or anything else, but at least involved in the discussion- it's a little difficult.

Although the dominant response to this question was that Board members should not be involved in operations, throughout the interviews various desires were expressed from Board members to become more involved in certain aspects of the College, some of these could be perceived as being operationally focused, adding to the complexity of this issue. Examples included: when considering the relationship between the Board and the SMT, two members felt that there was scope to have more input in the decision making process, rather than just being presented with final decisions to approve. In addition, one member was keen that Board members should be allocated a specific school to be involved with, this could be perceived as operational.

#### 10.2.3 Board Composition

The composition of the board is outline below:

Table 10.2 Board composition of Mackay

Interviewee Title	Position
Chief Executive	
(Principal)	
Vice-Principal (Finance)	
Board Secretary	External – from legal firm
Chairman	Retired Civil Servant
Board Member 1	Business background
Board Member 2	Support staff rep
Board Member 3	Academic staff rep
Board Member 4	IT specialist
Board Member 5	Local Government – Economic Development
Board Member 6	Managing Director of Consultancy Business
Board Member 7	Investment Banker
Board Member 8	Director of HR in large Public Sector
	Organisation
Board Member 9	Director of Finance in Large Charity
Board Member 10	Director of External Relations in Public
	Sector
Board Member 11	Owns of Business
Board Member 12	Retired Accountant

Many Board members commented on the very good mix of skills currently on the Board. Many felt that the skills of the Board should reflect the needs of the College, and should evolve as the College evolves. Examples of some additional areas of expertise which were suggested included: Marketing; Finance and Accountancy; IT; Political background; Local Government; Community Representatives; and someone from another FE College or education institute who could challenge what was happening on the Education front. In comparison, a report by Davis (2002) found that with regard to the employment background of Board members of English Colleges 37% were from business, 12% were co-opted, 10% were staff, 7% were from the community, 6% were from the local authority, and 8% were unfilled vacancies.

Issues surrounding recruitment were highlighted, in particular a previous recruitment drive in which the number of potential candidates dropped significantly when they discovered the posts were not remunerated.

#### 10.3 The Impact of Business Expertise on the Board

This section considers the impact of business on the board of Mackay by considering: the role of the individual board member and the use of the experience and expertise.

#### 10.3.1 The Role of the Individual Board Member

Board members were asked what they felt their most important role as an individual Board member was. The responses to this question tended outline a broader role in line with role of the Board previously outlined rather than bringing specific functional expertise. For example, one Board member with an HR background noted: 'Well I suppose it's in line with what I think the role of the board is, which is to provide a strategic direction or lead. Also to be an ambassador for the college and an advocate for the college'. Similarly, another Board member with a Business background also talked generally about their experience, this time in terms of working towards the fulfilment of College objectives:

It's helping the senior management to achieve the agreed objectives for the College and that's using whatever expertise and experience I have to do that, so it may be that sometimes I'm asked to go and have a one to one meeting with somebody to help them on some point, it may be sitting at a Board meeting with the rest of the Board analysing a particular issue, it may be meeting people from a department in the College and giving them encouragement and advice or...it's in whatever capacity is appropriate providing the benefit of whatever experience and expertise I have. So I think it's pretty clear, the responsibility is to provide the expertise and experience, but it may be needed in lots of different ways and I have to be flexible enough to fit in with what's appropriate. And what's appropriate for me as a Board member might be quite different from somebody else.

And:

Well for me it is very much about compliance with the Scottish Funding Council Financial Memoranda. It's for the management, administration of revenue and property and conduct of affairs. And that's what statutorily we are charges to do. So it has to be the top priority.

Another member with a business background described their role in terms of the role of the board, but also noted the emphasis on their specific expertise:

I don't see my role as a lot different to those things I gave you at the outset about the generic role of the board or Board members. I mean, I think I do the same thing. I suppose I do that with maybe two slight areas of emphasis; one is generally on the people issues within the College, and the other one —which is linked to the people issues— is about one of the things that I do now, and have done for a number of years, is lean process improvement.

These board members with business backgrounds see their role as broad and in line with other board members. They do not appear to bring their specific expertise to the board, this would point towards their presence as enhancing the legitimacy of the board (Meyer and Rowan, 1977). This issue is explored in more detail in the following section.

#### 10.3.2 The Use of Individual Boards Members Experience and Expertise

The issue surrounding the extent to which Board members feel their experience and expertise is being used was also investigated. It was established through discussions with both the Chair and the Principal that there were procedures in place to ascertain the expertise possessed by Board members, for example; one to one meetings with the Chair and each Board member to establish the individual Board members interests, and the placing of Board members onto Committees with consideration of their specific expertise. However, there was a varied response from Board members in terms of whether they felt that best use was actually being made of their skills. This section specifically considers the impact of Board members with specific

business expertise, including; finance, IT, lean process improvement, accounting and HR.

#### **Finance**

There was one Board member who worked in the Finance Industry, this person was positive in terms of the way in which his expertise was used and gave a number of examples of how he was using his expertise within the College:

...at the moment I'm trying to get the College to look at —in terms of the commercial company, to think about - the financial services sector, and I think both (the Chair) and I believe that financial services is a huge contributor to both the Scottish and in particular, the (city) economy and therefore, the College should be servicing that sector in some shape or form, and it doesn't really at the moment. So, trying to get the College to think about what it might do in that area, for example, through the provision of commercial training courses; so looking to provide introductions for College members, to other organisations in the sector, so that they can get a feel for what are the issues that the sector has, and how the College might be able to respond to those.

This examples shows an apparent desire from this member to use his background to contribute to the College, however, at present this seems aspirational.

#### Information Technology

The insufficiencies of College IT systems were highlighted during this research, with members highlighting the time lag in receiving up-to-date information. There was one member on the Board who previously owned an IT services company, and had been recruited by a previous Finance Director who had contacted him with regard to issues the College was having with its IT systems. The contribution made by this member is outlined below:

for example, because I have pretty good experience in IT I'm constantly being asked to be involved with things to do with IT strategy and that kind of thing. But also because I have experience in the private sector, people try to call on that from time to time, and that's in different ways...for example, just a couple of weeks ago one of the departments in the College which had a relatively poor report from HMIE a year or two ago was business management, now that department since then has been reorganised and is now business, technology and tourism (BTT) and they

had a planning day and I was asked if I would go along to that for part of the time at least; first of all to listen to what they were trying to achieve, but then to give them a wee chat about some ideas I had and then to be involved in some of their discussions. And I think, that was because of the background I have, because of my economics I was kind of on their wavelength, but also because I've come from the private sector, some of the things they're trying to do are relevant. So I think that actually that was a very good use of my time.

This is a further example of providing advice to the organisation, however it is not apparent that this has had clear instrumental value to the College, this role was perhaps more symbolic.

#### Human Resources

Another had a HR background in the private sector, and had specific interest in continuous improvement, and in particular lean process improvement:

I'm a big fan of processes so a lot the stuff that the College has done around KPIs, and around the World Class Indicators, and around improving processes has been initiated by me...I've done some stuff for the College as well in terms of trying to help some of their managers develop skills in the area of lean process improvement so I've done some training with their managers. So in that respect I think that the College have used my expertise quite a lot...and it's quite difficult as well because you cannot go along wearing a Board member hat, I've, for example, done a couple of presentations to the senior management team on lean process improvement, and you know, I try and say at the outset there, I'm not here as a Board member, I'm here to talk to you about this, but I think the people there are always probably a little bit, 'he is a board, member, I'm not going to say too much' kind of thing. So again there's a difficult balance to strike there, but I think they've used my expertise in those areas quite effectively. I've been given the opportunity to contribute there and to add value to the College management team and I think similarly with HR as well.

This business member has clearly had an instrumental role in the organisation, with the introduction of KPIs and training on lean process improvement.

#### Accounting

There was one accountant on the Board, who described his role in interpreting the financial information presented to the Board:

I think, unlike colleagues who are not familiar, I have (a) an understanding of being able to challenge the executive in terms of – how can I describe this...I guess it's the ability to understand how the finances translate into how...in terms of the performance of the College and providing that, if you like, I have the ability, if you like, the finance is discussed to say yey or nay as to yes this is ok, is it material, should we be concerned about this, what should we be concerned about, what are the issues here. So if you like, there's an element of translation from taking a set of numbers to making it, not idiot proof, that's the wrong expression, but if you know what I mean

Although this member highlighted his role in terms of understanding and explaining accounting information, they felt there were other areas in which they could make a significant contribution, and they also questioned how much the SMT knew about Board member backgrounds:

I've been involved in business, particularly in commercialisation, so I'd like to believe I've got a fair bit of experience, know-how, and have a lot to offer in helping in that thing... Barely ever been discussed, there's one or two things where it's sort of come to light, but...for example what they've just done, they're establishing a commercialisation; waterfront enterprises it's called, which one of my colleagues on the Board is, he's actually stood down from the Board but he's going to, so he can be the acting Chief Executive while they set it up. But that sort of process is what I've been here for the last 4,5,6 years or so, and yet I've had no contact about that at all, no issues about that, so yeah, that part of my expertise and background is not tapped into. And I'm sure that's true of one or two others. The only thing I would say, what'll be interesting in this exercise is how much, I think you endeavour to try and do this, but how much the Executive, or even just the SMT, how much they know about the Board members know-how and background. I would say quite honestly it's a very good Board, just thinking across the individuals, there's a plethora of skills and know-how...I'd be interested in who contacts the Board members. I've only ever really been contacted by the Principal and the Director of Finance in X years, that's pretty limited contact, I'm not saying they don't deal with anybody else but then in turn nobody's ever picked up the phone and asked me about something.

This is clearly an example of legitimation.

#### 10. 4 Conclusion: Mackay College

Previous studies on governance in charities have highlighted issues regarding the role of the board, in particular the confusion of board members with regard to their role, and a tension when board members attempt to become involved in what are perceived as managerial issues. In this study the role of the board of a FE College was investigated. Although on the surface they communicated their perception of this role as being strategic, it was apparent that beyond this there was a seeming lack of clarity over which specific activities they should be including in their role. Within this, there seemed to be differing interpretations in terms of what this strategic role encompassed, from whether it was a solely compliance role, or something more visionary and taking the College forward. Looking at more specific roles, board members were able to convey their involvement in setting the strategic direction of the College, and clearly viewed this as being part of their remit, however, comments that the setting of the strategic direction at the board away day was the only time the board looked at the longer term picture rather than the 'day-to-day' issues may indicate a tendency to stray into the more operational side of things as described in the literature. Other roles investigated all brought out a divergence of opinion as to whether these were within the scope of the boards roles. Targets setting and monitoring, and monitoring and controlling finance, were all viewed by some members as being the role of management.

The second aspect of this study was the existence of tensions caused when the board stays into matters which are viewed as operational. The board highlighted difficulty in distinguishing between strategic and operational activities, noting the existence of a grey area in many cases. A widely used example was the case of the recruitment of temporary staff from an agency. The Principal viewed this as a management decision, the board, however, viewed it as strategic, due to the impact the decision had on the College. In the aftermath of this example, the decision was taken to increase communication between management and the board, however other instances of the difficulties in defining a dividing line between the roles were apparent. A Further example was outlined whereby the restructuring of the SMT was undertaken by the Principal – viewing this as an operational activity. When the role

of the Finance Director was changed, the Chair of the Finance Committee felt he should have been consulted, as this was strategic. There is clearly confusion in existence with regard to the exact role of the charity board, with many tensions apparently created through the differing interpretations of activities which are perhaps more down to judgement and interpretation. It may be apparent that the division of activities is perhaps less of a boundary line and more of a boundary area occupied by both the board and management.

Finally, the impact of business expertise on the board was considered. The majority of business people tended to view themselves as having a broad role in line with the general role of the board, rather than to bring their own specific functional expertise. Although practices were identified by the chair and the chief executive to ascertain and utilise board members experience, there was a general sense that they served more of a legitimating role than an instrumental one. This was with the exception of the HR professional, who noted that he was instrumental in bringing in KPIs and also did training with management on lean process improvement.

#### **PART THREE**

#### **CHAPTER 11**

#### **CONCLUSION**

This thesis has investigated governance in Scottish charities. At the outset, two research questions were identified, the first sought to identify the main governance issues charities face, this was internally focused, and specifically related to the functioning of the board. The second research question addressed the issue of the influence of business on charities through examining the impact of business expertise on charity boards. This was a multi-method study utilising both a survey and a series of case studies, and social construction and institutional theory were used as the framework for the analysis. In particular, ideas of institutional isomorphism (Meyer and Rowan, 19977; Di Maggio and Powell, 1983) have been used to explore the similarities of governance practices between charities, and to offer a possible explanation for the widespread recruitment and impact of business people onto charity boards. Ideas of legitimation (Berger and Luckmann, 1967; Meyer and Rowan, 1977) were used to consider the recruitment of business people onto charity boards and the adoption of business practices. This chapter will initially present an overview of the findings, the main contributions of this research will then be outlined, and some suggestions for future research will be presented.

### 11.1 An Overview of Key Findings

This section will present findings from each of the empirical chapters, before drawing together key aspects around the themes: governance issues; and business expertise.

#### Governance: An Overview of the Largest Section of Scottish Charities

The results of a survey of the largest section of Scottish charities were presented in chapter 6 and 7. Chapter 6 focused on governance issues related to: the role of the board, including how well defined roles were, how they were communicated to board members, and how awareness of roles was ensured among board members; the proportion of board time spent on strategy and operations, and the types of strategic and operational roles carried out by the board; the effectiveness of governance in charities; and control and regulation in charities.

On the surface, charities reported strong governance frameworks, with clearly defined roles and responsibilities, a predominant focus on strategy, and overall effectiveness in the governance of their organisations. However, further interrogation of the data identified some possible inconsistencies within these results. Respondents reported clarity in role definition, such a finding was in contrast to the literature where lack of role definition is reported. Although charities reported documenting their roles clearly, this may be in order to present themselves as legitimate to the outside world, with these roles not necessarily being carried out. This would be supported by further findings from the survey where a lack of clarity was found to exist in terms of what charities perceived to be strategic and operational tasks, low level activities were also shown to be carried out by the board. This could point towards a mismatch between role definition and fulfilment. Such a cause could be related to the communication of roles, which was predominantly through formal governance documentation, it is possible that these are not necessarily regularly referred to by board members, providing further evidence for the documentation of roles for legitimacy purposes (Meyer and Rowan, 1977; Parker, 2007).

Charities also highlighted a number of ways in which governance could be improved, largely relating to board structure and training. Issues were also raised around difficulties in adhering to regulation. The above factors suggest that these organisations may not be as effective as reported. These findings point towards the existence of loosely coupled organisations (Meyer and Rowan, 1977), where the

external façade of the organisation, namely the formal rules of the organisation such as the documented roles and responsibilities conflict with the inner core or the organisation and the work that is actually being carried out. Organisations document and report adhering to prescribed governance practices in order to maintain legitimacy (Meyer and Rowan, 1977).

## The Impact of Business Expertise: An Overview of the Largest Section of Scottish Charities

Chapter 7 focused specifically on the impact of business expertise on charity boards, and presented findings from the largest section of Scottish charities. The main issues investigated included: the proportion of business people on charity boards; the impact of business people on charity boards, including the main practices introduced by business people and how they were received within the organisation; the motivations of business people, their understanding of the charity context; and the impact of business people on public confidence.

The findings highlighted the widespread recruitment of business people on charity boards. This was attributed to coercive isomorphism (Di Maggio and Powell, 1983), with pressure from the Scottish charity regulator to appear more business-like. Charities reported a general satisfaction with business people on their boards, with some wishing for more. In contrast to the literature outlined in chapter 3, relatively few organisations expressed concerns that such individuals may not fully understand the nature and values of the organisation.

The impact of business people on charity boards was reported as two-fold, in the form of a general business expertise, or specific functional expertise, with some charities reporting both. However, due to the nature of the survey data, it was difficult to fully understand the extent of the impact of such expertise and reported practices, and whether they had been fully adopted or were merely a legitimacy tool. Some previous studies have reported that the reaction within charities to the

implementation of business practices is that it undermines the charities fundamental ethos (e.g. Landsberg, 2004), however the survey results point towards a passive acceptance of the adoption of business practices. This acceptance may be due to the lack of impact that these 'business practices' had on the organisation – many were quite minor innovations rather than fundamental changes – meaning they may have a merely symbolic role within the organisation.

With regards to the motivations of business people serving on non-profit boards, the main reasons were personal. This raises questions around the intentions of these business people, and whether they are intent on using their expertise to bring business practices and fundamental changes to the organisation, or whether they are there for reasons of personal fulfilment. It was also found that charities view business people as having a role in increasing public confidence in the organisation. Such a finding would suggest that charities recruit business people for legitimation – in order to appear modern and up-to-date (Meyer and Rowan, 1977).

#### Governance in a Social Care Charity: Some Case Study Evidence

Chapter 8 reported findings from a case study of a social care charity. At the centre of this case study was the adoption of a private sector governance model, with a board member from the private sector being instrumental in its adoption. The board cycle was at the centre of the governance structure in this organisation, and was brought into the organisation by a board member with business expertise, in this case, a management consultant. Meyer and Rowan (1977) suggest that the adoption of such practices and recruitment of such people are adopted ceremoniously, having a merely symbolic role, in order to present the organisation as legitimate to its external environment. However, it is apparent from this case that this model has become embedded within the organisation. It has a clear impact on governance, where on an annual basis, the main roles of the board are addressed: direction setting, strategic thinking, monitoring and evaluation, accountability. This translates into the monthly board meetings, which are themed around these areas and therefore focusing

the board and their role, and keeping them from straying into more operational matters.

The successful implementation of the model was the result of a shared vision between the chief executive and the management consultant, who worked together to implement the board cycle within the organisation, the rest of the board were also receptive to the change. It has been reported that tensions can exist between the board and management (Chait & Taylor, 1989; Taylor et al 1996; Cornforth, 2003), however in this case a partnership approach appears to have been adopted between the Chief Executive and a Board Member in order implement of new model of governance.

This case provided an example of the successful implementation of a model of governance from the private sector, this is in contrast to a growing literature which highlights concerns over charitable organisations importing models used in the private sector (Bush, 1992; Backman & Smith, 2000; Eikenberry & Kluver, 2004; Foster & Bradach, 2005; Harris, 2001; Landsberg, 2004). Such research suggests that business practices can undermine the values of the charity, however in this case it is made clear by the management consultant, that governance is driven by the values of the organisation.

In contrast, the other two board members with business expertise – a public and private sector accountant – did not appear to be so instrumental in contributing to the effective governance of the organisation. Their role appeared to consist of a more general oversight role. In particular, the private sector accountant was noted as having difficulty in adapting to the different context of the charity sector. The lack of any instrumental contribution can be viewed as a possible example of legitimation (Meyer and Rowan, 1977), where the recruitment of these individuals reflects a desire of the organisation to present itself as modern and up-to-date.

#### Governance in a Housing Association: Some Case Study Evidence

This case confirmed the existence of a number of issues which are apparent in the literature on nonprofit governance. There appeared to be a lack of clarity in their role of the board, with a perceived lack of understanding of the full extent of the role. In addition, the board appears significantly operationally focused. Findings from this case confirmed reports from the Scottish Housing Regulator (2009) that boards become caught up in low-level operational issues. This could be due in part to the high level of tenant members on the board, which is viewed as a contributing factor in the Boards involvement in activities that are operational, due to the difficulties they have in adapting to the strategic nature of the role. It has been reported that tenant members view their role as being that of a representative, tending to focus on operational housing issues, at the expense of strategic issues impacting on the entire organisation (Audit Commission, 2004). This is a potentially significant issue in Scottish Housing Associations, where around 45% of volunteer Board members in the sector are tenants (Scottish Housing Regulator, 2009).

This apparent tendency for the board to stray into operational matters is in contrast to the literature which suggests tensions exist when boards stray into operational matters. In McPherson, the chief executive appears to take operational matters to the board, possibly to seek approval. This points towards a shared approach to both the governance and management of the organisation, perhaps at times with little separation between the two. There was also a lack of a clear governance structure, although a recent review brought about structural changes in terms of meeting numbers and committee structure these appeared to be for reasons of legitimacy rather than reflecting increased effectiveness.

In terms of the presence of business people on the board, they do not appear to have any instrumental value. Although there appears to be a desire from board member 1 to have more of an impact, the current structure does not seem to allow for this, meaning this role is limited and symbolic (Meyer and Rowan, 1977).

#### Governance in a Further Education College: Some Case Study Evidence

In the FE College there was an apparent lack of clarity over which specific activities the board should be undertaking. There were differing interpretations in terms of what this strategic role encompassed, from whether it was a solely compliance role, or something more visionary and taking the College forward. Looking at more specific roles, board members were able to convey their involvement in setting the strategic direction of the College, and clearly viewed this as being part of their remit, however, comments that the setting of the strategic direction at the board away day was the only time the board looked at the longer term picture rather than the 'day-to-day' issues may indicate a tendency to stray into the more operational side of things as described in the literature. Other roles investigated all brought out a divergence of opinion as to whether these were within the scope of the boards roles. Targets setting and monitoring, and monitoring and controlling finance, were all viewed by some members as being the role of management.

The board highlighted difficulty in distinguishing between strategic and operational activities, noting the existence of a grey area in many cases. A widely used example was the case of the recruitment of temporary staff from an agency. The Principal viewed this as a management decision, the board, however, viewed it as strategic, due to the impact the decision had on the College. In the aftermath of this example, the decision was taken to increase communication between management and the board, however other instances of the difficulties in defining a dividing line between the roles were apparent. A Further example was outlined whereby the restructuring of the SMT was undertaken by the Principal – viewing this as an operational activity. When the role of the Finance Director was changed, the Chair of the Finance Committee felt he should have been consulted, as this was strategic. There is clearly confusion in existence with regard to the exact role of the charity board, with many tensions apparently created through the differing interpretations of activities which are perhaps more down to judgement and interpretation. It may be apparent that the division of activities is perhaps less of a boundary line and more of a boundary area occupied by both the board and management.

Finally, the impact of business expertise on the board was considered. The majority of business people tended to view themselves as having a broad role in line with the general role of the board, rather than to bring their own specific functional expertise. Although practices were identified by the chair and the chief executive to ascertain and utilise board members experience, there was a general sense that they served more of a legitimating role than an instrumental one. This was with the exception of the HR professional, who noted that he was instrumental in bringing in KPIs and also did training with management on lean process improvement.

## **Concluding Comments**

This thesis set out to investigate two specific aspects of charity governance, the first was surrounding the main issues charities face in governing their organisations, and the second related to the impact of business expertise on charity boards. Two major findings are therefore presented around these two areas:

## Issues in Governing Charities

With regard to the issues charities faced in governing their organisations, these tended to be centred around two areas: the role of the board, and the distinction between strategy and operations and how this impacted on the relationship between the board and management. To a large extent, the situation in Scottish charities reflects the situation elsewhere, as documented in the review of literature in chapter 2.

#### The Impact of Business Expertise on Charity Boards

The results presented mixed findings on the impact of business people on charity boards. Their role was largely symbolic, however substantive impact was reported in the social care case. However, what is apparent is the lack of the severe undermining of values that is reported in some previous studies in this area. Business people can be a valuable resource for charities, and their ideas are not always at odds with the

values of the organisation. Charities must be able to harness the skills of business people in order to help achieve their missions.

#### 11.2 Contribution of Research

This research makes a contribution in a number of ways, these are outlined below and encompass: a contribution to literature on charity governance; and a contribution to policy and practice.

#### Literature

This research contributes to the existing literature on charity governance, with regard to both the research context and the specific focus of the impact of business people on charity boards.

In terms of the research context, it was highlighted at the outset that although there is a considerable amount of research on charity governance, research on governance in Scottish charities was scant, with none featuring in any of the academic literature. Two recent research reports have focused on the Scottish context, notably Crawford et al (2009) and Kakabadse et al (2009). However, this thesis focuses on different aspects. The Crawford et al study is very much focused on 'external' aspects of governance and Kakabadse et al (2009) focuses solely on board effectiveness and utilising only a survey.

Another contribution is specifically in relation to the focus on the impact of business expertise on charity boards. Although other authors have touched on this area in broader governance studies, there have been few that have given specific attention to this area. Of those that have, they have tended to be quantitative in nature, such as Austin (1998). This research therefore offers an insight into the role of business people on charity boards and highlights the potential value they can provide as board members. A particularly interesting insight is offered through the Robertson case

study which highlights the significant value that business people can have on these organisations.

## Implications for Policy and Practice

This research will also be useful at a practical level, it will be of interest to charities, regulators and policy makers as it offers an insight into the functioning of charity boards and identifies significant issues which large charities face in governing their organisations.

One specific issue is that of Housing Associations, where regulatory failings are contributing to governance issues. At present, there are no term limits on how long board members of Housing Associations are allowed to serve, there is also substantial pressures on these organisations to have significant tenant membership on the board.

It has also helped to identify the widespread recruitment of business people in charities. The research suggests that these individuals have a largely symbolic role, however in the case of the social care charity, business people can be a valuable resource that charities can learn from. Charities should ensure that they utilise business members on the board.

## Suggestions for Future Research

The focus of this thesis was on the governance of large Scottish charities, with a specific emphasis on the impact of business people on the board. In this section I will offer some suggestions on the possible way forward for this area of research.

The focus of this thesis was on large charities, the rationale behind this was because in smaller charities there is less distinction between the board and management. In larger charities business practices were not adopted within the organisation. It may however, be relevant to consider the incidence and impact of business people on smaller charities to see whether they had greater impact. The focus of this thesis was on business people, with the majority of the board member interviews conducted with individuals with a business background. A further aspect of investigation may be to consider in more depth the views of non-business members of the boards of charities to consider their views on how business people's ideas sit with the vales of the organisation, and the nature of their impact.

The issue of the influence of business on charities has been highlighted as a contentious issue within these organisations. This focus of this thesis was specific in considering this issue in relation to business people sitting on charity boards. This business influence could be investigated at other levels of the organisation, for example at a top management level and functional level. It would be interesting to investigate the impact of for-profit managers, accountants on charities and whether such individuals try to implement business techniques in the organisation.

#### REFERENCES

Abzug, R. & J. Galaskiewicz (2001) Nonprofit Boards: Crucibles of Expertise or Symbols of Local Identities? *Nonprofit and Voluntary Sector Quarterly* 30(1) pp. 51-73.

Alcock, P. & Scott, D. (2005) Introduction in 'Close Work', Alcock & Scott (eds) Charities Aid Foundation.

Alexander, J.A. & B.J. Weiner (1998) The Adoption of the Corporate Governance Model by Nonprofit Organizations. *Nonprofit Management and Leadership* 8(3) pp. 223-242.

Arnaboldi, M. and I. Lapsley (2003) Activity Based Costing, Modernity and the Transformation of Local Government. *Public Management Review* 5(3) pp.345-375.

Arnaboldi, M. and I. Lapsley (2004) Modern Costing Innovations and Legitimation: A Health Care Study. *Abacus* 40(1) pp.1-20

Audit Commission (2004) *Housing – Improving Services through Resident Involvement*. National Report. London: Audit Commission.

Austin, D.M. & C. Woolever (1991) Voluntary Association Boards: A Reflection of Member and Community Characteristics. *Nonprofit and Voluntary Sector Quarterly* 21(2) pp. 181-193.

Austin, J.E. (1998) Business Leaders and Nonprofits. *Nonprofit Management and Leadership* 9(1) pp. 39-51.

Axelrod, N.R. (2005) 'Board Leadership and Development' in Herman & Associates The Jossey-Bass Handbook of Nonprofit Leadership and Management (2<sup>nd</sup> Ed) Jossey-Bass, San Francisco

Backman, E.V. & S.R. Smith (2000) Healthy Organizations, Unhealthy Communities? *Nonprofit Management and Leadership* 10(4) pp.355-373.

Baker, T.L. (1994) *Doing Social Research* (2<sup>nd</sup> Ed), New York: McGraw-Hill Inc.

Baysinger, B.D. & R.E. Hoskisson (1990) The Composition of Boards of Directors and Strategic Control: Effects on Corporate Strategy. *Academy of Management Review* 15(2) pp. 72-87.

Berger, T. and P.L. Luckmann (1967) *The Social Construction of Reality*. Penguin Books: London.

Bielfield, W (2006) Quantitative Research for Nonprofit Management. *Nonprofit Management and Leadership*, 16(4) pp.395-409.

Bowen (1994) When a Business Leader Joins a Nonprofit Board. *Harvard Business Review* Sept-Oct pp. 38-43.

Bradley, P., P. Jansen & C. Silverman (2003) The Nonprofit Sector's \$100 Billion Opportunity. *Harvard Business Review* May pp.94-103.

Bradshaw, P., V. Murray & J. Wolpin (1992) Do Nonprofit Boards Make a Difference? An Exploration of the Relationships Among Board Structure, Process, and Effectiveness. *Nonprofit and Voluntary Sector Quarterly* 21(3) pp. 227-249.

Brown, W.A. (2005) Exploring the Association Between Board and Organizational Performance in Nonprofit Organizations. *Nonprofit Management and Leadership* 15(3) pp. 317-339.

Brown, W.A. (2007) Board Development Practices and Competent Board Members. *Nonprofit Management and Leadership* 17(3) pp. 301-317.

Brown, W.A. & C. Guo (2010) Exploring the Key Roles for Nonprofit Boards. *Nonprofit and Voluntary Sector Quarterly* 39(3) pp.536-546.

Brown, W.A. & J.O. Iverson (2004) Exploring Strategy and Board Structure in Nonprofit Organizations. *Nonprofit and Voluntary Sector Quarterly* 33 (3) pp. 377-400.

Brudney, J.L. & V. Murray (1998) Do Intentional Efforts to Improve Boards Really Work? The Views of Nonprofit CEOs. *Nonprofit Management and Leadership* 8(4) pp. 333-348.

Brunsson, N. and K. Sahlin-Anderson (2000) Constructing Organizations: The Example of Pubic Sector Reform. *Organization Studies* 21(4) pp.??

Bryman, A. (1989) *Research Methods and Organization Studies*. Chapter 2: Measurement in Organizational Research: Questionnaires and Structured Interviews. P. 34-70. Routledge

Bryman, A. (2001) Structured Interviewing in *Social Research Methods*, Oxford University Press, Oxford

Bush, R. (1992) Survival of the Nonprofit Spirit in a For-Profit World. *Nonprofit and Voluntary Sector Quarterly* 21(4) pp.391-410.

Cabinet Office (2010a) Building a Stronger Civil Society: A strategy for voluntary and community groups, charities and social enterprises. Office for Civil Society: Cabinet Office, London.

Cabinet Office (2010b) Building the Big Society. Cabinet Office, London.

Callen, J.L., A. Klein & D. Tinkelman (2003) Board Composition, Committees, and Organizational Efficiency: The Case of Nonprofits. *Nonprofit and Voluntary Sector Quarterly* 32(4) pp. 493-520.

Carver, J. (2006) *Boards That Make a Difference* (3<sup>rd</sup> Ed) Jossey-Bass, San Francisco.

Chait, R.P. & B.E. Taylor (1989) Charting the Territory of Nonprofit Boards. *Harvard Business Review*. Jan-Feb pp. 44-54.

Charities and Trustee Investment (Scotland) Act (2005)

CIPFA (2004) The Good Governance Standard for Public Services. CIPFA: London

Cornforth, C. (2001) What Makes Boards Effective? An Examination of the Relationships Between Board Inputs, Structures, Processes and Effectiveness in Non-Profit Organisations. *Corporate Governance* 9(3) pp. 217-227.

Cornforth, C. & C. Edwards (1999) Board Roles in the Strategic Management of Non-profit Organisations: Theory and Practice. *Corporate Governance* 7(4) pp. 346-362.

Cornforth, C. (2003) The Changing Context of Governance – Emerging Issues and Paradoxes in C. Cornfoth (ed) *The Governance of Public and Non-Profit Organisations: What Do Boards Do?* Routledge: Oxford

Cornforth, C. & C. Simpson (2002) Change and Continuity in the Governance of Nonprofit Organizations in the United Kingdom. *Nonprofit Management and Leadership* 12(4) pp.451-470.

Crawford, L., T. Dunne, G. Hannah, & L. Stevenson (2009) *An Exploration of Scottish Charities' Governance and Accountability*. Institute of Chartered Accountants Scotland: Edinburgh.

Crotty, M. (1998) The Foundations of Social Research. Sage Publications. London

Dart, R. (2004) Being "Business-Like" in a Nonprofit Organization: A Grounded and Inductive Typology. *Nonprofit and Voluntary Sector Quarterly* 33(2) pp.290-310.

Davidson, L. (2003) Charity Boss Denies £3 m Cash Scandal. Edinburgh Evening News, Wed 15 October 2003

Dees, J.G. (1998) Enterprising Nonprofits. *Harvard Business Review Jan-Feb* pp.55-67.

Denzin, N.K. (1978) Strategies of Multiple Triangulation in *The Research Act: A Theoretical Introduction to Sociological Methods*. McGraw-Hill Book Company.

De Vaus, D.A. (1991) Surveys in Social Research (3<sup>rd</sup> Ed) London: UCL Press

Dillman, D.A. (1978) Mail and Telephone Surveys: The Total Design Method. Wiley: New York.

Di Maggio, P.J. and W.W. Powell (1983) The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Fields. *American Sociological Review* April

Eikenberry, A.M. & J.D. Kluver (2004) The Marketization of the Nonprofit Sector: Civil Society at Risk? *Public Administration Review* 64(2) pp.132-140.

Foster, W. & J. Bradach (2005) Should Nonprofits Seek Profits? *Harvard Business Review* Feb pp.92-100.

Frumkin, P. & A. Andre-Clark (2000) When Missions, Markets, and Politics Collide: Values and Strategy in the Nonprofit Human Services. *Nonprofit and Voluntary Sector Quarterly* 29(supp 1) pp, 141-163.

Garratt, B. (2003) The Fish Rots from the Head: The Crisis in our Boardrooms: Developing the Crucial Skills of the Competent Director. Profile Business: London.

Giddens, A. (1979) Central Problems in Social Theory: Action, Structure, and Contradiction in Social Analysis. Berkeley: University of California Press.

Gill, M., R.J. Flynn & E. Reissing (2005) The Governance Self-Assessment Checklist. *Nonprofit Management and Leadership* 15(3) pp. 271-294.

Glasby, J. & Manthorpe, J. (2005) The use and abuse of history: researching the voluntary sector through case-study analysis in P. Alcock & D. Scott (eds) 'Close work' Doing qualititative research in the voluntary sector. Charities Aid Foundation.

Golden-Biddle, K. and H. Rao (1997) Breaches in the Boardroom: Organizational Identity and Conflicts of Commitment in a Nonprofit Organization. *Organization Science* 8(6) pp.593-611.

Green, J.C. & D.W. Griesinger (1996) Board Performance and Organizational Effectiveness in Nonprfit Social Service Organizations. *Nonprofit Management and Leadership* 6(4) pp. 381-402.

Green, J., F. Madijidi, T. Dudley & F. Gehlen (2001) Local Unit Performance in a National Nonprofit Organization. *Nonprofit Management and Leadership* 11(4) pp.459-467.

Guo, B. (2006) Charity for Profit? Exploring Factors Associated with the Commercialization of Human Service Nonprofits. *Nonprofit and Voluntary Sector Quarterly* 35(1) pp.123-138.

Hager, M.A., S. Wilson, T.H. Pollak & P.M. Rooney (2003) Response Rates for Mail Surveys of Nonprofit Organizations: A Review and Empirical Test. *Nonprofit and Voluntary Sector Quarterly* 32(2) pp.252-267.

Harris, M. (1989) The Governing Body Role: Problems and Perceptions in Implementation. *Nonprofit and Voluntary Sector Quarterly* 18(4) pp. 317-333.

Harris, M. (2001) This Charity Business: Who Cares? *Nonprofit Management and Leadership* 12(1) pp.95-109.

Helmer, H.W. (1996) A Director's Role in Strategy. *Directors and Boards* 20 pp. 22-25.

Herman, R.D., D.O. Renz & R.D. Heimovics (1997) Board Practices and Board Effectiveness in Local Nonprofit Organizations. *Nonprofit Management and Leadership* 7(4) pp. 373-385.

Herman, R.D. & D.O. Renz (1997) Multiple Constituencies and the Social Construction of Nonprofit Organization Effectiveness. *Nonprofit and Voluntary Sector Quarterly* 26(2) pp. 185-206.

Herman, R.D. & D.O. Renz (1998) Nonprofit Organizational Effectiveness: Contrasts Between Especially Effective and Less Effective Organizations. *Nonprofit Management and Leadership* 9(1) pp. 23-38.

Herman, R.D. & D.O. Renz (1999) Thesis on Nonprofit Organizational Effectiveness. *Nonprofit and Voluntary Sector Quarterly* 28(2) pp.107-126.

Herman, R.D. & D.O. Renz (2000) Board Practices of Especially Effective and Less Effective Local Nonprofit Organizations. *American Review of Public Administration* 30(2) pp. 146-160.

Herman, R.D. & D.O. Renz (2008) Advancing Nonprofit Organizational Effectiveness Research and Theory. *Nonprofit Management and Leadership* 18(4) pp. 399-415.

Herman, R.D. & F.P. Tulipana (1985) Board-Staff Relations and Perceived Effectiveness in Nonprofit Organizations. *Nonprofit and Voluntary Sector Quarterly* 14 pp. 48-59.

Holland, T.P. & D.K. Jackson (1998) Strengthening Board Performance. *Nonprofit Management and Leadership* 9(2) pp. 121-134.

Horton, J. (2006) Confidence in Charities Returning After Scandals. Edinburgh Evening News, Tue 7 Mar 2006

Houle, C.O. (1990) Governing Boards: Their Nature and Nurture. Jossey-Bass. San Francisco

Hoye, R. (2004) Leader-Member Exchanges and Board Performance of Voluntary Sport Organizations. *Nonprofit Management and Leadership* 15(1) pp.55-70.

Hyndman, N (1990) Charity Accounting – An Empirical Study of the Information Needs of Contributors to UK Fund Raising Charities. *Financial Accountability and Management*. 6(4) p. 295-307.

Hyndman, N. & P. McDonnell (2009) Governance and Charities: An Exploration of Key Themes and the Development of a Research Agenda. *Financial Accountability and Management* 25(1) pp.5-31.

Iecovich, E. (2004) Responsibilities and Roles of Boards in Nonprofit Organizations. *Nonprofit Management and Leadership* 15(1) pp. 5-24.

Inglis, S., T. Alexander & L. Weaver (1999) Roles and Responsibilities of Community Nonprofit Boards. *Nonprofit Management and Leadership* 10(2) pp. 153-167.

Inglis, S. & L. Weaver (2000) Designing Agendas to Reflect Board Roles and Responsibilities. *Nonprofit Management and Leadership* 11(1) pp. 65-77.

Ingley, C.B. & N.T. Van der Walt (2001) The Strategic Board: The Changing Role of Directors in Developing and Managing Corporate Capability. *Corporate Governance: An International Review* 9 pp. 174-185.

Ingram, R.T. (2002) Ten Basic Responsibilities of Nonprofit Boards. Boardsource.

Jackson, D.K. & T.P. Holland (1998) Measuring the Effectivness of Nonprofit Boards. *Nonprofit and Voluntary Sector Quarterly* 27(2) pp. 159-182.

Jeavons, T.H. (1992) When Management is the Message: Relating Values to Management Practice in Nonprofit Organizations. *Nonprofit Management and Leadership* 2(4) pp. 403-417.

Kakabadse, A., N. Kakabadse & A. Myers (2009) *Boards, Governance and Leadership of the Third Sector.* OSCR: Scotland.

Keen and Packwood (1997) 7. Case Study Evaluation, in Mays and Pope (eds) *Qualitative Research in Health Care*, BMJ Publishing Group.

Kennedy, J.M. & B. Vargus (2001) Challenges in Survey Research and Their Implications for Philanthropic Studies Research. *Nonprofit and Voluntary Sector Quarterly* 30(3) pp. 483-494.

Landsberg, B.E. (2004) The Nonprofit Paradox: For-Profit Business Models in the Third Sector. *The International Journal of Not-for-Profit Law* 6(2).

Lapsley, I. (2004) 'Challenges to Nonprofit Organisations in a Changing World' I D. Witt, R. Putschert, & R. Schauer (eds.) Funktionen und Leistungen von Nonprofit-Organisationen, Deutscher Universitäts Verlag, Gabler, Edition Wissenschaft.

Leatherwood, M.L. & D. O'Neal (1996) The Transformation of Boards in Corporate and Not-for-Profit Sectors: Diminishing Differences and Converging Contexts. *Corporate Governance* 4(3) pp. 180-192

Leiter, J. (2005) Structural Isomorphism in Australian Nonprofit Organizations. *Voluntas* 16(1) pp.1-31.

Leiter, J. (2008) Nonprofit Isomorphism: An Australia-United States Comparison. *Voluntas* 19(?) pp.67-91.

Lorsch, J.W. (1995) Empowering the Board. *Harvard Business Review* 73 pp.107-117

McAdam, T.W. & D.L. Gies (1985) Managing Expectations: What Effective Board Members Ought to Expect from Nonprofit Expectations. *Nonprofit and Voluntary Sector Quarterly* 14 pp. 77-88.

McCambridge, R. (2004) Underestimating the Power of Nonprofit Governance. *Nonprofit and Voluntary Sector Quarterly* 33(2) pp.346-354.

McClusky, J.E. (2002) Re-thinking Nonprofit Organization Governance: Implications for Management and Leadership. *International Journal of Public Administration* 25(4) pp. 539-559.

McFarlan, F.W. (1999) Working on Nonprofit Boards: Don't Assume the Shoe Fits. *Harvard Business Review* Nov-Dec, pp.64-80.

Meyer, J.W. (1994) Institutional and Organizational Rationalization in the Mental Health System in W.R. Scott and J.W. Meyer (eds.) *Institutional Environments and Organizations*. Sage Publications: London.

Meyer, J. and B. Rowan (1977) Institutionalized Organizations: Formal Structure as Myth and Ceremony. *American Journal of Sociology*, 83(2) pp.340-363.

Miller, J.L. (2002) The Board as a Monitor of Organizational Activity. *Nonprofit Management and Leadership* 12(4) pp. 429-450.

Morgan, D.L. (1998) Practical Strategies for Combining Qualitative and Quantitative Methods: Applications to Health Research. *Quantitative Health Research* 8(3) pp. 362-376.

Morrill, C. and C. McKee (1993) Institutional Isomorphism and Informal Social Control: Evidence from a Community Mediation Center. *Social Problems* 40(4) pp.445-463.

Mouritsen, J. and P. Skaerbaek (1995) Civilization, Art and Accounting: The Royal Danish Theater – An Enterprise Straddling Two Institutions, in W.R. Scott and S. Christensen (eds) *The Institutional Construction of Organizations*, Sage.

Myers, J. & R. Sacks (2003) Tools, Techniques and Tightropes: The Art of Walking and Talking Private Sector Management in Non-profit Organisations, is it Just a Question of Balance? *Financial Accountability and Management* 19(3) pp.287-305.

National Council for Voluntary Organisations (NCVO) (1999) *Research Quarterly*, No. 6, July. NCVO. London.

National Statistics (2009) *Housing Statistics for Scotland: Key Trends Summary*. National Statistics: Edinburgh

Neary, K. & K. Gibb (2010) *New Climate, New Challenges*. Chartered Institute of Housing (Scotland): Edinburgh.

Office of the Scottish Charity Regulator (OSCR) Scottish Charities 2005. Available at www.oscr.org.uk

Office of the Scottish Charity Regulator (OSCR) Scottish Charities 2008. Available at www.oscr.org.uk

Office of the Scottish Charity Regulator (2006a) Guidance for Charity Trustees.

Office of the Scottish Charity Regulator (2006b) Charity Regulation in Scotland: A Brief Guide. OSCR.

OSCR (2008) One Plus: One Parent Families. OSCR

OSCR (2010) Charity Trustee Duties. Available at: http://www.oscr.org.uk/CharityTrusteeDuties.stm

O'Regan, K. & S.M. Oster (2005) Does the Structure and Composition of the Board Matter? The Case of Nonprofit Organizations. *The Journal of Law, Economics and Organization* 21(1) pp. 205-227.

Ostrower, F. & M.M. Stone (2006) 'Governance: Research Trends, Gaps and Future Prospects', in W.W. Powell & R. Steinberg (eds.) *The Nonprofit Sector, A Research Handbook* (2<sup>nd</sup> ed, Yale University Press, New Haven CT).

Parker, L.D. (2003) Financial Management Strategy in a Community Welfare Organisation: A Boardroom Perspective. *Financial Accountability and Management* 19(4) pp. 341-374.

Parker, L.D. (2007) Boardroom Strategizing in Professional Associations: Processual and Institutional Perspectives. *Journal of Management Studies* 44(8) pp. 1454-1480.

Perren, L. and P. Grant (2000) The Evolution of Management Accounting Routines in Small Businesses: A Social Construction Perspective. *Management Accounting Research* 11(?) pp.391-411.

Pharoah, C. (2005) Foreword in 'Close Work' Alcock and Scott (eds) Charities Aid Foundation,

Polit, D.F., Beck, C.T. & Hungler, B.P. (2001), Essentials of Nursing Research: Methods, Appraisal and Utilization, 5<sup>th</sup> Ed., Philidelphia: Lippincott Williams & Wilkins

Prescott, P.A. & Soeken, K.L. (1989) The Potential Uses of Pilot Work. *Nursing Research* 38, pp.60-62

Provan, K.G. (1980) Board Power and Organizational Effectiveness Among Human Service Agencies. *Academy of Management Journal* 23(2) pp. 221-236.

Putnam, R. (2000) *Bowling alone: The collapse and revival of American community*. New York: Simon & Schuster.

Ritchie, J., L. Spencer & W. O'Connor (2003) Carrying Out Qualitative Analysis, in Ritchie & Lewis (eds) *Qualitative Research Practice*. Sage: London.

Robertson, J. (2003) Cancer Charity's Missing £11.5 m. The Scotsman, Sat 24 May 2003

Ryan, W.P. (1999) The New Landscape for Nonprofits. *Harvard Business Review* Jan-Feb pp.127-136.

Russell, L. & D. Scott (2005) The Challege of Qualitative Case Studies: Close Encounters of a Paradoxical Kind in P. Alcock & D. Scott (eds) 'Close work' Doing Qualitative Research in the Voluntary Sector. Charities Aid Foundation.

Scott, W.R. & J.W. Meyer and associates (1994) *Institutional Environments and Organizations: Structural Complexity and Individualism.* Beverly Hills: Sage.

Scotland's Colleges (2006) Guide for College Board Members

Scottish Housing Regulator (2008) Regulatory Code of Governance. SHR: Glasgow

Scottish Housing Regulator (2009) Social Landlords in Scotland: Shaping up for Improvement. SHR: Glasgow

Sergeant, A., J. Ford & D.C. West (2000) Widening the Appeal of Charity. *International Journal of Nonprofit & Voluntary Sector Marketing* 5(4) pp.318-332.

Shelter (2010) www.shelter.co.uk

Siciliano, J.I. (1996) The Relationship of Board Member Diversity to Organizational Performance. *Journal of Business Ethics* 15 pp. 1313-1320.

Siciliano, J.I. (2008) A Comparison of CEO and Director Perceptions of Board Involvement in Strategy. *Nonprofit and Voluntary Sector Quarterly* 37(1) pp.152-162.

Silverman, D. (2001) *Interpreting Qualitative Data – Methods for Analysing Talk, Text and Interaction* (2<sup>nd</sup> Ed.) Sage: London

Smith, J (1997) Nonresponse Bias in Organizational Surveys: Evidence From a Survey of Groups and Organizations Working for Peace. *Nonprofit and Voluntary Sector Quarterly* 26(3) pp.359-368

Spaeth, J.L. & D.P. O'Rourke (1994) Designing and Implementing the national Organization Study. *American Behavioural Scientist* 37 pp. 872-890

Steane, P.D. & M. Christie (2001) Nonprofit Boards in Australia: a Distinctive Governance Approach. *Corporate Governance* 9(1) pp. 48-58.

Stone, M.M. (1991) The Propensity of Governing Boards to Plan. *Nonprofit Management & Leadership* 1(3) pp.203-215.

Stone, M.M. & F. Ostrower (2007) Acting in the Public Interest? Another look at Research on Nonprofit Governance. *Nonprofit and Voluntary Sector Quarterly* 36(3) pp.416-438.

Storey, J. (2001) Charities are Facing a Crisis of Their Own, *Marketing Week*, April 26<sup>th</sup>, p.22.

Taylor, B.E., R.P. Chait & T.P. Holland (1991) Trustee Motivation and Board Effectiveness. *Nonprofit and Voluntary Sector Quarterly* 20(2) pp. 207-224.

Taylor, B.E., R.P. Chait & T.P. Holland (1996) The New Work of the Nonprofit Board. *Harvard Business Review* Sept-Oct pp. 36-46.

Tiejlingen van E.R. & V. Hundley (2001) *The Importance of Pilot Studies*. Social Research Update Issue 35, University of Surrey

Tomaskovic-Devey, D., Leiter, J. & Thompson, S. (1994) Organizational Survey Nonresponse. *Administrative Science Quarterly*, 39 pp. 439-457.

Weisbrod, B.A. (1988) The Nonprofit Economy. Lexington Heath.

Widmer, C. (1993) Role Conflict, Role Ambiguity, and Role Overload on Boards of Directors of Nonprofit Human Service Organizations. *Nonprofit and Voluntary Sector Quarterly* 22(4) pp. 339-356.

Wood, M. (1992) Is Governing Board Behaviour Cyclical? *Nonprofit Management and Leadership* 3(2) pp.139-163.

Yin, R.K. (2003) Case Study Research. Sage Publications.

Yin, R.K. (1993) Applications of Case Study Research. London: Sage.

Westerdahl, S (2001) 'Between Business and Community': A rural Co-op and its Accounting Practice. *Financial Accountability and Management*. 17 (1) pp. 59-72

Young, D.R. (1998) Commercialism in Nonprofit Social Service Associations: Its Character, Significance, and Rationale. *Journal of Policy Analysis and Management* 17(2) pp.278-297.

# **APPENDICES**

Appendix 1: Pilot Study Questionnaire

Appendix 2: Analysis of Pilot Study Questions

Appendix 3: Main Survey Questionnaire

Appendix 4: Pilot Study Results

Appendix 5: Interview Schedule

Appendix 6: Publications

## PILOT STUDY QUESTIONNAIRE

# **Governance Issues in Charities**

- 1. How well defined are the duties and responsibilities of the board?
- 2. How would you describe how your governance model works?
- 3. Are you satisfied with the number of business people on your board?
- 4. What do business people bring to your board in terms of expertise/ experience? (please provide examples)
- 5. Can you give any examples of business practices which members of the board with business experience have introduced to your organisation? How were they received?
- 6. Do board members with a business background understand the specific context of your charity?
- 7. How much time do board members spend on strategic rather than operational issues?
- 8. What is the contribution of people with a business background to functions such as:
  - setting the organisation's mission and values;
  - helping to raise funds and resources;
  - financial management oversight;
  - reviewing and deciding strategic direction; and
  - reviewing board performance.
- 9. How does the board work with the management of the charity?
- 10. Does the inclusion of business people as board members increase public confidence in the organisation?
- 11. Are controls/ regulation of charities excessive?

## Please return to:

Vicky Lambert

University of Edinburgh, Management School, William Robertson Building, 50 George Square, EH8 9JY

Email: Vicky.lambert@ed.ac.uk

# ANALYSIS OF PILOT STUDY QUESTIONS

**Table A1: Analysis of Pilot Questions** 

Question	Response rate	Issues	Action
1. How well defined are the duties and responsibilities of the Board?	14/14	Generally well answered with sufficient detail. One org did only reply 'well defined' and gave no other info.	To eliminate answers such as 'well defined', give respondents an option along a scale then ask specifically where they are defined to ensure sufficient information.
2. How would you describe how your governance model works?	14/14	Responses varied – for ex. 2 orgs responded simply 'very well' and another 'successfully' Term 'governance model' possibly too board and open to misunderstanding.	Eliminate this qu and ask something more specific
3. Are you satisfied with the number of business people on your board?	14/14	Some one word answers – 2 'yes' one 'very'.	Make this yes/ no answer
4. What do business people bring to your board in terms of expertise/experience? (please provide examples).	14/14	Not all provided specific examples	
5. Can you give any examples of business practices which members of the board with business experience have introduced into your organisation? How were they received?	13/14	6 orgs were unable to describe any instances where this had happened.	Qu 4 & 5 are similar – roll into one. Put 'how were they received' as separate qu on a scale.
6. Do board members with a business background understand the specific context of your charity?	14/14	Lack of detail – tendency for one word answers	Put this as closed qu with scale. May need to also ask how they ensure that business people understand the specific context.

7. How much time do board members spend on strategic rather than operational issues?  8. What is the contribution of people with a business background to functions such as: -setting the orgs mission and values -helping to raise funds and resources -financial management oversight -reviewing and deciding strategic direction -reviewing board performance	14/14	Some did not quantify specifically  Confusion – 7 respondents responded to each item individually, 5 responded as a whole	May be better to ask respondents to tick specific times on a scale.  Omit this question
9. How does the Board work with the management of the charity?	12/14	Many detailed responses but also a tendency towards one word answers.	
10. Does the inclusion of business people as board members increase public confidence in the organisation?	14/14	Answers not very detailed – tendency towards 'yes' answers.	Have this as a yes/ no option
11. Are controls/ regulation of charities excessive?	14/14	Generally well answered although tendency towards 'yes' answers.	Introduced scale to this qu and then ask for justification in next question.

## MAIN SURVEY QUESTIONNAIRE



## Governance in Charities

**Please return to**: Vicky Lambert, University of Edinburgh, Management School, William Robertson Building, 50 George Square, EH8 9JY Vicky.lambert@ed.ac.uk Tel: 0131 650 3791

# Name of Organisation:

# **Governance**

**1.** (a) How well defined are the roles and responsibilities of your board? Please highlight one description on the scale below:

Very well defined well defined not well defined not defined don't know

- **(b)** Where are they outlined?
- (c) What is done to make sure board members are aware of these specific roles and responsibilities?
- **2.** (a) How much time do your board members spend on strategic rather than operational issues? Please indicate below.

0-49% 50-59% 60-69% 70-79% 80-89% 90-100%

- (b) Can you give examples of strategic and operational tasks carried out?
- **3.** (a) Is the control/ regulation of charities excessive? Please highlight the appropriate answer.

Very excessive excessive not very excessive not excessive don't know

- **(b)** Please explain why this is the case.
- (c) In what way has the Office of the Scottish Charity Regulator had an impact on the way your charity is governed?
- **4** (i) Please indicate how effective governance is in your charity by highlighting the appropriate description below:

Effective Very effective Ineffective Very Ineffective Don't Know (ii) Is there anything that could be done to improve governance in your charity? **Business People on the Board 5.** (a) How many people sit on your board? **(b)** What proportion of them are from business? (c) Are you satisfied with the number of business people on your board? Please indicate: Yes/No 6. (a) Can you give any examples of practices which members of the board with business experience have introduced to your organisation? **(b)** How were they received? Please highlight appropriate description below: Very well received well received badly received very badly received don't know 7. (a) Why do business people serve on your board? (bi) To what extent do they understand the specific context of your charity? Please indicate below: Very well well not well not at all don't know (b ii) Do you have a system in place to ensure that they do? Please give details. 8. Please rate how strongly you agree or disagree with the following statement by highlighting one of the options below:

Strongly Agree Agree Disagree Strongly Disagree Don't Know

'The inclusion of business people as board members in charities increases public confidence in the organisation in comparison with the inclusion of other individuals'

**9.** Is there anything else you wish to comment on?

#### PILOT STUDY RESULTS

#### Findings from the Pilot Study

An Investigation of Business Expertise in Charity Governance: The Case of Arts, Heritage, Culture and Science

The results from the pilot study outlined in the previous chapter are discussed below, these are analysed under three separate headings; 1. Governance Frameworks, 2. Business Expertise in Charities and 3. Trust and Confidence.

#### 1. Governance Frameworks

Governance was investigated through asking charities about the duties and responsibilities the board actually carries out, the relationship between the board and management, their governance model, and the split between strategic and operational duties carried out by the board.

Charities were asked how well defined the duties and responsibilities of their Board were. The majority of organisations looked to external guidance in defining these roles with few relying on internal guidance. The four public bodies included in the study adhered to government guidance published in 2006 'On Board' which is a guide for board members of public bodies in Scotland. In addition two of the other charities noted that they made use of OSCR for their governance structure. Of the remaining organisations, one used an act of Parliament and the others relied on their memorandum and articles of association although these may be less prescriptive and tight than the public bodies and those guided by OSCR.

In terms of a specific model of governance, three of these organisations described their governance model as a business management approach. For example, two of the charities noted the increasing emphasis being placed on the strategic direction of the organisation, with the other noting that the board operates on a portfolio basis, with a deliberate mix of professional people with contrasting skills. Three other organisations described their model as being more procedural, with committee structures. Of the remaining organisations, one noted that due the rapid growth of the organisation in recent years, the governance model is currently under review, the rest described their model as working satisfactorily.

The issue of the amount of time Board members spend on strategic rather than operational issues was also investigated. The majority of organisations noted that the focus was predominantly on strategy, only occasionally were board members brought in to advise on operational issues. One organisation noted that the split between strategic and operational issues was 60% to 40% in favour of strategic issues. Another organisation's board split their time equally between strategic and operational issues. This was in contrast to some commentators on this issue, for example Chait & Taylor (1989) who give examples of situations whereby there is a tendency for board members to take on management and administrative responsibilities.

The relationship between the board and the management of the charity was also considered. Contrary to some literature which highlights tensions between management and the board (Chait and Taylor, 1989), the general picture uncovered here was one of management working in harmony with the board; for example, one organisation noted that the management carries out the strategy agreed by the board and trustees. Another notes that 'the board provides a great deal of expert help and is seen as one member of the team, with the management and staff as the others'. A further example includes 'the board meet and is advised by senior managers of the charity at board meetings. The board members also attend our events, visit our operational sites and make themselves familiar with the activities of the charity and its managers in a variety of ways'. For other organisations it is more in terms of operational efficiency, for example; 'the management service the committees and boards. The general director is responsible for working closely with the chair to drive and direct the board in line with agreed policy'.

Contrary to the view of many commentators on charity governance, such as Taylor et al (1996), there appears to be a strong governance framework in place in these organisations.

## Business Expertise in Charities.

In terms of the number of business people charities actually had on their Boards, the general feeling among these organisations was that they were satisfied with the numbers serving on their boards. However, there were three organisations that would like more. For example, one of these organisations noted that it was increasingly important for their organisation and nonprofit organisations in general to have people with appropriate strategic/ management experience to lead the organisation. This particular organisation placed substantial reliance on donations and fundraising, and therefore felt that more business people on the board would help to bring in money by helping with fundraising and increasing business contacts. Ideally the felt they would like at least 4-5 business practitioners on the board. However, this organisation and one of the other organisations that expressed a desire for increased numbers of business people on their board were both public bodies and therefore had their boards appointed by the Scottish Executive, so had a lack of control over who served on them. The final organisation that wished more individuals from business noted that it only had one business person currently serving on its board, this was in comparison to six academics.

When the question of what business people actually bring to the Board in terms of experience and expertise was posed, there seemed to be two sets of answers to this question; firstly they brought functional expertise, and secondly they brought with them a different way of thinking. In terms of functional expertise, examples given included; finance, marketing, accounting, IT. In terms of helping the organisation to think in different way, two organisations in particular noted how they benefited from a more business orientated way of thinking; for example one organisation noted that

arts organisations tended to have small management teams that benefited hugely from the guidance and mentoring of a board with a range of business expertise. They went on to note that a board member with a background in marketing, for example can provide expertise an arts organisation could not hope to buy. Another organisation emphasised the need to now focus on the commercial aspects of their operations, and they noted how business people advised them on managing risk and also helped them to seek efficiencies.

When asked to give examples of business practices which members of the Board with business experience had introduced to the organisation, six of the 14 organisations were unable to describe any instances where this had happened. However, others gave various examples of this. In terms of accounting, one organisation noted the presence of two chartered accountants on their board, who modernised the practice of the audit committee in terms of managing major risks, they also developed a clearer format for the strategic plan, and they insisted on a 10 year business plan in terms of potential investments in new developments with assumptions on income and expenditure clearly spelt out. Another organisation noted that their management accounts format and covering explanatory notes had been greatly improved by input from Board members with the responsibility for reporting to their own boards. Similarly, another organisation noted that there had been more of a push towards management accounts on a regular basis. A further example given referred to human relations management, this had been updated and was well received by staff who previously had erratic terms and conditions and no performance appraisal. In terms of property development, one individual assisted on the development that the organisation intended to carry out on a spare piece of land. These positive findings in terms of what charities gain from having business people on their board in terms of expertise and business practices they bring concur with the views of Bowen (1994) who notes that nonprofit board have a considerable amount to learn from those with a business background.

After investigating the involvement of business people in the organisation, the question was posed as to whether board members with a business background

understood the context of the specific charity. The majority of these organisations said that they did, but four in particular noted the need to educate these business people through training and dissemination of information. Another noted that although they feel that the business people on their board do understand the specific context of their charity, they felt that business members were not interested in the role of the charity. They felt that charitable status is not important to business people, they are only concerned with the best interests of organisation.

The contribution of people with a business background to functions such as: setting the organisations mission and values; fundraising; overseeing financial management; reviewing and deciding strategic direction; and reviewing the performance of the Board was also investigated. These functions were found by Cornforth and Simpson (2001) to be vital in terms of the overall effectiveness of the Board. Five of these organisations felt that business people made vital contributions to all these functions, one organisation in particular noted the contribution of business people in fundraising and also financial management oversight. In terms of the contribution of business people in setting the organisations mission and values, six of these organisations described the importance of business people in carrying out this function - they described them as 'vital', 'fully involved', 'heavily involved', 'considerable contribution', and 'significant'. The contribution of business people to fundraising was described as 'invaluable/key' by three of the organisations, however, the other three noted they were not involved as they received their funds from the Scottish Executive. With regards to financial management oversight, organisations noted that there was some form of contribution, however there were variations; one noted that the contribution was 'critical', however another noted that they only contributed 'up to a point' and the organisation was large enough to employ professional financial managers. All organisations noted a positive influence of business people to reviewing and deciding strategic direction. Three noted the extensive involvement of business people in this function; one noted that there was significant involvement, however they were also influenced by the Scottish Executive, and another responded that setting strategic direction was a two-way process and was carried out in partnership with the director and his senior management team.

## Trust and Confidence

In terms of the issue surrounding public confidence, most charities felt that the presence of business people enhanced this, however some remained unsure. One organisation felt that the presence of business people did not noticeably increase confidence, they felt that the inclusion of members from the art sector was more likely to be welcomed. Another organisation seemed unsure of whether they enhanced public confidence, noting that 'While the inclusion of highly competent business people on our board is very helpful to the operation of the board and the organisation I am not sure that it does...the public are more likely to be influenced by the performance of the organisation itself in delivering its objectives than any big names on the board'. Another organisation felt that confidence would be increased 'only if those people are known outside the organisation in their own field and are well respected'.

In contrast, the remainder of these organisations felt that the presence of these business people did undoubtedly increase public confidence. For example, one organisation responded to this question in the following way; 'Without a doubt. Those considering donating large sums just would not without the assurance of sound business expertise at the top'. Another organisation said 'Yes. The Chairman is setting up mechanisms to enhance the relationship that Board Members need to have with senior business people, local authority elected representatives, and other relative board level people in the public and NGO sectors for precisely this reason. This is nothing new, but from this year onwards, it is being driven forward much more, thus improving the links already there'. Two other organisations noted that public confidence was increased particularly if the individuals were well known in their own field.

The question was also posed as to whether controls/ regulation of charities were excessive, of the 14 respondents, ten felt that they were not, and four implied that that in some ways they may well be. Examples of those that felt they may be excessive included the following 'they are certainly ever increasing and as a result absorb more resources away from our core business', another organisation noted differences between the types of controls used, noting that they were 'becoming more onerous but OSCR have taken a sensible proportionate approach. Scottish Executive and politicians less so!'. In addition, another organisation highlighted some internal problem caused by these regulations; 'There are certainly many complex rules and regulations in place. The new Charities Act and related operating procedures can conflict with our own government policy guidance which makes it complex for organisations like ours to meet all requirements'.

Of those organisations that felt regulation was not excessive, two noted that this was the case due to the potential for abuse of charitable giving/ public trust. For example, one response here was 'the previous legislative regime in Scotland was weak and open to abuses of public trust which damaged the reputation of legitimate charities. The 2005 Act and the setting up of OSCR was a welcome step'. In addition to this one other organisation noted that they also welcomed the introduction of OSCR and related reporting requirements. Another organisation noted that charities should be required to work economically, efficiently and effectively.

#### Conclusion of Results

There is considerable literature on how charities conduct their activities, with many concerns around the increasing influence of business persons in the charity world. There have been suggestions this direction will distort or undermine the fundamental ethos and values of charitable organisations. It has emerged from the findings of this pilot study that there is indeed an increasing reliance on business expertise in charities. However, the findings suggest that charities are very positive about the kinds of contribution business persons can make to the conduct of the charities.

There is evidence that business members of charitable boards make contributions to the strategic direction and also to the wider adoption of modern methods of management in charities. These findings are based on a selection of large charities in an important area of charitable activity, namely those with the purpose of the advancement of arts, heritage, culture and science. These findings may also be found in other areas of charitable activity, but this requires further investigation.

#### INTERVIEW SCHEDULE

How long have you been a member of the Board? Do you have any other experience of voluntary board membership? What were the main reasons you became a Board member?

What in your view are the key roles of the Committee?

- Do you see a clear distinction between strategic and operational activities?
- What if any operational activities should the Committee be interested in?
- How are Committee members actively involved in determining the strategic direction of the HA?
- Target setting and monitoring performance against targets?
- Risk management?
- Monitoring finances?

What is done to make sure the Committee carries out these roles effectively?

- Reflected in agendas?
- Handbook

What do you see as your most important role as a Committee member?

To what extent do you feel that effective use is made of your experience and expertise?

• Can you give any examples of how you have used your specific expertise on the Committee?

In which areas do you feel the Committee can provide examples of good practice in governance?

How effective is the relationship between the Committee and management – how is it maintained?

Sub-Committees; what are they, how does the information feed into the main committee?

Committee member development

How do you feel about current control/regulation of housing associations? What has been the impact of OSCR & Scottish Housing Regulator?

What are the biggest barriers to good governance in the sector?

Are there any ways in which governance could be improved?

# **PUBLICATIONS**

The following section presents two publications – one forthcoming and one published – which emanate from this thesis.

## **Published:**

- Lambert, V. & Lapsley, I. (2011) Business Expertise in Charities, in A. Heilmair, Y. Russ, H. Sturm & T. Zitzmann (eds) *Verbandmanagement*, Gabler Verlag.
- Lambert, V. & Lapsley, I. (2010) Leadership and Governance in *Third Sector Research*, Taylor, R. (ed) Springer. New York.